WSR 23-23-016 EXPEDITED RULES DEPARTMENT OF REVENUE

[Filed November 2, 2023, 2:02 p.m.]

Title of Rule and Other Identifying Information: WAC 458-20-18801 Medical substances, devices, and supplies for humans—Drugs prescribed for human use-Medically prescribed oxygen-Prosthetic devices-Mobility enhancing equipment—Durable medical equipment.

Purpose of the Proposal and Its Anticipated Effects, Including Any Changes in Existing Rules: The department is updating WAC 458-20-18801 due to changes from recent legislation, SSB 5218 (2023). The changes include updates to RCW 82.08.814 and 82.12.814 to provide a new retail sales and use tax exemption for mobility enhancing equipment for complex needs patients.

Reasons Supporting Proposal: The update is to conform the rule to SSB 5218, which passed during the 2023 legislative session.

Statutory Authority for Adoption: RCW 82.02.060, 82.08.814, and 82.12.814.

Statute Being Implemented: RCW 82.08.814 and 82.12.814.

Rule is not necessitated by federal law, federal or state court decision.

Name of Proponent: Department of revenue, governmental.

Name of Agency Personnel Responsible for Drafting: Justin Morehouse, 6400 Linderson Way S.W., Tumwater, WA, 360-534-1533; Implementation and Enforcement: John Ryser, 6400 Linderson Way S.W., Tumwater, WA, 360-534-1605.

This notice meets the following criteria to use the expedited adoption process for these rules:

Adopts or incorporates by reference without material change federal statutes or regulations, Washington state statutes, rules of other Washington state agencies, shoreline master programs other than those programs governing shorelines of statewide significance, or, as referenced by Washington state law, national consensus codes that generally establish industry standards, if the material adopted or incorporated regulates the same subject matter and conduct as the adopting or incorporating rule.

Corrects typographical errors, makes address or name changes, or clarifies language of a rule without changing its effect.

Explanation of the Reason the Agency Believes the Expedited Rule-Making Process is Appropriate: The expedited rule-making process is appropriate for this rule update because the department is incorporating changes resulting from 2023 legislation.

NOTICE

THIS RULE IS BEING PROPOSED UNDER AN EXPEDITED RULE-MAKING PROC-ESS THAT WILL ELIMINATE THE NEED FOR THE AGENCY TO HOLD PUBLIC HEAR-INGS, PREPARE A SMALL BUSINESS ECONOMIC IMPACT STATEMENT, OR PROVIDE RESPONSES TO THE CRITERIA FOR A SIGNIFICANT LEGISLATIVE RULE. IF YOU OBJECT TO THIS USE OF THE EXPEDITED RULE-MAKING PROCESS, YOU MUST EX-PRESS YOUR OBJECTIONS IN WRITING AND THEY MUST BE SENT TO Justin Morehouse, Department of Revenue, P.O. Box 47453, Olympia, WA 98504-7453, phone 360-534-1533, fax 360-534-1606, email JustinM@dor.wa.gov, AND RECEIVED BY January 22, 2024.

November 2, 2023

Atif Aziz Rules Coordinator

OTS-5000.1

AMENDATORY SECTION (Amending WSR 22-19-025, filed 9/13/22, effective 10/14/22)

WAC 458-20-18801 Medical substances, devices, and supplies for humans—Drugs prescribed for human use—Medically prescribed oxygen— Prosthetic devices—Mobility enhancing equipment—Durable medical equipment.

PART 1 - INTRODUCTION

- (101) **Introduction**. This rule provides tax-reporting information for persons making sales of medical products. It also provides information about the retail sales tax and use tax exemptions available for the sale and use of certain medical products for humans.
- (102) How is this rule organized? This rule is divided into five parts as follows:
- (a) Part 1 Introduction. Part 1 provides information relating to the purpose of the rule, how the rule is organized, and provides a listing of additional rules that may be helpful to the reader in determining taxability related to medical products.
- (b) Part 2 Medical products. Part 2 of this rule identifies what "medical products" include for purposes of this rule. Medical products is not a statutory term, but instead, is a term used simply to collectively describe the medical items addressed by this rule.
- (c) Part 3 Applicable taxes. Part 3 of this rule provides information on the taxes that apply to the sale, use, purchase, or manufacture of medical products.
- (d) Part 4 Common exemptions. Part 4 of this rule provides information on common retail sales tax and use tax exemptions related to medical products.
- (e) Part 5 Bundled transactions. Part 5 of this rule addresses the treatment of bundled transactions involving medical products.
- (103) How are examples included in this rule to be used? This rule contains examples, which identify a number of facts and then state ((s)) a conclusion. The examples should be used only as a general quide. The tax results of other situations must be determined after a review of all of the facts and circumstances.
- (104) What are some other department of revenue rules that address medical or health related providers that might apply? The department of revenue (department) has adopted other rules addressing the taxability of various activities related to the providing of health care. Readers may want to refer to the following rules for additional information:
- (a) WAC 458-20-150, Optometrists, ophthalmologists, and opticians:
- (b) WAC 458-20-151, Dentists, audiologists, and other health care providers—Dental laboratories and dental technicians;

(c) WAC 458-20-168, Hospitals, nursing homes, assisted living facilities, adult family homes and similar health care facilities.

PART 2 - MEDICAL PRODUCTS

- (201) What are medical products for purposes of this rule? Medical products include durable medical equipment, drugs, mobility enhancing equipment, over-the-counter drugs, and prosthetic devices as defined by Washington statute. Medical products also include other tangible personal property used for medical purposes, not covered by one of the statutory definitions. The remainder of Part 2 of this rule describes these medical products.
- (202) What is durable medical equipment? Durable medical equipment is equipment, including repair and replacement parts for durable medical equipment that:
 - (a) Can withstand repeated use;
 - (b) Is primarily and customarily used to serve a medical purpose;
- (c) Generally is not useful to a person in the absence of illness or injury; and
- (d) Is not worn in or on the body. See RCW 82.08.0283. Also, see subsection (206)(b) of this rule for an explanation of what is considered "worn in or on the body."

Table 1 provides a nonexclusive list of durable medical equipment product examples.

Table 1

Durable Medical Equipment Examples

- · Anesthesia machine and ventilator
- · Apnea monitors
- Atomizers (medical Reusable)
- Beds, bags, trays, bedpans, commodes, pads, pillows, crash carts, lamps, bulbs, and tables (medical)
- Blood parameter monitor, pulse oximetry equipment, and blood gas analyzer
- Bone growth stimulator (not worn on the body)
- Bovie (cauterization)
- Cardiopulmonary bypass machine
- Cofflator
- Continuous passive motion devices
- Continuous positive airway pressure (CPAP & BI-PAP) machine (not worn on the body)
- Diagnostic equipment Audiology, cardiology, mammography, radiology
- Electronic speech aids (not worn on the body)
- Endoscopes
- Enteral feeding bags, tubing, and connectors
- Feeding plugs
- · Glucose meters
- Instruments ((Reuseable)) Reusable, e.g., clamps, drills, forceps, retractors, scalpels, reamers, scissors
- Intravenous (IV) stands and poles
- Kidney dialysis devices
- Lasers
- Lithotripters

Durable Medical Equipment Examples

- Nebulizers
- Respiratory humidifier
- Reusable needles or reusable staplers
- Sling scales
- Stapler (must be empty as staples are not durable medical equipment)
- Stethoscopes, stirrups, and stretchers (medical)
- Suction regulators
- TENS units (not worn on the body)
- Tourniquets
- Ultrasound probes, transducers, and mini dopplers
- Whirlpools (medical)
- X-ray equipment
- (203) (a) What is a drug? A "drug" is a compound, substance, or preparation, and any component of a compound, substance, or preparation, other than food and food ingredients, dietary supplements, alcoholic beverages, or cannabis, useable cannabis, or cannabis-infused products:
- (i) Recognized in the official United States pharmacopoeia, official homeopathic pharmacopoeia of the United States, or official national formulary, or any supplement to any of them; or
- (ii) Intended for use in the diagnosis, cure, mitigation, treatment, or prevention of disease; or
- (iii) Intended to affect the structure or any function of the body. See RCW 82.08.0281.

Table 2 provides a nonexclusive list of drug product examples.

Table 2

Drug Examples

- Dermal fillers Injectable
- · Dialysis dialysate solution
- Federal prescription (RX) drugs, including biologicals
- Gases Medical grade (nitrous oxide, oxygen, carbon dioxide, helium)
- Implanted radioactive isotopes
- Insulin
- Parenteral nutrition formulas By prescription
- Prescription medicated cotton swabs and gauze wraps
- Sterile water 1cc, 5cc, 10cc vials, sterile normal saline (.9%) - 1cc, 5cc, 10cc vials - Solutions for adding to mixtures and irrigation
- Vaccines
- (b) Substances that are necessary to the performance of durable medical equipment are not drugs. A compound, substance, or preparation that is necessary for durable medical equipment to perform its function is not a drug, even when it otherwise meets the definition of drug in this subsection.
- (c) Examples of compounds, substances, preparations that are necessary in order for the durable medical equipment to perform its function.

- Example 1. A Coulter Blood Cell Counter uses an electrolytic solution to perform its function. The solution is entirely contained within the device and does not physically interact with the patient's tissue (blood) apart from the device. The device cannot perform its function without the electrolytic solution. The solution is an integral part of the Coulter Blood Cell Counter and is not a drug even though the device is used to diagnose disease and the test it performs is conducted pursuant to a prescription.
- **Example 2.** A cryoablation device uses extremely cold, thermally conductive solution inside a hollow probe or needle to freeze and remove diseased or malfunctioning cells within a patient's body. The solution is entirely contained within the device and does not physically interact with the patient's tissue apart from the device. The device cannot perform its function without the solution. The solution is an integral part of the device and is not a drug even though the device is used in the cure, mitigation, and treatment of disease as part of a prescribed procedure.
- Example 3. A specialized medical laser uses certain gases (e.g., argon, helium) to determine the wavelength of the light emitted. This allows the laser to identify specific cells or substance types. The gas is entirely contained within the laser and does not physically interact with the patient's tissue apart from the device. The device cannot perform its function without the gas. The gas is an integral part of the device and is not a drug even though the gas is consumed and the laser is used in the cure, mitigation, and treatment of disease as part of a prescribed procedure.
- (204) What is mobility enhancing equipment? Mobility enhancing equipment is equipment, including repair and replacement parts for mobility enhancing equipment that:
- (a) Is primarily and customarily used to provide or increase the ability to move from one place to another and is appropriate for use either in a home or a motor vehicle;
 - (b) Is not generally used by persons with normal mobility; and
- (c) Does not include any motor vehicle or equipment on a motor vehicle normally provided by a motor vehicle manufacturer. See RCW 82.08.0283.

Table 3 provides a nonexclusive list of mobility enhancing equipment products.

Table 3

Mobility Enhancing Equipment Examples

- Bath aids Raised toilet seat, tub and shower stools
- · Bed pull-up T
- Canes
- Car seats (mobility enhancing)
- Handrails and grab bars to assist in rising from commode, tub, or shower
- Lift chairs and replacement parts
- Lifts (hydraulic or electric) used to raise or transfer patients from bed to chair, commode, or bath
- Replacement parts for mobility enhancing equipment, e.g., batteries for electric wheelchairs
- · Scooters and transporters

Mobility Enhancing Equipment Examples

- Swivel seats enabling the disabled to rotate in order to rise from a chair
- Transfer belts to assist in the transfer of patients
- · Walkers
- · Wheelchairs
- Wheelchairs adapted for specific uses or functions, e.g., all terrain wheelchairs
- (205) Over-the-counter drugs. An over-the-counter drug is a drug that contains a label that identifies the product as a drug required by 21 C.F.R. Sec. 201.66, as amended or renumbered on January 1, 2003. The label includes:
 - (a) A "drug facts" panel; or
- (b) A statement of the "active ingredient(s)" with a list of those ingredients contained in the compound, substance, or preparation. See RCW 82.08.0281.

Table 4 provides a nonexclusive list of over-the-counter drug products.

Table 4

Over-the-Counter Drug Examples

- Antihistamines
- Anti-inflammatory
- Analgesic
- · Contact lenses solution
- Eternal nutrition formulas with drug facts box
- Hydrogen peroxide
- Medicated cotton swabs and gauze wraps (nonlegend)
- · Paviodine iodine
- · Rubbing alcohol
- (206) (a) What is a prosthetic device? A prosthetic device is a replacement, corrective, or supportive device, including repair and replacement parts for a prosthetic device, worn on or in the body to:
 - (i) Artificially replace a missing portion of the body;
- (ii) Prevent or correct a physical deformity or malfunction; or (iii) Support a weak or deformed portion of the body. See RCW 82.08.0283.

Table 5 provides a nonexclusive list of prosthetic device products.

Table 5

Prosthetic Device Examples

- Abdominal belts, binders, and supports
- Acetabular cups
- · Ankle brace
- Antiembolism stocking
- Artificial eyes, heart valves, larynx, limbs
- · Back braces
- · Bone cement and wax
- Bone pins, plates, nails, screws
- Breast implants and external prosthesis

Prosthetic Device Examples

- · Cervical collars
- · Cochlear implant
- Continuous positive airway pressure (CPAP) machines which are specifically designed to be wholly worn on the body and portable
- · Corrective eye glasses and contact lenses
- Dental prostheses including, but not limited to, full and partial dentures, crowns, inlays, fillings, braces, and retainers
- Drainage devices for single patient use because they serve the same drainage functions as the body's natural systems
- Ear, nose, and throat implants
- · Eye glass frames and lenses
- · Foley catheter
- · Gastric bands and intragastric balloons
- Hand and feet implants
- · Head halters
- · Hearing aids
- · Implanted pacemakers
- · Insulin pumps
- Knee immobilizers
- · Mastectomy surgical bras
- · Maxillofacial devices implanted
- Membrane implants (neutron, spinal, joint)
- · Ocular implants
- Orthobiologics implants
- Orthopedic shoes, shoe lifts, inserts, arch supports, heel protectors
- Pressure garments Edema gloves
- Pressure garments Mast pants, burn garments
- Salem sump with anti-reflux valve
- Shoulder and elbow implants
- Skin implants Synthetic
- Slings, braces, collars, casts, splints, embolism stockings, arch pads, pelvic traction belts, traction pulley clamp assemblies and cords
- · Slings Medical
- Specialized orthotic shoes, post-operation shoes, cast shoes, diabetic shoes and inserts, and other similar apparatus
- Speech aids (electronic) worn on the body
- · Sphincters Medical
- · Splints and splint materials
- Stent implants through endoscopy
- Stents (biliary, coronary and urinary)
- Stockings Compression
- · Sutures, staples, and skin glue for closing wounds
- Tendon implants
- TENS units worn on the body

Prosthetic Device Examples

- Testicular and penile implants
- · Trachea tubes
- Trusses
- (b) When is a device not worn on or in the body? For the purpose of this exemption, "worn on the body" means the entire device is something a person puts on or has on their person, to be carried with and habitually becomes part of the person as a whole, much in the sense that a person wears clothing or other personal items. Such devices are designed to be wholly worn on the body and portable. A device is not "worn on or in the body" simply because part of it is attached to the body in some way for a period of time. These devices cannot be partially floor-standing, plugged into an outlet, or moved by virtue of dragging, wheels, or with the assistance of a separate device (e.g., a cart or intravenous stand).
- (C) Examples of items that are not prosthetic devices worn on or in the body. The following are examples of items not considered prosthetic devices worn on or in the body.
- Example 4. Continuous positive airway pressure (CPAP) machines are commonly used by patients with sleep apnea disorders to facilitate normal breathing. Patients using a CPAP machine are normally hooked up to the machine via tubing and individually tailored masks. Even though the mask is normally "worn" for significant periods of time each night, the mask by itself cannot accomplish the intended purpose. The machine performing the function is not worn on the body as a complete system. Neither the mask separately, nor the machine as a whole system, is a prosthetic device.
- **Example 5.** Heart-lung machines generally replace the function of the heart and lungs during surgery, as well as regulating body temperature and providing an avenue of introduction for anesthetics or other medications directly into a patient's bloodstream. While a heart-lung machine is attached to the patient, it is commonly a floor-standing or wheeled unit and is not a prosthetic device.

PART 3 - APPLICABLE TAXES

- (301) What basic tax information do I need to be aware of when selling, purchasing, using, or manufacturing medical products? This subsection provides general tax-reporting information for persons who sell, purchase, use, or manufacture, medical products.
- (302) How are medical products taxed? In general, sales of medical products are taxable. Sales of medical products to consumers such as doctors, hospitals, or patients are subject to retailing business and occupation (B&O) tax and the retail sales tax. These taxes apply to the sale of medical products as follows:
- (a) Retail sales tax. Retail sales tax applies to the sale of medical products to a consumer unless a specific exemption applies. RCW 82.04.050 and 82.08.020. Specific exemptions are discussed in Part 4 of this rule.
- (b) Retailing B&O tax. There is no general B&O tax exemption for sales of medical products. Even if a sale of a medical product is exempt from retail sales tax, the gross proceeds from the sale of the medical product to a consumer is subject to the retailing B&O tax.
- (c) Wholesaling B&O tax. Sales to persons who reself the medical products (e.g., pharmacies) are subject to the wholesaling B&O tax.

Persons making wholesale sales should refer to WAC 458-20-102 for information regarding their responsibility to obtain a reseller permit.

(d) Manufacturing B&O tax. Persons who manufacture products including medical products, in this state are subject to the manufacturing B&O tax upon the value of these products. Manufacturers selling the products at retail or wholesale in this state are also subject to either the retailing or wholesaling B&O tax, as the case may be. In such cases, the manufacturer must report under both the "production" (manufacturing) and "selling" (wholesaling or retailing) classifications of the B&O tax, and claim a Multiple Activities Tax Credit (MATC). Refer to WAC 458-20-19301 for a more detailed explanation of the MATC.

Persons who manufacture molds or other products that they use in a manufacturing process are subject to the manufacturing B&O tax upon the value of the product manufactured. (See also WAC 458-20-112 and 458-20-134 regarding "value of products" and "commercial or industrial use," respectively.) Such persons also incur a use tax liability with respect to their use of the molds or products, unless a specific exemption applies. For example, RCW 82.12.02565 provides a use tax exemption for the use of certain molds in a manufacturing operation. Refer to WAC 458-20-13601 for additional information regarding the manufacturers machinery and equipment sales tax and use tax exemptions.

- (e) Use tax or deferred retail sales tax. Purchases of medical products at retail are subject to retail sales tax unless a specific exemption exists in the law. If the seller does not collect retail sales tax, a buyer who is not reselling the products must pay the retail sales tax (commonly referred to as the "deferred retail sales tax") or use tax directly to the department, unless the specific items purchased are exempt under the law. For additional information on use tax see WAC 458-20-178.
- (303) Retail sales tax should be paid by the consumer based on the principal use of the product. Some medical products can be put to both an exempt and taxable use. At the time of purchase a buyer may not know exactly how the item or items will be used. In such cases, retail sales tax must be paid to the seller at the time of purchase when the buyer expects to principally (i.e., more than 50 percent of the time) put the item to a taxable use in the normal course of business. However, if the buyer expects to principally put the item to use in an exempt manner, the buyer may provide the seller with an appropriately completed exemption certificate that lists the retail sales tax exempt item or types of items included in the purchase, such as a Streamlined Sales Tax Agreement Certificate of Exemption (SSUTA exemption certificate), or the seller may capture the relevant data elements that would otherwise be captured in a completed SSUTA exemption certificate, or otherwise meet the requirements of RCW 82.08.050(7). See subsection (304) of this rule for more information on exemption certificates and other department approved documentation. When a seller receives an appropriately completed exemption certificate or other approved documentation, that seller is relieved of the responsibility to collect the retail sales tax for those specific items or types of items identified on the certificate and sold in that transaction.
- (a) Items put to taxable use where tax was not paid. If the buyer does not pay sales tax on an item, and later puts that item to use in a manner that is not exempt of sales tax, the buyer must pay deferred sales or use tax to the department. The deferred sales tax liability should be reported by the buyer on the use tax lines of the excise tax return (including both state and local portions of the tax). The tax

should be reported based on the location and sales tax rate which is in effect where the buyer took possession of the item.

(b) Items put to exempt use where tax was paid. If the buyer does not give an exemption certificate or other approved documentation to the seller indicating a certain item is exempt of retail sales tax, or the seller does not capture the relevant data elements required under SSUTA or otherwise meet the requirements of RCW 82.08.050 $\bar{(7)}$, the seller must collect the tax at the time of purchase on that item. If the buyer later puts that item to first use in an exempt manner, the buyer may take a deduction on the excise tax return equal to the value of the item. This deduction should be claimed in the deduction column of the retail sales tax line, and should be identified as a "taxable amount for tax paid at source" deduction on the deduction detail worksheet. When completing the local sales tax section of the tax return, the value of the item must be credited using the seller's tax location code (assuming the buyer took possession of the item at the seller's location) and computed at the local sales tax rate paid to the seller.

(c) Examples.

Example 6. Purchase of items which are principally exempt. ABC Medical Center (ABC) purchases a case of sterile silicon tubing. One case contains 20 units of sterile tubing in individually sealed sterile packaging. The tubing purchased by ABC is either used to deliver medically prescribed oxygen from tanks to a patient (an exempt use), or used by ABC's laboratory to conduct certain tests (not an exempt use). At the time of purchase, ABC does not know how many of the 20 packages in the case will be used for oxygen tank systems versus how many will be drawn out of inventory by the lab. However, according to ABC's inventory records from past periods, the tubing will principally be used as part of the medically prescribed oxygen systems. ABC provides the seller of the tubing with a properly completed exemption certificate (in this case, the "Sales Tax Exemption Certificate for Health Care Providers") or other approved documentation. The seller is not required to collect retail sales tax on the case of sterile tubing. As ABC puts the tubing to use, it must keep track of when a package of tubing is used by the laboratory. Deferred sales tax is due and should be reported on and remitted with the excise tax return for the period in which ABC used the tubing.

Example 7. Purchase of items which are principally taxable. Assume the same items and situation as in Example 6, except that for this example, according to ABC's inventory records from past periods, the tubing will be principally used for retail sales taxable purposes in the laboratory. ABC cannot provide an exemption certificate or other approved documentation for purchase of the tubing and must pay retail sales tax to the seller. As ABC puts the tubing to use, it may keep track of when a package of tubing is put to exempt use with a medically prescribed oxygen system. ABC may then take on its excise tax return a tax paid at source deduction for the value of the package

(304) Sellers must obtain required exemption documentation or information on any retail sales exempted from the retail sales tax. Unless otherwise provided in this rule, sellers making retail sales to medical practitioners, nursing homes, and hospitals must obtain an exemption certificate approved by the department, such as a SSUTA exemption certificate, capture the relevant data elements required in completing a SSUTA exemption certificate, or otherwise meet the requirements of RCW 82.08.050(7) to document any tax-exempt sales of the

products discussed in this rule when those businesses are the consumers. Information about exemption certificates may be obtained by:

- (a) Using the department's website at dor.wa.gov/;
- (b) Reference to RCW 82.08.050(7); or
- (c) Calling the department's telephone information center at 1-360-705-6705.

PART 4 - COMMON RETAIL SALES TAX AND USE TAX EXEMPTIONS

- (401) What common retail sales tax and use tax exemptions apply to the sale of medical products? This part of the rule provides a nonexhaustive list of retail sales tax and use tax exemptions available with respect to various medical products.
- (402) Sales of medical products pursuant to a prescription. Most retail sales tax exemptions available for sales of medical products require that the item is purchased under authority of a prescription.
- (a) What is a prescription? A "prescription" is an order, formula, or recipe issued in any form of oral, written, electronic, or other means of transmission by a duly licensed practitioner authorized by the laws of this state to prescribe. See RCW 82.08.0281. The specific requirements for a prescription may differ depending on the item exempted and the RCW chapter under which the person issuing the prescription is licensed. Close attention must be paid to the details given for each specific exemption explained in the following subsections of this rule.
- (b) No automatic exemption. A prescription does not automatically qualify a sale of a medical product for a sales tax or use tax exemption. Unless a specific exemption exists in statute for the sale or use of the item in question the item is not exempt, even with a prescription. For example, if a physician prescribes a regimen of exercise at the local fitness club, the mere issuance of the prescription does not qualify the sales of that service for a retail sales tax exemption because no such exemption exists in statute.
- (c) When medical procedures are prescribed. When a medical procedure is prescribed by a duly licensed practitioner authorized to prescribe the same, that overall prescription fulfills the prescription requirement (if any) for each eligible exempt item used in the procedure. For example, an orthopedic surgeon conducts joint replacement surgery for a patient's diseased joint. As part of that surgical procedure, prescription drugs and other eligible exempt items are used. The surgeon does not specifically issue a separate written prescription for each eligible exempt item. The surgeon's order for the surgical procedure and the oral directions provided by the surgeon during the procedure fulfill any prescription requirement for each eligible item used in an exempt manner during that procedure.
- (d) Dispensed pursuant to a prescription. The purchase of drugs to be dispensed in the diagnosis, cure, mitigation, treatment, or prevention of disease or to affect the structure or any function of the body, by hospitals or other persons licensed to prescribe such drugs, are considered dispensed pursuant to a prescription and therefore exempt, providing the buyer gives the seller an exemption certificate or other approved documentation as discussed in Part 3 of this rule.
- (403) Sales tax and use tax exemptions available with respect to various medical products.
- (a) Sales to a free hospital are exempt from sales tax and use ${\tt tax.}$ RCW 82.08.02795 and 82.12.02745 provide retail sales tax and use tax exemptions for items sold to and used by a "free hospital" when those items are reasonably necessary for the operation of, and provi-

sion of health care by a free hospital. For the purpose of these exemptions, "free hospital" is a hospital that does not charge patients for health care provided by the hospital.

- (b) Sales of drugs for human use can be exempt from retail sales tax and use tax when sold under the authority of a prescription. RCW 82.08.0281 and 82.12.0275 provide retail sales tax and use tax exemptions for drugs for human use dispensed or to be dispensed to patients, pursuant to a prescription. These exemptions apply to the distribution of "sample" prescription drugs provided free of charge to duly licensed practitioners authorized by the laws of this state to prescribe. For the exemptions to apply, the drug involved must be intended to interact with a specific patient through direct contact with that patient, whether applied internally or externally to the patient's body, or as part of a test conducted on a tissue sample taken from that patient. A seller is not required to collect sales tax when it obtains a properly completed exemption certificate indicating prescription drugs, intended for human use sold to medical practitioners, nursing homes, and hospitals, will be put to an exempt use under the authority of a prescription, captures the data elements described in subsection (304) of this rule, or otherwise meets the requirements of RCW 82.08.050(7). Otherwise, the retail sales tax must be collected. See Part 3 of this rule for information about exemption certificates and other approved documentation.
- (c) Sales of disposable devices used to deliver prescription drugs for human use. RCW 82.08.935 and 82.12.935 provide retail sales tax and use tax exemptions for disposable devices used to deliver drugs for human use, pursuant to a prescription.
- (i) What are disposable devices used to deliver drugs? "Disposable devices used to deliver drugs" include single-use items such as a single-use syringe, intravenous (IV) tubing, and IV catheters. A stand or device that holds the tubing or catheter is not a disposable device used to deliver drugs.
- (ii) Example 8. Disposable devices. A nursing home purchases single-use syringes, tubing used to deliver drugs, and stands used to hold the IV fluid containers. If the nursing home provides the seller with a completed "Sales Tax Exemption Certificate for Health Care Providers," or other approved documentation, retail sales tax does not apply to the purchase of single-use syringes and tubing. However, retail sales tax applies to the IV stands because the stands are "durable medical equipment," not disposable or single-use, and no specific exemption for them exists in the law. For information about durable medical equipment, see Part 2 of this rule.
- (d) Sales of "over-the-counter" drugs with a prescription are exempt from retail sales tax and use tax. RCW 82.08.940 and 82.12.940 provide retail sales tax and use tax exemptions for over-the-counter drugs sold for human use, pursuant to a prescription. See subsection (205) of this rule for the definition of over-the-counter drug.
- (i) Example 9. A patient's medical practitioner prescribes overthe-counter pain relief medication. The patient takes the prescription to a pharmacy. The sale of the over-the-counter drug is exempt from retail sales tax. In contrast, if the patient's medical practitioner simply recommends that the patient use an over-the-counter pain relief medication, without completing a prescription for the medication, the sale of the over-the-counter drug is subject to retail sales tax.
- (ii) Example 10. A hospital makes bulk purchases of various overthe-counter drugs to dispense to patients pursuant to a doctor's prescription. The hospital's purchases of such drugs are exempt from re-

tail sales tax providing the hospital gives the seller an exemption certificate or other approved documentation as discussed in Part 3 of this rule.

- (iii) Example 11. An employer purchases drug test kits from a local drug store and administers them to current and prospective employees as a condition of employment. The employer's purchase of the drug tests is subject to retail sales tax because the tests are not prescribed by a licensed physician for the employees or prospective emplovees.
- (e) Dietary supplements (also known as nutrition products) with a prescription are exempt from retail sales and use taxes. Sales of dietary supplements not covered by either of the retail sales tax or use tax exemptions for "food and food ingredients" are generally subject to retail sales tax or use tax. See RCW 82.08.0293 and 82.12.0293. However, RCW 82.08.925 and 82.12.925 provide specific retail sales tax and use tax exemptions for sales of "dietary supplements" for human use, pursuant to a prescription. A "dietary supplement" is any product, other than tobacco, intended to supplement the diet, and that satisfies all three of the criteria listed in (e)(i) through (iii) of this subsection.
 - (i) Contains one or more of the following dietary ingredients:
 - (A) A vitamin;
 - (B) A mineral;
 - (C) An herb or other botanical;
 - (D) An amino acid;
- (E) A dietary substance for use by humans to supplement the diet by increasing the total dietary intake; or
- (F) A concentrate, metabolite, constituent, extract, or combination of any ingredient described in this subsection.
- (ii) Is intended for ingestion in tablet, capsule, powder, softgel, gelcap, or liquid form, or if not intended for ingestion in such form, is not represented as conventional food and is not represented for use as a sole item of a meal or of the diet; and
- (iii) Is required to be labeled as a dietary supplement, identifiable by the "supplement facts" box found on the label as required pursuant to 21 C.F.R. Sec. 101.36, as amended or renumbered as of January 1, 2003. See RCW 82.08.0293.
- (f) Licensed naturopaths have their own retail sales tax and use tax exemptions available. The sale or use of medicines of mineral, animal, and botanical origin which are prescribed, administered, dispensed, or used by a licensed naturopath in the treatment of a human patient are exempt from retail sales and use taxes. See RCW 82.08.0283 and 82.12.0277.

"Naturopathic medicines" are vitamins, minerals, botanical medicines, homeopathic medicines, hormones, and those legend drugs and controlled substances consistent with naturopathic medical practice in accordance with rules established by the secretary of health. Controlled substances are limited to codeine and testosterone products that are contained in Schedules III, IV, and V in chapter 69.50 RCW. See RCW 18.36A.020.

(g) Drugs and devices used for family planning may be exempt. $\ensuremath{\text{RCW}}$ 82.08.0281 and 82.12.0275 provide sales tax and use tax exemptions for drugs and devices sold or used under certain conditions for family planning purposes. Family planning purposes include promoting, inhibiting, preventing, and determining of conception. This includes all single-patient use items, whether ingested, attached, or applied to persons for family planning purposes. Persons making tax-exempt sales

of these drugs and devices to medical practitioners, clinics, or hospitals must obtain an exemption certificate, capture the data elements described in subsection (304) of this rule, or otherwise meet the requirements of RCW 82.08.050(7) to substantiate the exempt nature of any sale, as discussed in Part 3 of this rule.

The purchase, sale, or use qualifies for exemption when either one of the following conditions exists:

- The drug or device is supplied by a family planning clinic that is under contract with the Washington state department of health to provide family planning services; or
- The family planning items are or will be dispensed to patients, pursuant to a prescription. Persons dispensing these items are required to obtain and maintain files of prescriptions to document the exempt nature of such sales.
- (h) Medically prescribed oxygen is exempt from retail sales tax and use tax. RCW 82.08.0283 provides a retail sales tax exemption for sales of medically prescribed oxygen for an individual prescribed by a person licensed under chapter 18.57 RCW (Osteopathy—Osteopathic medicine and surgery) or chapter 18.71 RCW (Physicians) for use in the medical treatment of that individual. A comparable use tax exemption is provided in RCW 82.12.0277. Persons making tax-exempt sales of these items must obtain an exemption certificate, capture the data elements described in subsection (304) of this rule, or otherwise meet the requirements of RCW 82.08.050(7) to substantiate the exempt nature of any sale as discussed in Part 3 of this rule.
- (i) What is medically prescribed oxygen? The exemption for "medically prescribed oxygen" is not limited to gaseous or liquid oxygen (chemical designation O^2). Medically prescribed oxygen is defined by RCW 82.08.0283 to include, among other things, oxygen concentrator systems, oxygen enricher systems, liquid oxygen systems, and gaseous, bottled oxygen systems. The primary use of the equipment must be for the generation or storage of medically prescribed oxygen (0^2) . These systems include regulators, cannulae, masks, and similar items used to deliver the oxygen to the individual from the tax-exempt oxygen generation or storage device.
- (ii) Accessories may not be exempt. Exempt medical oxygen systems are sometimes connected to the patient through taxable systems. The exemption for medically prescribed oxygen only applies to items up to the point the exempt oxygen system is connected to the taxable system. From that point of connection forward to the patient, masks, tubing, or other similar items remain part of the taxable system and are subject to retail sales tax.

(iii) Examples.

- (A) Example 12. A physician prescribes oxygen for a patient. The patient rents an oxygen concentrator system and a separate cart to transport the system. The prescribed oxygen concentrator system can be rented exempt of sales tax. However, the exemption for "medically prescribed oxygen" does not include a separate cart used to transport a tax-exempt system. For information about durable medical equipment, see Part 2 of this rule. If the oxygen concentrator system and cart are rented for one nonitemized price the rental may be a bundled transaction. See Part 5 of this rule for information on how tax applies to a bundled transaction.
- (B) Example 13. A physician prescribes a "continuous positive airway pressure (CPAP) " system for a patient diagnosed with a sleep apnea disorder. The CPAP system primarily supplies room air, under

pressure, to keep the patient's airway passages open and thereby prevent obstruction of airflow in and out of the lungs. As a result, the sale of the CPAP system is subject to retail sales tax because it is not a system that satisfies the statutory definition of "medically prescribed oxygen." Note: Certain CPAP systems, when designed to be entirely worn on the body, can qualify for exemption from retail sales tax as prosthetic devices. See Part 2 of this rule for more information.

- (C) **Example 14.** Assume the same facts for a CPAP system as provided in the previous example (h)(i)(B) of this subsection. In addition, the physician prescribes an oxygen trickle by which medical oxygen is provided to the patient from an oxygen tank through a tube attached to the mask of the CPAP system. The addition of an oxygen trickle does not change the purpose or taxability of any part of the CPAP system. The CPAP system does not generate or store oxygen and is not eligible for the exemption provided for medically prescribed oxygen. The oxygen, oxygen tank, and any tubing used to convey the oxygen is covered by the exemption for medically prescribed oxygen, but only up to the point that it attaches to the taxable CPAP system.
- (i) Insulin has its own specific exemption from retail sales tax and use tax - No prescription is required. RCW 82.08.985 and 82.12.985 provide specific sales tax and use tax exemptions for insulin for human use. A prescription is not required for the sale of insulin to be exempt from tax.
- (i) Sales of laboratory reagents and other diagnostic substances may be exempt from retail sales and use taxes, under the right circumstances. The definition of drug includes compounds, substances, or preparations (e.g., laboratory reagents and other diagnostic substances) used for the diagnosis of disease. Thus, sales of laboratory reagents and other diagnostic substances are not subject to retail sales tax when prescribed for an individual by a duly licensed practitioner and used to diagnose, cure, mitigate, treat, or prevent disease in humans. RCW 82.08.0281. A comparable use tax exemption is provided in RCW 82.12.0275. Laboratory reagents and diagnostic substances must physically interact with a specific patient's specimen to qualify for exemption. Persons making tax-exempt sales of these items must obtain an exemption certificate, capture the data elements described in subsection (304) of this rule, or otherwise meet the requirements of RCW 82.08.050(7) to substantiate the exempt nature of any sale as discussed in Part 3 of this rule.
- (i) What are laboratory reagents and other diagnostic substances? "Laboratory reagents and other diagnostic substances" are substances employed to produce a chemical reaction in order to detect, measure, or produce, other substances. To be a diagnostic substance, the application of the substance to a patient's specimen must result in identification of the characteristics of a particular disease.
- (ii) Laboratory reagents, other diagnostic substances or prepared media when sold in a container. Reagents, diagnostic substances, and prepared media often come prepared in a container (test tube, vial, cylinder, Petri dish, etc.) ready for use. It makes no difference to the taxability of the substance if it is sold with or without a container. The function of the substance determines its taxability. The term "prepared media" includes transport media if the resulting culture grown on the medium is used in performing diagnostic tests for specific patients.
- (iii) Laboratory reagents and other diagnostic substances. This subsection provides examples of laboratory reagents and other diagnos-

tic substances that may qualify for sales and use tax exemptions under RCW 82.08.0281 and 82.12.0275, provided all requirements for the exemptions are met. The following items are reagents or other diagnostic substances:

- (A) Stains, dyes, and decolorizers that react with and cause a change in a cellular tissue. The substances are used to stain the cell tissues in a manner that will mark or highlight certain portions of cells;
- (B) Decalcifying solution, dehydrating solution, and clearing agents that chemically react with the patient's specimen; and
- (C) Test strips impregnated with a reagent which, when applied to a patient's specimen, test for indicators of a disease.
- (iv) What substances are not reagents? Some substances are used solely for purposes of preparing specimens for examination and diagnosis or to facilitate examination of a specimen. Such substances do not themselves produce a chemical reaction resulting in the detection, measurement, or production of another substance. They merely facilitate or enable specimen testing and are not exempt under RCW 82.08.0281 or 82.12.0275. The following lists examples of substances and items which are not reagents:
- (A) Paraffin that is extracted from a tissue specimen without having chemically altered the cells;
- (B) Gelatin that is extracted out of the specimen before staining and leaves the cell structures unaffected;
 - (C) Electrodes;
 - (D) Tissue cassettes;
 - (E) Freezing medium;
 - (F) Liquid agar when used to gel patient specimens;
 - (G) Test tubes or cylinders that do not contain a reagent;
- (H) Plain slides and cover slips that are not coated with a re-
 - (I) Mounting medium to adhere the cover slip to the slide; and
 - (J) Acids and other solutions when used for cleaning purposes.
- $(\,{\bf v}\,)$ What about reagents and diagnostic substances that can be used in more than one way (multiple use substances)? Some reagents or other diagnostic substances have multiple uses, some of which may qualify for a sales or use tax exemption. Such substances are exempt only to the extent they are used as part of a test prescribed to diagnose disease in humans. For example, alcohol can be used either as a reagent (e.g., to react with a cellular tissue) or to clean counters, furniture, etc. Alcohol used as a cleaning agent is subject to retail sales or use tax. See Part 3 of this rule for quidance on when to apply retail sales tax to products with multiple uses, with both retail sales taxable and exempt uses being possible.
- (k) Sales of controls, calibrators, and standards used with laboratory test equipment are not exempt from retail sales and use taxes. The sales tax and use tax exemptions provided by RCW 82.08.0281 and 82.12.0275 do not apply to drugs (compounds, substances, or preparations) used as a control, calibrator or standard in conjunction with the test of patient specimens in a medical laboratory.
- (i) What are controls? A "control" is a material, solution, lyophilized (freeze-dried) preparation or pool of collected serum designed to be used in the process of quality control. Controls do not physically interact with a specific patient's specimen. The concentrations of the substances of interest in the control are known within limits determined during its preparation or before routine use. Con-

trols are generally used with each test of patient specimens to validate the accuracy of that particular test.

(ii) What are calibrators? A "calibrator" is a material, solution, or lyophilized (freeze-dried) preparation designed to be used in calibration of medical laboratory machines. The values or concentrations of substances of interest in the calibration material are known within limits determined during its preparation or before use. Calibrators are generally used at specified intervals such as every eight hours, at midnight, or at shift changes, in accordance with the machine manufacturer's requirements or the requirements of administering agencies to verify the accuracy of the machine.

Calibrators are subject to retail sales tax or use tax because they are used to diagnose problems with machines and they do not physically interact with a patient's specimen to diagnose disease.

(iii) What are standards? A "standard" is a reference material of fixed and known chemical composition capable of being prepared in an essentially pure form. Standard also includes any certified reference material generally accepted or officially recognized as the unique standard used to test and calibrate medical lab equipment. Standards are often used in the original setup of medical lab equipment.

A standard is subject to retail sales tax and use tax because it is used to test and calibrate equipment and does not physically interact with a patient's specimen.

- (1) Sales of human blood, tissue, organs, or body parts may be exempt from retail sales and use taxes - No prescription or exemption certificate is required. RCW 82.08.02806 provides a retail sales tax exemption for human blood, tissue, organs, bodies or body parts when used for medical research and quality control testing purposes. RCW 82.12.02748 provides a comparable use tax exemption.
- (i) Definitions of human blood, tissue, organs, or body parts. For the purposes of this exemption the following definitions apply:
- (A) "Blood" means human whole blood, plasma, blood derivatives, and related products (e.g., bone marrow).
- (B) "Tissue" includes human musculoskeletal tissue, musculoskeletal tissue derivatives, ligament tissue, skin tissue, heart valve tissue, human bone, and human eye tissue.
- (C) "Organs" or "body parts" means a part of a human body having a special function.
- (ii) Materials consisting of both human and animal components. Materials consisting of both human and animal components are not "human blood, tissue, organs, or body parts" and do not qualify for this exemption.
- (iii) Sales of spermatozoa. These retail sales tax and use tax exemptions do not apply to sales or purchases of spermatozoa (male reproductive cell).
- (m) (i) Durable medical and mobility enhancing equipment Retail sales tax or use tax applies in most cases. Retail sales tax or use tax applies to the sale or use of durable medical equipment and mobility enhancing equipment, unless a specific exemption applies. See subsections (202) and (204) of this rule for the definition of durable medical and mobility enhancing equipment.
- (ii) (A) Mobility enhancing equipment Complex needs patient exemption. Beginning on August 1, 2023, retail sales tax and use tax does not apply to the sale or use of mobility enhancing equipment when that equipment is purchased for or used by a complex needs patient. To qualify for this exemption the mobility enhancing equipment must meet the user's specific and unique medical, physical, or functional needs

and capacities for basic activities when medically necessary to prevent hospitalization or institutionalization of the complex needs patient.

- (B) For the purposes of this subsection (403) (m) (ii), "complex needs patient" means an individual with a diagnosis or medical condition that results in significant physical or functional needs and capacities.
- (C) This exemption includes repair service and replacement parts for mobility enhancing equipment.
- (D) To claim this exemption, the buyer must provide the seller with a retail sales tax exemption certificate. The seller must retain a copy of the certificate for the seller's files. Information about exemption certificates may be obtained by:
 - (I) Using the department's website at dor.wa.gov;
 - (II) Reference to RCW 82.08.050(7); or
- (III) Calling the department's telephone information center at 1-360-705-6705.
- (n) Sales of prosthetic devices may be exempt of retail sales and use taxes. RCW 82.08.0283 provides a retail sales tax exemption for sales of prosthetic devices prescribed, fitted, or furnished for an individual by a person licensed under the laws of this state to prescribe, fit, or furnish prosthetic devices. The exemption includes repair and replacement parts, as well as labor and services rendered in respect to repairing, cleaning, altering, or improving prosthetic devices. RCW 82.12.0277 provides a corresponding use tax exemption. Persons making tax-exempt sales of these prosthetic devices to medical practitioners, nursing homes, and hospitals, must obtain an exemption certificate, capture the data elements described in subsection (304) of this rule, or otherwise meet the requirements of RCW 82.08.050(7) to substantiate the exempt nature of any sale as described in Part 3 of this rule. See subsection (206) of this rule for the definition of prosthetic device.
- (o) Kidney dialysis devices are exempt of retail sales and use taxes with a prescription. RCW 82.08.945 provides a retail sales tax exemption for sales of kidney dialysis devices for human use pursuant to a prescription. The exemption also includes repair and replacement parts, as well as labor and services rendered in respect to repairing, cleaning, altering, or improving kidney dialysis devices. RCW 82.12.945 provides a comparable use tax exemption. For the purpose of this exemption, a "kidney dialysis device" is a device which physically performs the dialyzing or separating process on blood. Kidney dialysis device does not include other equipment or tools used in conjunction with a kidney dialysis device.
- **Example 15.** A kidney dialysis device is wired to a dedicated backup generator that exists only to service the dialysis device when the main source of power is interrupted or is unavailable. Under those conditions the dialysis process cannot be performed without the use of the generator to power the dialysis device. Even so, the generator does not perform the actual dialysis process on the patient's blood and is not a kidney dialysis device.
- (p) Nebulizers are exempt of retail sales and use taxes with a prescription. RCW 82.08.803 and 82.12.803 provide sales tax and use tax exemptions in the form of a refund for the sale or use of a nebulizer for human use pursuant to a prescription. A nebulizer is "a device, and not a building fixture, that converts a liquid medication into a mist so that it can be inhaled." The exemptions include repair

and replacement parts, as well as labor and services rendered in respect to repairing, cleaning, altering, or improving a nebulizer.

Under these exemptions, sellers must collect the tax on sales subject to these exemptions. To obtain a refund of tax paid, buyers must apply for a refund directly from the department by submitting a completed refund application form to the department and including the original sales receipt. Any buyer submitting an application for refund should refer to WAC 458-20-229 or use the department's website at dor.wa.gov/content/ContactUs.

(q) Ostomic items are exempt of retail sales and use taxes - No prescription is required. RCW 82.08.804 and 82.12.804 provide specific sales tax and use tax exemptions for ostomic items for colostomy, ileostomy, or urostomy patients. "Ostomic items" are disposable medical supplies used by colostomy, ileostomy, and urostomy patients and include bags, belts to hold up bags, tapes, tubes, adhesives, deodorants, soaps, jellies, creams, germicides, and related supplies. "Ostomic items" do not include undergarments, pads and shields to protect undergarments, sponges, or rubber sheets. A prescription is not required for the sale of ostomic items to be exempt from tax.

PART 5 - BUNDLED TRANSACTIONS

- (501) What is a bundled transaction? A "bundled transaction" is the retail sale of two or more products, except real property and services to real property, where:
 - The products are otherwise distinct and identifiable; and
 - The products are sold for one nonitemized price.

A bundled transaction does not include the sale of any products in which the sales price varies, or is negotiable, based on the selection by the buyer of the products included in the transaction.

- (a) How are bundled transactions generally taxed for retail sales tax purposes? A transaction is generally considered a bundled transaction subject to retail sales tax if more than 10 percent of the purchase price or sales price is attributable to retail sales taxable products. RCW 82.08.190 and 82.08.195.
- (b) Exception. A transaction which otherwise meets the definition of a "bundled transaction" is not a bundled transaction when both of the following are true:
- (i) The transaction includes food and food ingredients, drugs, durable medical equipment, mobility enhancing equipment, over-thecounter drugs, prosthetic devices, or medical supplies; and
- (ii) The seller's purchase price or sales price of the taxable tangible personal property is 50 percent or less of the total purchase price or sales price of the bundled tangible personal property. Sellers may not use a combination of the purchase price and sales price of the tangible personal property when making the 50 percent determination for a transaction.
- (502) How are kits (or trays) used for medical procedures taxed if they contain a combination of individually taxable and nontaxable items? Medical procedure kits are often purchased as a plastic-wrapped package that includes the various items needed to perform a particular medical procedure. A procedure kit can combine items that are either subject to retail sales tax or exempt from retail sales tax if sold separate from a kit or tray, as individual items. However, when a kit involves a bundled transaction sold for one nonitemized price, the sale of the entire kit is either subject to retail sales tax or exempt. This subsection explains how to determine whether a particular medical procedure kit is subject to or exempt from retail sales tax.

Persons making a tax-exempt sale of a kit must obtain an exemption certificate from the buyer that lists the general item types within the kit that are exempt as discussed in Part 3 of this rule, capture the data elements described in subsection (304) of this rule, or otherwise meet the requirements of RCW 82.08.050(7). If a particular item within a kit is only exempt pursuant to a prescription, the item (or the procedure in which the item is used) must be prescribed by a duly licensed practitioner authorized by the laws of this state to prescribe the same.

Example 16. A glucose testing kit is prescribed for a human patient. The kit includes a glucose meter, five sample test reagent strips, and a lancet. The glucose meter is durable medical equipment, has a purchase price of \$40.00, and is subject to retail sales tax when sold separately. (See Part 2 of this rule for more information concerning durable medical equipment.) The lancet is a single-use tool not covered by any exemption, has a purchase price of \$40.00, and is subject to retail sales tax when sold separately. In this case, the test reagent strips qualify as disposable drug delivery devices, have a purchase price of \$20.00, and are exempt from retail sales tax when sold separately pursuant to a prescription. The total purchase price of the kit is \$100.00.

To determine if the full purchase price of the kit is subject to retail sales tax, the purchase (or sales) price of the taxable components should be compared to the total purchase (or sales) price of the kit. If the taxable components exceed 50 percent of the price, the entire kit is subject to retail sales tax. In this case, the purchase price for both the glucose meter and lancet (\$40.00 + \$40.00 = \$80.00)are more than 50 percent of the total kit purchase price of \$100.00. Therefore, retail sales tax is due on the sale of the kit. But if the taxable components were 50 percent or less of the total kit purchase price, sales tax would not be due on the kit.

WSR 23-23-020 EXPEDITED RULES DEPARTMENT OF REVENUE

[Filed November 2, 2023, 4:06 p.m.]

Title of Rule and Other Identifying Information: WAC 458-29A-400 Leasehold excise tax—Exemptions.

Purpose of the Proposal and Its Anticipated Effects, Including Any Changes in Existing Rules: The department proposes to amend this rule to recognize provisions of SHB 1163 (2023) which added a new leasehold excise tax exemption for all leasehold interests in the public or entertainment areas, as well as some of the office areas, of a qualified arena that: (a) Has a seating capacity of more than 4,000; (b) is located on city-owned land; (c) is located in a city with a population of over $10\overline{0},000$; and (d) has a cost of constructing improvements to the arena that were 100 percent, incurred by private entities that were not reimbursed by the public owner. Taxpayers claiming an exemption for this type of arena must file a complete tax performance report as provided in RCW 82.32.534.

Reasons Supporting Proposal: To recognize provisions of SHB 1163

Statutory Authority for Adoption: RCW 82.29A.140.

Statute Being Implemented: RCW 82.29A.130.

Rule is not necessitated by federal law, federal or state court decision.

Name of Proponent: Department of revenue, governmental.

Name of Agency Personnel Responsible for Drafting: Darius Massoudi, 6400 Linderson Way S.W., Tumwater, WA 98501, 360-534-1572; Implementation and Enforcement: 6400 Linderson Way S.W., Tumwater, WA 98501.

This notice meets the following criteria to use the expedited adoption process for these rules:

Adopts or incorporates by reference without material change federal statutes or regulations, Washington state statutes, rules of other Washington state agencies, shoreline master programs other than those programs governing shorelines of statewide significance, or, as referenced by Washington state law, national consensus codes that generally establish industry standards, if the material adopted or incorporated regulates the same subject matter and conduct as the adopting or incorporating rule.

Explanation of the Reason the Agency Believes the Expedited Rule-Making Process is Appropriate: The expedited rule-making process is applicable to this rule update because the department is amending the language to reflect statutory requirements.

NOTICE

THIS RULE IS BEING PROPOSED UNDER AN EXPEDITED RULE-MAKING PROC-ESS THAT WILL ELIMINATE THE NEED FOR THE AGENCY TO HOLD PUBLIC HEAR-INGS, PREPARE A SMALL BUSINESS ECONOMIC IMPACT STATEMENT, OR PROVIDE RESPONSES TO THE CRITERIA FOR A SIGNIFICANT LEGISLATIVE RULE. IF YOU OBJECT TO THIS USE OF THE EXPEDITED RULE-MAKING PROCESS, YOU MUST EX-PRESS YOUR OBJECTIONS IN WRITING AND THEY MUST BE SENT TO Darius Massoudi, Department of Revenue, 6400 Linderson Way S.W., Tumwater, WA 98501, phone 360-534-1572, fax 360-534-1606, email DariusM@dor.wa.gov, AND RECEIVED BY January 22, 2024.

November 2, 2024

Atif Aziz Rules Coordinator

OTS-5021.1

AMENDATORY SECTION (Amending WSR 22-24-105, filed 12/6/22, effective 1/6/23)

WAC 458-29A-400 Leasehold excise tax—Exemptions. (1) Introduction.

- (a) This rule explains the exemptions from leasehold excise tax provided by RCW 82.29A.125, 82.29A.130, 82.29A.132, 82.29A.134, 82.29A.135, and 82.29A.136. To be exempt from the leasehold excise tax, the property subject to the leasehold interest must be used exclusively for the purposes for which the exemption is granted.
- (b) This rule also explains the expiration date for new tax preferences for the leasehold excise tax pursuant to the language found at RCW 82.32.805.
- (c) Rule examples. This rule includes a number of examples that identify a set of facts and then states a conclusion. The examples should be used only as a general guide. The department of revenue (department) will evaluate each case on its particular facts and circumstances and apply both this rule and other statutory and common law authority.
- (2) Definitions. For purposes of this rule, the following definitions apply:
- (a) "New tax preference" means a tax preference that initially takes effect after August 1, 2013, or a tax preference in effect as of August 1, 2013, that is expanded or extended after August 1, 2013, even if the expanding or extending legislative amendment includes any other changes to the tax preference.
- (b) "Tax preference" has the same meaning as in RCW 43.136.021 with respect to any state tax administered by the department, except does not include the Washington estate and transfer tax in chapter 83.100 RCW.
 - (3) Operating properties of a public utility.
- (a) All leasehold interests that are part of the operating properties of a public utility are exempt from leasehold excise tax if the leasehold interest is assessed and taxed as part of the operating property of a public utility under chapter 84.12 RCW.
- (b) Example. Assume ABC Railroad Company is a public utility. Tracks leased to ABC Railroad Company are exempt from leasehold excise tax because ABC Railroad Company is a public utility assessed and taxed under chapter 84.12 RCW and the tracks are part of the railroad's operating properties.
 - (4) Student housing at public and nonprofit schools and colleges.
- (a) All leasehold interests in facilities owned or used by a school, college, or university which leasehold provides housing to students are exempt from leasehold excise tax if the student housing is exempt from property tax under RCW 84.36.010 and 84.36.050.
- (b) Example. Assume State Public University leases a building to use as a dormitory for its students. The leasehold interest associated

with this building is exempt from the leasehold excise tax. This is because the dormitory is used to house State Public University's students.

- (5) Subsidized housing.
- (a) All leasehold interests of subsidized housing are exempt from leasehold excise tax if the property is owned in fee simple by the United States, the state of Washington or any of its political subdivisions, and residents of the housing are subject to specific income qualification requirements.
- (b) Example. Assume an apartment building and the property on which it is located is:
 - · Owned in fee simple by the state of Washington; and
- Used as subsidized housing for residents subject to income qualification requirements.

If the United States Department of Housing and Urban Development holds the leasehold interest on the property it is exempt from leasehold excise tax. This is because the property is owned in fee simple by the state of Washington, used for subsidized housing, and the residents are subject to income qualification requirements.

- (6) Nonprofit fair associations.
- (a) All leasehold interests used for fair purposes of a nonprofit fair association are exempt from leasehold excise tax if the fair association sponsors or conducts a fair or fairs supported by revenues collected under RCW 67.16.100 and allocated by the director of the department of agriculture. The property must be owned in fee simple by the United States, the state of Washington or any of its political subdivisions. However, if a nonprofit association subleases exempt property to a third party, the sublease is a taxable leasehold interest.
- (b) Example. Assume a leasehold interest held by Local Nonprofit Fair Association is exempt from leasehold excise tax. Local Nonprofit Fair Association subleases some of the buildings on the fairgrounds to private parties for storage during the winter. These subleases are subject to the leasehold excise tax.
 - (7) Public employee housing.
- (a) All leasehold interests in public property or property of a community center which is exempt from property tax used as a residence by an employee of the public owner or the owner of the community center which is exempt from property tax are exempt from leasehold excise tax if the employee is required to live on the public property or community center which is exempt from property tax as a condition of his or her employment. The "condition of employment" requirement is met only when the employee is required to accept the lodging in order to enable the employee to properly perform the duties of his or her employment. However, the "condition of employment" requirement can be met even if the employer does not compel an employee to reside in a publicly owned residence or residence owned by a community center which is exempt from property tax.
 - (b) **Examples**.
- (i) A park ranger employed by the National Park Service, an agency of the United States government, resides in a house furnished by the agency at a national park. The ranger is required to be on call 24 hours a day to respond to requests for assistance from park visitors staying at an adjacent overnight campground. The use of the house is exempt from leasehold excise tax because the lodging enables the ranger to properly perform her duties.

- (ii) An employee of the Washington department of fish and wildlife resides in a house furnished by the agency at a fish hatchery although, under the terms of a collective bargaining agreement, the agency may not compel the employee to live in the residence as a condition of employment. In exchange for receiving use of the housing provided by the agency, the employee is required to perform additional duties, including regularly monitoring certain equipment at the hatchery during nights and on weekends and escorting public visitors on tours of the hatchery on weekends. The use of the house is exempt from leasehold excise tax because the lodging enables the employee to properly perform the duties of his employment. The use is exempt even though the employee would continue to be employed by the agency if the additional duties were not performed and even though state employees of an equal job classification are not required to perform the additional duties.
- (iii) A professor employed by State University is given the choice of residing in university-owned campus housing free of charge or of residing elsewhere and receiving a cash allowance in addition to her regular salary. If she elects to reside in the campus housing free of charge, the value of the lodging furnished to the professor would be subject to leasehold excise tax because her residence on campus is not required for her to perform properly the duties of her employment.
 - (8) Interests held by enrolled Indians.
- (a) Leasehold interests held by enrolled Indians are exempt from leasehold excise tax if the lands are owned or held by any Indian or Indian tribe, and the fee ownership of the land is vested in or held in trust by the United States, unless the leasehold interests are subleased to a lessee which would not qualify under chapter 82.29A RCW, RCW 84.36.451 and 84.40.175 and the tax on the lessee is not preempted due to the balancing test (see WAC 458-20-192).
- (b) Any leasehold interest held by an enrolled Indian or a tribe, where the leasehold is located within the boundaries of an Indian reservation, on trust land, on Indian country, or is associated with the treaty fishery or some other treaty right, is not subject to leasehold excise tax.
- (c) Example. Assume an enrolled member of the Puyallup Tribe leases port land at which the member keeps his or her boat, and the boat is used in a treaty fishery. The leasehold interest is exempt from the leasehold tax. For more information on excise tax issues related to enrolled Indians, see WAC 458-20-192 (Indians-Indian coun-
 - (9) Leases on Indian lands to non-Indians.
- (a) Leasehold interests held by non-Indians (not otherwise exempt from tax due to the application of the balancing test described in WAC 458-20-192) in any real property of any Indian or Indian tribe, band, or community that is held in trust by the United States or subject to a restriction against alienation imposed by the United States are exempt from leasehold excise tax if the amount of contract rent paid is greater than or equal to 90 percent of fair market rental value. In determining whether the contract rent of such lands meets the required level of 90 percent of market value, the department will use the same criteria used to establish taxable rent under RCW 82.29A.020 (2)(q) and WAC 458-29A-200.
- (b) **Example.** Harry leases land held in trust by the United States for the Yakama Nation for the sum of \$900 per month. The fair market value for similar lands used for similar purposes is \$975 per month.

The lease is exempt from the leasehold excise tax because Harry pays at least 90 percent of the fair market value for the qualified lands. For more information on the preemption analysis and other tax issues related to Indians, see WAC 458-20-192.

- (10) Annual taxable rent is less than \$250.
- (a) Leasehold interests for which the taxable rent is less than \$250 per year are exempt from leasehold excise tax. For the purposes of this exemption, if the same lessee has a leasehold interest in two or more contiquous parcels of property owned by the same lessor, the taxable rent for each contiguous parcel will be combined and the combined taxable rent will determine whether the threshold established by this exemption has been met. To be considered contiguous, the parcels must be in closer proximity than merely within the boundaries of one piece of property. When determining the annual leasehold rent, the department will rely upon the actual substantive agreement between the parties. Rent payable pursuant to successive leases between the same parties for the same property within a 12-month period will be combined to determine annual rent; however, a single lease for a period of less than one year will not be projected on an annual basis.
 - (b) Examples.
- (i) The yacht club rents property from the Port of Bay City for its clubhouse and moorage. It also rents a parking stall for its commodore. The parking stall is separated from the clubhouse only by a common walkway. The parking stall lease is a part of the clubhouse lease because it is contiquous to the clubhouse, separated only by a necessary walkway.
- (ii) Ace Flying Club rents hangars, tie downs, and ramps from the Port of Desert City. It has separate leases for several parcels. The hangars are separated from the tie down space by a row of other hangars, each of which is leased to a different party. Common ramps and roadways also separate the club's hangars from its tie-downs. The hangars, because they are adjacent to one another, create a single leasehold interest. The tie downs are a separate taxable leasehold interest because they are not contiguous with the hangars used by Ace Flying Club.
- (iii) Grace leases a lot from the City of Flora, from which she sells crafts at different times throughout the year. She pays \$50 per month for the lot, and has a separate lease for each season during which she sells. She has one lease from May through September, and a separate lease for the time between Thanksgiving and Christmas, which might run 30 to 40 days, depending on the year. The leases will be combined for the purposes of determining the leasehold excise tax. They relate to the same piece of property, for the same activity by the same lessee, and occur within the same year.
- (iv) Elizabeth owns a Christmas tree farm. Every year she rents a small lot from the Port of Capital City, adjacent to its airport, to sell Christmas trees. She pays \$125 to the port to rent the lot for six weeks. It is the only time during the year that she rents the lot. Her lease is exempt from the leasehold excise tax, because it does not exceed \$250 per year in taxable rent.
- (11) Leases for a continuous period of less than 30 days. Leasehold interests that provide use and possession of public property or property of a community center which is exempt from property tax for a continuous period of less than 30 days are exempt from leasehold excise tax. In determining the duration of the lease, the department will rely upon the actual agreement and/or practice between the parties. If a single lessee is given successive leases or lease renewals

of the same property, the arrangement is considered a continuous use and possession of the property by the same lessee. A leasehold interest does not give use and possession for a period of less than 30 days based solely on the fact that the lessor has reserved the right to use the property or to allow third parties to use the property on an occasional, temporary basis.

(12) Month-to-month leases in residential units to be demolished or removed.

- (a) Leasehold interests in properties rented for residential purposes on a month-to-month basis pending destruction or removal for construction of a public highway or public building are exempt from the leasehold excise tax. Thus, if the state or other public entity has acquired private property for purposes of building or expanding a highway, or for the construction of public buildings at an airport, the capitol campus, or some other public facility, and the public entity rents the property for residential purposes on a month-to-month basis pending destruction or removal for construction, these leases do not create taxable leasehold interests. This exemption does not require evidence of imminent removal of the residential units; the term "pending" merely means "while awaiting." The exemption is based upon the purpose for which the public entity holds the units.
- (b) Example. State University has obtained capital development funding for the construction of new campus buildings, and has purchased a block of residential property adjacent to campus for the sole purpose of expansion. Jim leases these houses from State University pursuant to a month-to-month rental agreement and rents them to students. Construction of the new buildings is not scheduled to begin for two years. Jim is not subject to the leasehold excise tax, because State University is holding the residential properties for the sole purpose of expanding its facilities, and Jim is leasing them pending their certain, if not imminent, destruction.

(13) Public works contracts.

- (a) Leasehold interests in publicly owned real or personal property held by a contractor solely for the purpose of a public improvements contract or work to be executed under the public works statutes of Washington state or the United States are exempt from leasehold excise tax. To receive this exemption, the contracting parties must be the public owner of the property and the contractor that performs the work under the public works statutes.
- (b) Example. Assume Tinker Construction is a contractor performing work to construct a second deck on the Nisqually Bridge pursuant to a public works contract between the state of Washington and Tinker Construction. During construction of the second deck on the Nisqually Bridge any leasehold interest in real or personal property created for Tinker Construction solely for the purpose of performing the work necessary under the terms of the contract is exempt from leasehold excise tax.

(14) Correctional industries in state adult correctional facilities.

- (a) Leasehold interests for the use and possession of state adult correctional facilities for the operation of correctional industries under RCW 72.09.100 are exempt from leasehold excise tax.
 - (b) Examples.
- (i) Assume ABC Retail Company, a for-profit corporation, operates and manages a business within a state prison under an agreement between it and the department of corrections. ABC Retail Company is ex-

empt from leasehold excise tax for its use and possession of state property.

- (ii) Assume ABC Charitable Society, a nonprofit organization, operates and manages a business within a state prison under an agreement between it and the department of corrections. ABC Charitable Society is exempt from leasehold excise tax for its use and possession of state property.
 - (15) Camp facilities for persons with disabilities.
- (a) Leasehold interests in a camp facility are exempt from leasehold excise tax if the property is used to provide organized and supervised recreational activities for persons with disabilities of all ages, and for public recreational purposes, by a nonprofit organization, association, or corporation which would be exempt from property tax under RCW 84.36.030(1) if it owned the property.
- (b) **Example.** Assume a county park with camping facilities is leased to Charity Campgrounds, a nonprofit charitable organization that allows the property to be used by the general public for recreational activities throughout the year and as a camp for disabled persons for two weeks during the summer. Charity Campgrounds is exempt from leasehold excise tax because the nonprofit allows the property to be used by the general public for recreational activities throughout the year, and to be used as a camp for disabled persons for two weeks during the summer.
 - (16) Public or entertainment areas of certain baseball stadiums.
- (a) Leasehold interests in public or entertainment areas of a baseball stadium with natural turf and a retractable roof or canopy, located in a county with a population of over 1,000,000 people, with a seating capacity of over 40,000, and constructed on or after January 1, 1995, are exempt from leasehold excise tax.
- (b) "Public or entertainment areas" for the purposes of this subsection include ticket sales areas, ramps and stairs, lobbies and concourses, parking areas, concession areas, restaurants, hospitality and stadium club areas, kitchens or other work areas primarily servicing other public areas, public rest rooms, press and media areas, control booths, broadcast and production areas, retail sales areas, museum and exhibit areas, scoreboards or other public displays, storage areas, loading, staging, and servicing areas, seating areas and suites, the playing field, and any other areas to which the public has access or that are used for the production of the entertainment event or other public usage, and any other personal property used for such purposes. "Public or entertainment areas" does not include locker rooms or private offices used exclusively by the lessee.
- (17) Public or entertainment areas of certain football stadiums and exhibition centers. Leasehold interests in the public or entertainment areas of an open-air stadium suitable for national football league football and for Olympic and world cup soccer, with adjacent exhibition facilities, parking facilities, and other ancillary facilities constructed on or after January 1, 1998, are exempt from leasehold excise tax. For the purpose of this subsection, the term "public and entertainment areas" has the same meaning as set forth in subsection (16) of this rule.
- (18) Public facilities districts. All leasehold interests in public facilities districts, as provided in chapter 36.100 or 35.57 RCW are exempt from leasehold excise tax.
- (19) State route 16 corridor transportation systems. All leasehold interests in the state route number 16 corridor transportation

systems and facilities constructed and operated under chapter 47.46 RCW are exempt from leasehold excise tax. RCW 82.29A.132.

- (20) Sales/leasebacks by regional transit authorities. All leasehold interests in property of a regional transit authority or public corporation created under RCW 81.112.320 under an agreement under RCW 81.112.300 are exempt from leasehold excise tax. RCW 82.29A.134.
- (21) Interests consisting of 3,000 or more residential and recreational lots. All leasehold interests consisting of 3,000 or more residential and recreational lots that are or may be subleased for residential and recreational purposes are exempt from leasehold excise tax. Any combination of residential and recreational lots totaling at least 3,000 satisfies the requirement of this exemption. RCW 82.29A.136.
- (22) Historic sites owned by the United States government or municipal corporations. All leasehold interests in property listed on any federal or state register of historical sites are exempt from leasehold excise tax if the property is:
- (a) Owned by the United States government or a municipal corporation; and
- (b) Wholly contained within a designated national historic reserve under 16 U.S.C. Sec. 461.

(23) Amphitheaters.

- (a) All leasehold interests in the public or entertainment areas of an amphitheater are exempt from leasehold excise tax if a private entity is responsible for 100 percent of the cost of constructing the amphitheater which is not reimbursed by the public owner, both the public owner and the private lessee sponsor events at the facility on a regular basis, the lessee is responsible under the lease or agreement to operate and maintain the facility, and the amphitheater has a seating capacity of over 17,000 reserved and general admission seats and is in a county that had a population of over 350,000, but less than 425,000 when the amphitheater first opened to the public.
- (b) For the purposes of this subsection, "public or entertainment areas" include box offices or other ticket sales areas, entrance gates, ramps and stairs, lobbies and concourses, parking areas, concession areas, restaurants, hospitality areas, kitchens or other work areas primarily servicing other public or entertainment areas, public rest room areas, press and media areas, control booths, broadcast and production areas, retail sales areas, museum and exhibit areas, scoreboards or other public displays, storage areas, loading, staging, and servicing areas, seating areas including lawn seating areas and suites, stages, and any other areas to which the public has access or which are used for the production of the entertainment event or other public usage, and any other personal property used for these purposes. "Public or entertainment areas" do not include office areas used predominately by the lessee.

(24) Military housing.

- (a) All leasehold interests in real property used for the placement of housing that consists of military housing units and ancillary supporting facilities are exempt from leasehold excise tax if the property is situated on land owned in fee by the United States, is used for the housing of military personnel and their families, and is a development project awarded under the military housing privatization initiative of 1996, 10 U.S.C. Sec. 2885, as existing on June 12, 2008.
- (b) For the purposes of this subsection, "ancillary supporting facilities" means facilities related to military housing units, including facilities to provide or support elementary or secondary edu-

cation, child care centers, day care centers, child development centers, tot lots, community centers, housing offices, dining facilities, unit offices, and other similar facilities for the support of military housing.

- (25) Community colleges and technical colleges.
- (a) All leasehold interests in facilities owned or used by a community college or technical college are exempt from leasehold excise tax if the leasehold interest provides:
 - (i) Food services for students, faculty, and staff;
 - (ii) The operation of a bookstore on campus; or
- (iii) Maintenance, operational, or administrative services to the community college or technical college.
- (b) Provisions of RCW 82.32.805 and 82.32.808 do not apply to the exemption specified in this subsection.
 - (26) Anaerobic digesters.
- (a) Beginning July 1, 2018, all leasehold interests in buildings, machinery, equipment, and other personal property which are used primarily for the operation of an anaerobic digester, the land upon which this property is located, and land that is reasonably necessary in the operation of an anaerobic digester are exempt from leasehold taxes for a period of six years from the date on which the facility or the addition to the existing facility becomes operational.
- (b) Claims for the exemption described in (a) of this subsection must be filed with the department on the form Leasehold excise tax exemption to operate an anaerobic digester available at https:// dor.wa.gov. Once filed, the exemption is valid for six assessment years following the date on which the facility or the addition to the existing facility becomes operational and may not be renewed. The department must verify and approve claims as it determines to be justified and in accordance with this subsection. No claims may be filed after December 31, 2024.
- (c) For the purposes of this subsection, "anaerobic digester" means a facility that processes organic material into biogas and digestate using microorganisms in a decomposition process within a closed, oxygen-free container as well as the equipment necessary to process biogas or digestate produced by an anaerobic digester into marketable coproducts including, but not limited to, biogas conditioning, compression, nutrient recovery, and electrical generation equipment. See RCW 82.08.900.
- (27) Exemptions for public or entertainment areas of certain arenas. Leasehold interests in the public or entertainment areas of ((an arena)) the following two types of arenas are exempt from the leasehold excise tax ((if the arena has)):
- (a) An arena with a seating capacity of more than 2,000((, and is)); located on land owned by a city with a population over 200,000; and within a county with a population of less than 1,500,000. For the purposes of this ((subsection)) paragraph, the term "public or entertainment areas" has the same meaning as set forth in subsection (23) of this rule.
- (b) Beginning October 1, 2023, an arena with a seating capacity of more than 4,000; located on land owned by, and within, a city with a population over 100,000; and private entities were responsible for 100 percent of the cost of constructing improvements to the arena which were not reimbursed by the public owner. For the purposes of this paragraph, "public or entertainment areas" has the same meaning as set forth in subsection (23) of this rule, except that it also includes office areas used predominately by the lessee.

- (i) A taxpayer claiming an exemption for this type of arena must file a complete tax performance report as provided in RCW 82.32.534.
- (ii) This exemption does not apply to leasehold interests on or after October 1, 2033.
- (28) Certain facilities owned by the state parks and recreation commission. Beginning January 1, 2023, leasehold interests in facilities owned by the state parks and recreation commission that are listed on the national register of historic places or the Washington heritage register are exempt from leasehold excise tax. This exemption expires January 1, 2034.
 - (29) Electric vehicle infrastructure.
- (a) Until July 1, 2025, leasehold interests in public lands for the purpose of installing, maintaining, and operating electric vehicle infrastructure are exempt from leasehold excise tax.
- (b) For purposes of this subsection, the following definitions apply:
- (i) "Battery charging station" means an electrical component assembly or cluster of component assemblies designed specifically to charge batteries within electric vehicles, which meet or exceed any standards, codes, and regulations set forth by chapter 19.28 RCW and consistent with rules adopted under RCW 19.27.540.
- (ii) "Battery exchange station" means a fully automated facility that will enable an electric vehicle with a swappable battery to enter a drive lane and exchange the depleted battery with a fully charged battery through a fully automated process, which meets or exceeds any standards, codes, and regulations set forth by chapter 19.28 RCW and consistent with rules adopted under RCW 19.27.540.
- (iii) "Electric vehicle infrastructure" means structures, machinery, and equipment necessary and integral to support an electric vehicle, including battery charging stations, rapid charging stations, battery exchange stations, fueling stations that provide hydrogen for fuel cell electric vehicles, green electrolytic hydrogen production facilities, and renewable hydrogen production facilities. See RCW 82.29A.125.
- (iv) "Green electrolytic hydrogen" means hydrogen produced through electrolysis, and does not include hydrogen manufactured using steam reforming or any other conversion technology that produces hydrogen from a fossil fuel feedstock.
- (v) "Rapid charging station" means an industrial grade electrical outlet that allows for faster recharging of electric vehicle batteries through higher power levels, which meets or exceeds any standards, codes, and regulations set forth by chapter 19.28 RCW and consistent with rules adopted under RCW 19.27.540.
- (vi) "Renewable hydrogen" means hydrogen produced using renewable resources both as the source for hydrogen and the source for energy input into the production process.
- (vii) "Renewable resource" means: Water, wind, solar energy; geothermal energy; renewable natural gas; renewable hydrogen; wave, ocean, or tidal power; biodiesel fuel not derived from crops raised on land cleared from old growth or first growth forests; or biomass energy.
 - (30) Expiration date for new tax preferences.
- (a) RCW 82.29A.025 incorporates the language found at RCW 82.32.805 establishing the expiration date of new tax preferences for the leasehold excise tax.

- (i) Generally, every new tax preference expires on the first day of the calendar year that is subsequent to the calendar year that is 10 years from the effective date of the tax preference.
- (ii) A future legislative amendment that expands a tax preference does not extend the tax preference beyond the period provided in this subsection unless an extension is expressly and unambiguously stated in the legislative amendment.
- (b) This subsection does not apply if legislation creating a new tax preference includes an expiration date for the new tax preference.
- (c) This subsection does not apply to an existing tax preference that is amended to clarify an ambiguity or correct a technical inconsistency. Future enacted legislation intended to make such clarifications or corrections must explicitly indicate that intent.

WSR 23-23-050 EXPEDITED RULES DEPARTMENT OF REVENUE

[Filed November 7, 2023, 1:03 p.m.]

Title of Rule and Other Identifying Information: WAC 458-02-200 Business licensing service—Applications, licenses, renewals—Fees.

Purpose of the Proposal and Its Anticipated Effects, Including Any Changes in Existing Rules: SHB [HB] 1742 (2023) authorizes the department to waive a late fee assessed on a business for not renewing its business license timely. A business is eligible for the waiver if the business has timely renewed all business licenses and paid the applicable business license fees for 24 months immediately prior to the period covered by the renewal application for which the waiver is being requested. This rule is being updated to reflect the availability of the waiver.

Reasons Supporting Proposal: To recognize the provisions of SHB [HB] 1742 (2023).

Statutory Authority for Adoption: RCW 19.02.030.

Statute Being Implemented: RCW 19.02.085.

Rule is not necessitated by federal law, federal or state court decision.

Name of Proponent: Department of revenue, governmental.

Name of Agency Personnel Responsible for Drafting: Perry Stern, 6400 Linderson Way S.W., Tumwater, WA 98501, 360-534-1588; Implementation and Enforcement: John Ryser, 6400 Linderson Way S.W., Tumwater, WA 98501, 360-534-1605.

This notice meets the following criteria to use the expedited adoption process for these rules:

Adopts or incorporates by reference without material change federal statutes or regulations, Washington state statutes, rules of other Washington state agencies, shoreline master programs other than those programs governing shorelines of statewide significance, or, as referenced by Washington state law, national consensus codes that generally establish industry standards, if the material adopted or incorporated regulates the same subject matter and conduct as the adopting or incorporating rule.

Explanation of the Reason the Agency Believes the Expedited Rule-Making Process is Appropriate: The expedited rule-making process is applicable to this rule update because the department is amending the language to reflect statutory requirements.

NOTICE

THIS RULE IS BEING PROPOSED UNDER AN EXPEDITED RULE-MAKING PROC-ESS THAT WILL ELIMINATE THE NEED FOR THE AGENCY TO HOLD PUBLIC HEAR-INGS, PREPARE A SMALL BUSINESS ECONOMIC IMPACT STATEMENT, OR PROVIDE RESPONSES TO THE CRITERIA FOR A SIGNIFICANT LEGISLATIVE RULE. IF YOU OBJECT TO THIS USE OF THE EXPEDITED RULE-MAKING PROCESS, YOU MUST EX-PRESS YOUR OBJECTIONS IN WRITING AND THEY MUST BE SENT TO Perry Stern, Department of Revenue, 6400 Linderson Way S.W., Tumwater, WA 98501, phone 360-534-1588, fax 360-534-1606, email PerryS@dor.wa.gov, AND RE-CEIVED BY January 22, 2024.

> November 7, 2023 Atif Aziz Rules Coordinator

AMENDATORY SECTION (Amending WSR 22-24-055, filed 12/1/22, effective 1/1/23)

- WAC 458-02-200 Business licensing service—Applications, licenses, renewals—Fees. (1) Introduction. This rule provides information about business license application handling fees, renewal application handling fees, and late filing delinquency fees as described in chapter 19.02 RCW. Information about individual licenses may be obtained from the business licensing service (BLS) of the department of revenue (department) and is available online at dor.wa.gov.
- (2) **Definitions**. The definitions in RCW 19.02.020 apply to this
- (3) What fee do I need to pay when applying for or renewing a license? Individual license fees vary depending on the license(s) for which you are applying or renewing. The fee payable is the total amount of all applicable individual license fees, business license application handling fees, renewal application handling fees, late filing delinquency fees, and other penalty fees. The method of payment may result in additional charges for credit or debit card processing.
- (4) What does the department do with the fees? The department will distribute the fees received for individual licenses to the respective regulatory agencies. The application and renewal handling fees and the late filing delinquency fees support the operation of the BLS. Credit or debit card payment processing fees are charged and retained by a third-party payment processor.
- (5) When do I get my business license? A business license will not be issued until the total fees due are collected and all required information has been submitted. Some individual licenses require review and approval by the regulating authorities, and the business license will not be issued until the regulating authorities have approved them.
- (6) Can I get a refund? The business license application handling fee and renewal application handling fee collected under RCW 19.02.075 are not refundable. The late filing delinquency fee under RCW 19.02.085 may not be waived or refunded unless:
- (a) The department determines that the licensee failed to renew a license by the business license expiration date due to an undisputable error or failure by the department that caused the late filing; or
- (b) The licensee requests the waiver and has timely renewed all business licenses and paid the applicable business license fees for a period of 24 months immediately preceding the period covered by the renewal application for which the licensee is requesting the waiver.

When a license is denied or when an applicant withdraws an application, a refund of any other refundable portion of the total payment will be made in accordance with the applicable licensing laws.

(7) What are the fees? The business license application handling fee, renewal application handling fee, late renewal filing delinquency fee, and individual license fee amounts are as follows:

Type of fee:	Fee amount:
Business license application handling fee to open the first business location	
of a new business, or to reopen a closed business:	\$50.00

Type of fee:	Fee amount:
Business license application handling fee for an existing business adding a new business location or requesting a city's license endorsement for a nonresident business:	\$0
Business license application handling fee for any other purpose(s):	\$10.00
Business license renewal application handling fee:	\$5.00
Late renewal filing delinquency fee:	Up to \$150.00 per business location. See subsection (9)(b) of this rule.
Individual license fee:	Varies depending on type of license.

- (8) What should I do with my business license? The business license document must be displayed in a conspicuous place at the business location for which the license is issued.
 - (9) Do I need to renew my business license?
- (a) The various licenses endorsed and displayed on the business license may each have a requirement to be renewed periodically. The department may prorate the terms of individual licenses and associated fees as needed so that all requested licenses on the account are due for renewal at the same time.
- (b) Licenses requiring renewal must be renewed by the expiration date or the department will assess a delinquency fee. The delinquency fee is calculated according to RCW 19.02.085 and must be paid by the licensee before a business license is renewed. Other regulatory agencies may also assess delinquency fees and/or penalties for late renewal, and may cancel the individual licenses for nonrenewal. Reissuance of individual licenses canceled for nonrenewal may require the filing of a new business license application.

WSR 23-23-066 EXPEDITED RULES DEPARTMENT OF REVENUE

[Filed November 9, 2023, 9:10 a.m.]

Title of Rule and Other Identifying Information: WAC 458-20-24001 Sales and use tax deferral—Manufacturing and research/development activities in high unemployment counties-Applications filed after June 30, 2010.

Purpose of the Proposal and Its Anticipated Effects, Including Any Changes in Existing Rules: The department is updating WAC 458-20-18801 due to changes from recent legislation, SSB 5565 (2023). The changes include updates to RCW 82.60.020.

Reasons Supporting Proposal: The update is to conform the rule to SSB 5565, which passed during the 2023 legislative session.

Statutory Authority for Adoption: RCW 82.01.060, 82.32.300.

Statute Being Implemented: RCW 82.60.020.

Rule is not necessitated by federal law, federal or state court decision.

Name of Proponent: Department of revenue, governmental.

Name of Agency Personnel Responsible for Drafting: Ryan Becklean, 6400 Linderson Way S.W., Tumwater, WA, 360-534-1576; Implementation and Enforcement: John Ryser, 6400 Linderson Way S.W., Tumwater, WA, 360-534-1605.

This notice meets the following criteria to use the expedited adoption process for these rules:

Adopts or incorporates by reference without material change federal statutes or regulations, Washington state statutes, rules of other Washington state agencies, shoreline master programs other than those programs governing shorelines of statewide significance, or, as referenced by Washington state law, national consensus codes that generally establish industry standards, if the material adopted or incorporated regulates the same subject matter and conduct as the adopting or incorporating rule.

Corrects typographical errors, makes address or name changes, or clarifies language of a rule without changing its effect.

Explanation of the Reason the Agency Believes the Expedited Rule-Making Process is Appropriate: The expedited rule-making process is appropriate for this rule update because the department is incorporating changes resulting from 2023 legislation.

NOTICE

THIS RULE IS BEING PROPOSED UNDER AN EXPEDITED RULE-MAKING PROC-ESS THAT WILL ELIMINATE THE NEED FOR THE AGENCY TO HOLD PUBLIC HEAR-INGS, PREPARE A SMALL BUSINESS ECONOMIC IMPACT STATEMENT, OR PROVIDE RESPONSES TO THE CRITERIA FOR A SIGNIFICANT LEGISLATIVE RULE. IF YOU OBJECT TO THIS USE OF THE EXPEDITED RULE-MAKING PROCESS, YOU MUST EX-PRESS YOUR OBJECTIONS IN WRITING AND THEY MUST BE SENT TO Ryan Becklean, Department of Revenue, P.O. Box 47453, Olympia, WA 98504-7453, phone 360-534-1533, fax 360-534-1576, email RyanBe@dor.wa.gov, AND RE-CEIVED BY January 22, 2024.

> November 8, 2023 Atif Aziz Rules Coordinator

AMENDATORY SECTION (Amending WSR 23-14-002, filed 6/21/23, effective 7/22/23)

WAC 458-20-24001 Sales and use tax deferral—Manufacturing and research/development activities in high unemployment counties-Applications filed after June 30, 2010. (1) Introduction. Chapter 82.60 RCW established a limited sales and use tax deferral program. The purpose of the program is to promote economic stimulation, create new employment opportunities in distressed areas, and reduce poverty in certain distressed counties of the state. RCW 82.60.010.

(a) Deferral program. This deferral program applies to an eligible investment project for sales and use taxes imposed on the construction, expansion, or renovation of qualified buildings or acquisition of qualified machinery and equipment. The program requires the recipient of the deferral to maintain the manufacturing or research and development activity for an eight-year period.

This rule does not address specific requirements of RCW 82.08.02565 and 82.12.02565 that provide statewide sales and use tax exemptions for machinery and equipment used directly in a manufacturing operation. Repayment of tax deferred under chapter 82.60 RCW is not required, and interest and penalties under RCW 82.60.070 will not be imposed, on machinery and equipment that qualifies for exemption under RCW 82.08.02565 or 82.12.02565. For additional information on statewide sales and use tax exemptions for machinery and equipment refer to WAC 458-20-13601.

- (b) Program enacted. The legislature first enacted this program in 1985. It has since made major revisions to the program criteria, specifically to the definitions of "eligible area," "eligible investment project," and "qualified building." For applications made prior to July 1, 2010, see WAC 458-20-24001A.
- (c) Administration of employment and related programs. The employment security department and the department of commerce administer programs for high unemployment counties and job training and should be contacted directly for information concerning these programs.
- (d) Examples. Examples found in this rule identify a number of facts and then state a conclusion. These examples should be used only as a general guide. The tax results of other situations must be determined after a review of all facts and circumstances.
- (2) **Definitions.** For the purposes of this rule, the following definitions apply:
- (a) "Acquisition of machinery and equipment" means the machinery and equipment is under the dominion and control of the recipient or its agent.
- (b) "Applicant" means a person applying for a tax deferral under chapter 82.60 RCW.
- (c) "Certificate holder" means an applicant to whom a tax deferral certificate has been issued.
- (d) "Community empowerment zone (CEZ)" means an area meeting the requirements of RCW 43.31C.020 and officially designated as a CEZ by the director of the department of community, trade, and economic development.
- (e) "Date of application" means the date of the U.S. Post Office postmark, fax, or electronic transmittal, or when the application is

hand delivered to the department. The statute in effect on the "date of application" will determine the program criteria the applicant must satisfy.

- (f) "Department" means the department of revenue.
- (q) "Eligible area" means:
- (i) Beginning July 1, 2010, an eligible area is a county that has an unemployment rate, as determined by the employment security department, which is at least 20 percent above the state average for the three calendar years immediately preceding the year in which the list of qualifying counties is established or updated, as the case may be. RCW 82.60.020.

The department, with the assistance of the employment security department, established a list of qualifying counties effective July 1, 2010. RCW 82.60.120. The list of qualifying counties is effective for a 24-month period and must be updated by July 1st of the year that is two calendar years after the list was established or last updated, as the case may be; or

- (ii) A designated community empowerment zone approved under RCW 43.31C.020. RCW 82.60.049.
- (h) "Eligible investment project" means an investment project that is located, as of the date the <u>deferral</u> application ((required by $\overline{\text{RCW }82.60.030}$)) is received by the department, in an eligible area. "Eligible investment project" does not include any portion of an investment project undertaken by a light and power business as defined in RCW 82.16.010, other than that portion of a cogeneration project that is used to generate power for consumption within the manufacturing site where the cogeneration project is an integral part. It also does not include investment projects that have already received deferrals under chapter 82.60 RCW. RCW 82.60.020 and 82.60.049.
- (i) "Industrial fixture" means an item attached to a building or to land. Examples of "industrial fixtures" are fuel oil lines, boilers, craneways, and improvements to land such as concrete slabs.
- (j) "Initiation of construction" means the date that a building permit is issued under the building code adopted under RCW 19.27.031 for:
- (i) Construction of the qualified building, if the underlying ownership of the building vests exclusively with the person receiving the economic benefit of the deferral;
- (ii) Construction of the qualified building ((when the lessor pays)), if the economic benefits of the deferral are passed to a lessee as provided in subsection (3) of this rule; or
- (iii) Tenant improvements for a qualified building ((when the owner/lessor pays)), if the economic benefits of the deferral are passed to a lessee as provided in subsection (3) of this rule((; or
- (iv) Tenant improvements for a qualified building when the lessee pays and receives the benefit of the deferral)).

"Initiation of construction" does not include soil testing, site clearing and grading, site preparation, or any other related activities that are initiated before the issuance of a building permit for the construction of the foundation of the building.

If the investment project is a phased project, "initiation of construction" shall apply separately to each phase.

(k) "Investment project" means an investment in qualified buildings or qualified machinery and equipment, including labor and services rendered in the planning, installation, and construction of the project.

(1) "Manufacturing" has the meaning given in RCW 82.04.120. Manufacturing, in addition, includes the activities performed by research and development laboratories and commercial testing laboratories, and the conditioning of vegetable seeds.

For purposes of this rule, both manufacturers and processors for hire may qualify for the deferral program as being engaged in manufacturing activities. For additional information on processors for hire, refer to WAC 458-20-136.

For purposes of this rule, "vegetable seeds" include the seeds of those crops that are grown in gardens and on truck farms and are generally known and sold under the name of vegetable or herb seeds in this state. "Vegetable seeds" include, but are not limited to, cabbage seeds, carrot seeds, onion seeds, tomato seeds, and spinach seeds. Vegetable seeds do not include grain seeds, cereal seeds, fruit seeds, flower seeds, tree seeds, and other similar properties.

- (m) "Office" means space used by professional, clerical, or administrative staff. For plant office space to be a qualified building its use must be essential or integral to the manufacturing or research and development operation. Office space that is used by supervisors and their staff, by technicians, by payroll staff, by the safety officer, and by the training staff are examples of qualifying office space. An office may be located in a separate building from the building used for manufacturing or research and development activities, but the office must be located at the same site as the qualified building to qualify. Each individual office may qualify or disqualify only in its entirety.
- (n) "Operationally complete" means the project is capable of being used for its intended purpose as described in the application.
- (o) "Person" has the meaning given in RCW 82.04.030. "Person" does not include the state of Washington or its institutions. "Person" may be either a lessee or a lessor/owner, who can apply separately for individual investment projects at the same site, if they comply with the other requirements of chapter 82.60 RCW.
- (p) "Qualified buildings" means construction of new structures, and expansion or renovation of existing structures for the purpose of increasing floor space or production capacity, used for manufacturing or research and development activities. "Qualified buildings" includes plant offices and warehouses or other facilities for the storage of raw material or finished goods if such facilities are an essential or an integral part of a factory, mill, plant, or laboratory used for manufacturing or research and development. "Qualified buildings" include construction of:
- · Specialized sewerage pipes connected to a qualified building that are specifically designed and used exclusively for manufacturing or research and development; and
- Parking lots connected to or adjacent to the building if the parking lots are for the use of workers performing manufacturing or research and development in the building. Parking lots may be apportioned based on qualifying use.

"Qualified buildings" does not include construction of landscaping or most other work outside the building itself, even though the landscaping or other work outside the building may be required by the city or county government in order for the city or county to issue a permit for the construction of a building.

(q) "Qualified employment position" means a permanent full-time employee employed in the eligible investment project during the entire tax year. The "entire tax year" means the full-time position is filled for a period of 12 consecutive months. "Full-time" means at least 35 hours a week, 455 hours a quarter, or 1,820 hours a year.

- (r) "Qualified machinery and equipment" means all new industrial and research fixtures, equipment, and support facilities that are an integral and necessary part of a manufacturing or research and development operation. "Qualified machinery and equipment" also includes computers; desks; filing cabinets; photocopiers; printers; software; data processing equipment; laboratory equipment; manufacturing components such as belts, pulleys, shafts and moving parts; molds, tools and dies; operating structures; and all equipment used to control or operate machinery. It also includes machinery and equipment acquired under the terms of a lease by the recipient. "New" as used in this subsection means either new to the taxing jurisdiction of the state or new to the certificate holder.
- (s) "Qualifying county" means a county that has an unemployment rate, as determined by the employment security department, which is at least 20 percent above the state average for the three calendar years immediately preceding the year in which the list of qualifying counties is established or updated, as the case may be.
- (t) "Recipient" means a person receiving a tax deferral under this program.
- (u) "Research and development" means the development, refinement, testing, marketing, and commercialization of a product, service, or process before commercial sales have begun, but only when such activities are intended to ultimately result in the production of a new, different, or useful substance or article of tangible personal property for sale. For purposes of this rule, "commercial sales" excludes sales of prototypes or sales for market testing if the total gross receipts from such sales of the product, service, or process do not exceed \$1,000,000.
- (v) "Site" means one or more immediately adjacent parcels of real property. Adjacent parcels of real property separated only by a public road comprise a single site.
- (w) "Warehouse" means buildings or facilities used for the storage of raw materials or finished goods. A warehouse may be located in a separate building from the building used for manufacturing or research and development activities, but to qualify the warehouse must be located at the same site as the qualified building. Warehouse space may be apportioned based on qualifying use.
- (3) Who is eligible for the sales and use tax deferral program? A person engaged in manufacturing or research and development activity is eligible for this deferral program for its eligible investment project.
- (a) The lessor or owner of the qualified building is not eligible for deferral unless:
- (i) The underlying ownership of the buildings, machinery, and equipment vests exclusively in the same person; or
- (ii) The lessor, by written contract, has agreed to pass the economic benefit of the deferral to the lessee;
- (iii) The lessee that receives the economic benefit of the deferral agrees in writing with the department to complete the annual tax performance report required under RCW 82.60.070; and
- (iv) The economic benefit of the deferral passed to the lessee is no less than the amount of tax deferred by the lessor and is evidenced by written documentation of any type of payment, credit, or other financial arrangement between the lessor or owner of the qualified building and the lessee.

For example, the economic benefit of the deferral can be passed through to the lessee when evidenced in writing that the amounts paid to the lessor for construction of tenant improvements are reduced by the amount of the sales tax deferred. Another method of passing the economic benefit is if the lessee receives a credit for tenant improvements or other mechanism in the lease, equal to the amount of the sales tax deferred.

(b) The lessor of the qualified building who receives a letter of intent from a qualifying lessee may be eliqible for deferral, assuming that all other requirements of chapter 82.60 RCW are met. At the time of application, the lessor, or another qualifying lessee must provide to the department a letter of intent by the lessee to lease the qualified building and any other information to prove that the lessee will engage in qualified manufacturing or research and development once the building construction is complete. After the investment project is certified as operationally complete, the lessee must actually occupy the building as a lessee and engage in qualified manufacturing or research and development. Otherwise, deferred taxes will be immediately due from the lessor.

The following examples illustrate the application process with lessors and lessees.

Example 1. Prior to the initiation of construction, Owner/Lessor AA enters into an agreement with Lessee BB, a company engaged in qualified manufacturing or research and development. Under the agreement, AA will build a building to house BB's research and development activities, will apply for a tax deferral on construction of the building, will lease the building to BB, and will pass on the economic benefit in the amount of the deferral to BB. BB agrees in writing with the department to complete annual tax performance reports. AA applies for the deferral before the initiation of construction that is prior to the date the building permit is issued. AA is entitled to a deferral on building construction costs assuming all eligibility qualifications are met.

Example 2. The following example assumes no deferral on initial construction activity. After the building construction has begun, Lessee CC asks that certain tenant improvements be added to the building. Lessor DD and Lessee CC each agree to pay a portion of the cost of the improvements. DD agrees with CC in writing that DD will pass on the entire value of DD's portion of the tax deferral to CC, and CC agrees in writing with the department to complete annual tax performance reports. CC and DD each apply for a deferral on the costs of the tenant improvements they are legally responsible for before the date the building permit is issued for the tenant improvements. The department will approve both applications assuming all eligibility qualifications are met. While construction of the building was initiated before submission of the applications, tenant improvements on a building under construction are deemed to be the expansion or renovation of an existing structure. In addition, lessees are entitled to the deferral only if they are legally responsible and actually pay contractors for the improvements, rather than merely reimbursing lessors for the costs.

Example 3. After building construction has begun but before machinery or equipment has been acquired, Lessee EE applies for a deferral on machinery and equipment. The department will approve the application assuming all eligibility qualifications are met, and EE will be required to complete annual tax performance reports. Even though it is too late to apply for a deferral of tax on building costs, it is not too late to apply for a deferral for the machinery and equipment.

- (4) What if an investment project is located in an area that qualifies as a high unemployment county and as a CEZ? If an investment project is located in an area that qualifies under more than one type of eligible area, the department will automatically assign the project to the eliqible area that imposes the least burden on the taxpayer and with the greatest benefit to the taxpayer. If the applicant elects to be bound by the requirements of the other potential eligible area, the applicant must make a written statement to that effect.
- Example 4. On October 1, 2014, a city in a high unemployment county qualifies as a CEZ, and the high unemployment county is on the list as a qualifying county. The CEZ employment requirements are more restrictive than those for qualifying counties. The department will assign the project to the qualifying county designation unless the applicant elects in writing to be bound by the CEZ employment requirements. Refer to subsection (7) of this rule for more information on the application process.
- (5) When is apportionment of qualified buildings appropriate? The deferral is allowable only in respect to investment in the construction of a new building or the expansion or renovation of an existing building used in manufacturing or research and development. Where a building(s) is used partly for manufacturing or research and development and partly for purposes that do not qualify for deferral under this rule, apportionment is necessary.
- (a) What are the apportionment methods? The deferral is determined by one of the following two apportionment methods. The first method of apportionment is based on square footage and does not require tracking the costs of materials for the qualifying/nonqualifying areas of a building. The second method of apportionment tracks the costs of materials used in the qualifying/nonqualifying areas, and it is primarily used by those industries with specialized building requirements.
- (i) First method. The applicable tax deferral is determined by apportionment according to the ratio of the square footage of that portion of the building(s) directly used for manufacturing or research and development purposes bears to the square footage of the total building(s).

Apportionment formula:

Eligible square feet of building(s) = Percent of building eligible Total square feet of building(s)

Percent of building eligible x Total Project Costs = Eligible Costs. "Total Project Costs" is the cost of multipurpose buildings and other improvement costs associated with the deferral project. Machinery and equipment are not included in this calculation.

Eliqible Costs (as determined above) x Tax Rate = Eliqible Tax Deferred.

Example 5. A taxpayer is constructing a 10,000 square foot building, of which 8,000 square feet will be eligible for tax deferral. The cost of the project is \$1,000,000. The combined sales/use tax rate at this location is 9.2%.

8,000 qualifying square feet building is eligible

Based on the above apportionment formula, 80% of the building is eligible for deferral. By multiplying the qualifying percentage 80% by the cost of \$1,000,000 to determine eligible costs of \$800,000. Multiply the eligible cost of \$800,000 by the sales/use tax rate of 9.2% to determine a sales/use tax deferral of \$73,600.

- (ii) Second method. If the applicable tax deferral is not determined by the first method, it will be determined by calculating the cost of construction of qualifying/nonqualifying areas as follows:
- (A) Tax on the cost of construction of areas devoted solely to manufacturing or research and development may be deferred.
- (B) Tax on the cost of construction of areas not used at all for manufacturing or research and development may not be deferred.
- (C) Tax on the cost of construction of areas used in common for manufacturing or research and development and for other purposes, such as hallways, bathrooms, and conference rooms, may be deferred by apportioning the costs of construction on a square footage basis. The apportioned costs of construction eligible for deferral are established by using the ratio, expressed as a percentage, of the square feet of the construction, expansion, or renovation devoted to manufacturing or research and development, excluding areas used in common, to the total square feet of the construction, expansion, or renovation, excluding areas used in common. That percentage is applied to the cost of construction of the common areas to determine the costs of construction eligible for tax deferral. Expressed as a formula, apportionment of the common areas is determined by:

Square feet devoted to manufacturing or research and Percentage of development, excluding square feet common areas of common areas eligible for deferral Total square feet, excluding square feet of common areas

Example 6. Taxpayer is planning to build a 10,000 square foot building of which 7,000 square feet will be used for manufacturing and 1,000 square feet will be common area. The remaining portion of the building will not be eligible for any deferral. The cost of the project will be \$850,000 for the manufacturing area, \$260,000 for the common area, and \$140,000 for the remaining portion of the building, for a total cost of construction of \$1,250,000. The combined sales/use tax rate at this location is 8.8%.

> 7,000 square feet devoted to manufacturing, excluding square 78% of common feet of common areas areas eligible for deferral 9,000 total square feet, excluding square feet of common areas

Based on the apportionment formula: 78% of common area costs are eligible. Multiply the common area costs of \$260,000 by 78% to determine that \$202,800 of common area costs are eligible for deferral. Therefore the \$850,000 for the manufacturing portion of the building plus the \$202,800 for common areas total \$1,052,800 of eligible project costs. Multiply the eligible project costs of \$1,052,800 by the tax rate of 8.8% to determine a sales/use tax deferral of \$92,646.

- (b) Are qualified machinery and equipment subject to apportionment? Unlike buildings, machinery and equipment cannot be apportioned if used for both qualifying and nonqualifying purposes.
- (c) To what extent is leased equipment eligible for the deferral? The amount of tax deferral allowable for leased equipment is the amount of the consideration paid by the recipient to the lessor over

the initial term of the lease, excluding any period of extension or option to renew, up to the last date for repayment of the deferred taxes. After that date, the recipient must pay the appropriate sales taxes to the lessor for the remaining term of the lease.

- (6) Are there any hiring requirements for an investment project? There may or may not be a hiring requirement, depending on the location of the project.
- (a) High unemployment county. There are no hiring requirements for qualifying projects located in high unemployment counties.
- (b) Community empowerment zone (CEZ). There are hiring requirements for qualifying projects located in CEZs or in counties containing CEZs. The applicant applies for a deferral of investment that correlates to the estimated number of persons to be hired based on a formula. The applicant will create a position and hire at least one qualified employee for each \$750,000 of qualified investment in the project. Refer to subsection (7) of this rule for more information on the application process. The recipient must fill the positions with persons who at the time of hire are residents of the CEZ. The persons must be hired after the date the application is filed with the department. As used in this subsection, "resident" means the person makes his or her home in the CEZ or the county in which the zone is located. A mailing address alone is insufficient to establish that a person is a resident. For example, a "P.O. Box" is not a valid address as it does not establish residence at a physical location where the person actually lives. A street address would be an example of a valid address.

The department has instituted a geographic information system (GIS) to assist taxpayers in determining taxing jurisdiction boundaries, local tax rates, and a mapping and address lookup system to determine whether a specific address is within a CEZ. The system is available on the department's website at dor.wa.gov. A recipient must fill the qualified employment positions by the end of the calendar year following the year in which the project is certified as operationally complete and retain the positions during the entire tax year. Refer to subsection (12) of this rule for more information on certification of an investment project as operationally complete. If the recipient does not fill the qualified employment positions by the end of the second calendar year following the year in which the project is certified as operationally complete, all deferred taxes are immediately due.

- (7) What are the application and review processes? An application for sales and use tax deferral under this program must be made prior to the initiation of construction, prior to taking possession of machinery and equipment, and prior to the filling of qualified employment positions. Persons, applying after construction is initiated or finished or after taking possession of machinery and equipment, are not eligible for the program. When an application for sales and use tax deferral is timely submitted, costs incurred before the application date are allowable, if they otherwise qualify. Applications for persons subject to hiring requirements must include information regarding the estimated total project cost and the qualified employment positions.
- (a) How does a taxpayer obtain an application form? Application forms may be obtained from the department's website at dor.wa.gov, or by contacting the department at 360-705-6705. Applications approved by the department under chapter 82.60 RCW are not confidential and are subject to disclosure. RCW 82.60.100.

- (b) Will the department approve the deferral application? In considering whether to approve or deny an application for a deferral, the department will not approve an application for a project involving construction unless:
- (i) The construction will begin within one year from the date of the application; or
- (ii) The applicant shows proof that, if the construction will not begin within one year of application, there is a specific and active program to begin construction of the project within two years from the date of application. Proof may include, but is not limited to:
- (A) Affirmative action by the board of directors, governing body, or other responsible authority of the applicant toward an active program of construction;
 - (B) Itemized reasons for the proposed construction;
 - (C) Clearly established plans for financing the construction; or
 - (D) Building permits.
- Similarly, after an application has been granted, a deferral certificate is no longer valid and should not be used if construction has not begun within one year from the date of application or there is not a specific and active program to begin construction within two years from the date of application. However, the department will grant requests to extend the period for which the certificate is valid if the holder of the certificate can demonstrate that the delay in starting construction is due to circumstances beyond the certificate holder's control such as the acquisition of building permit(s). Refer to subsection (9) of this rule for more information on the use of tax deferral certificates.
- (c) When will the department notify approval or disapproval of the deferral application? The department will verify the information contained in the application and approve or disapprove the application within 60 days. If approved, the department will issue a tax deferral certificate. If disapproved, the department will notify the applicant as to the reason(s) for disapproval.
- (d) May an applicant request a review of department disapproval of the deferral application? The applicant may request administrative review of the department's disapproval of an application, within 30 days from the date of notice of the disallowance, pursuant to the provisions of WAC 458-20-100, Appeals. The filing of a petition for review with the department starts a review of departmental action.
- (8) What happens after the department approves the deferral application? The department will issue a sales and use tax deferral certificate for state and local sales and use taxes due under chapters 82.08, 82.12, and 82.14 RCW for an eligible investment project. The department will state on the certificate the amount of tax deferral the recipient is eligible for. Recipients must keep track of how much tax is deferred.
- (9) How should a tax deferral certificate be used? A tax deferral certificate issued under this program is for the use of the recipient for deferral of sales and use taxes due on each eliqible investment project. Deferral is limited only to investment in qualified buildings or qualified machinery and equipment as defined in this rule. Thus, sales and use taxes cannot be deferred on items that do not become part of the qualified buildings, machinery, or equipment. In addition, the deferral is not to be used to defer the taxes of the persons with whom the recipient does business, persons the recipient hires, or employees of the recipient.

The certificate holder must provide a copy of the tax deferral certificate to the seller at the time goods or services are purchased. The seller will be relieved of the responsibility for collecting sales or use tax upon presentation of the certificate. The seller must retain a copy of the certificate as part of its permanent records for a period of at least five years. A blanket certificate may be provided by the certificate holder and accepted by the seller covering all such purchases relative to the eligible project. The seller is liable for business and occupation tax on all tax deferral sales.

- (10) May an applicant apply for multiple deferrals at the same project location? The department may not issue a certificate for an investment project that has already received a deferral under chapter 82.60 RCW. For example, replacement machinery and equipment that replaces qualified machinery and equipment is not eligible for the deferral. In addition, if an existing building that received a deferral under chapter 82.60 RCW for the construction of the building is renovated, the renovation is not eligible for the deferral unless the original deferral project is closed and has no more deferral requirements.
- (a) If expansion is made from an existing building that has already received a deferral under chapter 82.60 RCW for the construction of the building, the expanded portion of the building may be eligible for the deferral. The expansion must be made for new square footage, either vertically or horizontally. Acquisition of machinery and equipment to be used in the expanded portion of the qualified building may also be eligible.
- (b) A certificate may be amended or a certificate issued for a new investment project at an existing facility if all eligibility requirements are met.
- (11) May an applicant or recipient amend an application or certificate? Applicants and recipients may make a written request to the special programs division to amend an application or certificate when the original estimates change.
- (a) Assuming the project continues to meet all eligibility requirement, grounds for requesting amendment include, but are not limi-
 - (i) The project will exceed the costs originally stated;
- (ii) The project will take more time to complete than originally stated;
- (iii) The original application is no longer accurate because of changes in the project;
- (iv) The project location changes (only applicable to machinery and equipment); and
 - (v) Transfer of ownership of the project.
- (b) An application may not be amended if the location of the qualified building changes. Taxes become immediately due if the project location changes after the application has been approved.
- (c) The department must rule on the request within 60 days. If the request is denied, the department must explain in writing the basis for the denial. An applicant or recipient may appeal a denial within 30 days under WAC 458-20-100, Appeals.
- (12) What are the processes for an investment project? An applicant must provide the department with the estimated cost of the investment project at the time the application is made. Following approval of the application and issuance of a tax deferral certificate, a certificate holder must notify the department, in writing, when the

value of the investment project reaches the estimated cost as stated on the tax deferral certificate.

- (a) What should a certificate holder do if its investment project reaches the estimated costs but the project is not yet operationally complete? If an investment project has reached its estimated costs and the project is not operationally complete, the certificate holder may request an amended certificate stating a revised amount on which the deferral taxes are requested along with an explanation for the increase in estimated costs. Requests must be mailed, emailed, or faxed to the department.
- (b) What should a certificate holder do if its investment project reaches the completion date but the project is not yet operationally complete? If an investment project has reached the completion date and the project is not operationally complete, the certificate holder may request an amended certificate stating a revised completion date along with an explanation for the new completion date. Requests must be mailed, emailed, or faxed to the department prior to the expiration date on the certificate.
- (c) What should a certificate holder do when its investment project is operationally complete? The certificate holder must notify the department in writing when the investment project is operationally complete. The project is operationally complete once it can be used for its intended purpose as described in the application. The department will certify the qualifying costs and the date when the project became operationally complete. The certificate holder of the deferral must maintain the manufacturing or research and development activity beginning the year the project is operationally complete and the following seven calendar years. It is important to remember that annual tax performance report reporting requirements begin the year following the operationally complete date, even though the audit certification may not be complete. For information on submitting annual tax performance reports, see subsection (13) of this rule.
- Example 7. Taxpayer estimated a project end date of June 2018, but the project was actually operationally complete in November 2017. Taxpayer must submit the 2017 annual tax performance report by May 31, 2018. Taxpayer is responsible for notifying the department when the project is operationally complete regardless of the estimated completion date. If the 2017 annual tax performance report is not submitted timely, taxpayer will be assessed 12.5% of the deferred sales/use tax for this project.
- Example 8. Taxpayer estimated a project end date of May 2017, but the project was actually not operationally complete until December 2017. Taxpayer must submit the 2017 annual tax performance report by May 31, 2018. Taxpayer is responsible for notifying the department when the project is operationally complete regardless of the estimated completion date. If the 2017 annual tax performance report is not submitted timely, taxpayer will be assessed 12.5% of the deferred sales/use tax for this project.
- (i) If all or any portion of the project does not qualify, the recipient must repay all or a proportional part of the deferred taxes. The department will notify the recipient of the amount due and the due date.
- (ii) The department must explain in writing the basis for not qualifying all or any portion of a project. The decision of the department to not qualify all or a portion of a project may be appealed under WAC 458-20-100, Appeals, within 30 days.

- (13) Is a recipient of a tax deferral required to submit annual tax performance report? RCW 82.32.534 requires each recipient of a tax deferral to complete an annual tax performance report, every year, by May 31st for eight years following the year in which the project is operationally complete, regardless if the department has audited the project. If the economic benefits of the deferral are passed to a lessee as provided in RCW 82.60.025, the lessee must agree in writing to complete the annual tax performance report and the applicant is not required to complete the annual tax performance report. If the annual tax performance report is not submitted by the due date, or any extension under RCW 82.32.590, the recipient of the tax deferral or lessee, if required to submit, will be billed 12.5% of the deferred tax amount. For example, the deferral project is operationally complete in 2017. The recipient is required to submit the 2017-2024 annual tax performance reports that are due by May 31, 2018-2025, respectively. For more information on the requirements to file annual tax performance reports refer to WAC 458-20-267.
- (14) Is a recipient of a tax deferral required to repay deferred taxes for reasons other than not submitting the annual tax performance report? Repayment of tax deferred under chapter 82.60 RCW is waived, as long as all eligibility requirements are met, except as provided in RCW 82.60.070 and this subsection (14).

The following describes the various circumstances under which repayment of the deferral may occur. Outstanding taxes are determined as of December 31st of each year by reference to the following table. No proration is allowed for completing a partial year of the deferral use requirement.

Repayment	Year	Percentage Deferred Tax V	
1	(Year operationally		0%
2			0%
3			0%
4			10%
5			15%
6			20%
7			25%
8			30%

Any action taken by the department to disqualify a recipient for tax deferral or assess interest will be subject to administrative review pursuant to the provisions of WAC 458-20-100, Appeals. The filing of a petition for review with the department starts a review of departmental action.

(a) Failure of investment project to satisfy general conditions. If based on the recipient's annual tax performance report or other information, including that submitted by the employment security department, the department finds that an investment project is not eligible for tax deferral for reasons other than failure to create the required number of qualified employment positions, including failure to continue qualifying activity, the department will declare the amount of deferred taxes outstanding to be immediately due. There is no proration of the amount owed under this subsection. No penalties or interest will be assessed on the deferred sales or use taxes; however, all other penalties and interest applicable to excise tax assessments may be assessed and imposed.

- (b) Failure of investment project to satisfy required employment positions conditions. If based on the recipient's annual tax performance report or other information, the department finds that an investment project has been operationally complete and has failed to create the required number of qualified employment positions the amount of taxes deferred will be immediately due. There is no proration of the amount owed under this subsection. No penalties or interest will be assessed on the deferred sales or use taxes; however, all other penalties and interest applicable to excise tax assessments may be assessed and imposed.
- (15) When will the tax deferral program expire? This tax deferral program is scheduled to expire July 1, 2020. No applications for deferral of taxes will be accepted after June 30, 2020. Businesses wishing to take advantage of this program are advised to apply to the department by April 30, 2020. While the department will make every effort to process applications in a timely manner, the department is allowed 60 days to review applications and issue deferral certificates. Applications received after April 30, 2020, may not be processed in time for the business to receive a deferral certificate and would not be eligible for the program. In addition, incomplete applications may be denied or not processed in time for the business to be issued a deferral certificate before July 1, 2020.
- (16) Is debt extinguishable because of insolvency or sale? Insolvency or other failure of the recipient does not extinguish the debt for deferred taxes nor will the sale, exchange, or other disposition of the recipient's business extinguish the debt for the deferred tax-
- (17) Does transfer of ownership terminate tax deferral? Transfer of ownership does not terminate the deferral. The deferral is transferred, subject to the successor meeting the eligibility requirements of chapter 82.60 RCW, for the remaining periods of the deferral. Any person who becomes a successor to such investment project is liable for the full amount of any unpaid, deferred taxes under the same terms and conditions as the original recipient of the deferral. For additional information on successorship or quitting business refer to WAC 458-20-216.

Any questions regarding the potential eligibility of deferrals to be transferred on the sale of a business, should be directed to the special programs division as provided for in subsection (7)(a) of this rule.

WSR 23-23-068 EXPEDITED RULES DEPARTMENT OF REVENUE

[Filed November 9, 2023, 9:52 a.m.]

Title of Rule and Other Identifying Information: WAC 458-20-15503 Digital products.

Purpose of the Proposal and Its Anticipated Effects, Including Any Changes in Existing Rules: The department is updating WAC 458-20-15503 due to changes from recent legislation, SSB 5565 (2023). The changes include repealing former RCW 82.12.02088, which was superseded by RCW 82.12.0208(7). This update amends the rule to incorporate reference to current definitions and citations.

Reasons Supporting Proposal: The update is to conform the rule to SSB 5565, which passed in the 2023 legislative session.

Statutory Authority for Adoption: RCW 82.02.060 and 82.12.0208. Statute Being Implemented: RCW 82.12.0208.

Rule is not necessitated by federal law, federal or state court decision.

Name of Proponent: Department of revenue, governmental.

Name of Agency Personnel Responsible for Drafting: Justin Morehouse, 6400 Linderson Way S.W., Tumwater, WA, 360-534-1533; Implementation and Enforcement: John Ryser, 6400 Linderson Way S.W., Tumwater, WA, 360-534-1605.

This notice meets the following criteria to use the expedited adoption process for these rules:

Adopts or incorporates by reference without material change federal statutes or regulations, Washington state statutes, rules of other Washington state agencies, shoreline master programs other than those programs governing shorelines of statewide significance, or, as referenced by Washington state law, national consensus codes that generally establish industry standards, if the material adopted or incorporated regulates the same subject matter and conduct as the adopting or incorporating rule.

Corrects typographical errors, makes address or name changes, or clarifies language of a rule without changing its effect.

Explanation of the Reason the Agency Believes the Expedited Rule-Making Process is Appropriate: The expedited rule-making process is appropriate for this rule update because the department is incorporating changes resulting from 2023 legislation.

NOTICE

THIS RULE IS BEING PROPOSED UNDER AN EXPEDITED RULE-MAKING PROC-ESS THAT WILL ELIMINATE THE NEED FOR THE AGENCY TO HOLD PUBLIC HEAR-INGS, PREPARE A SMALL BUSINESS ECONOMIC IMPACT STATEMENT, OR PROVIDE RESPONSES TO THE CRITERIA FOR A SIGNIFICANT LEGISLATIVE RULE. IF YOU OBJECT TO THIS USE OF THE EXPEDITED RULE-MAKING PROCESS, YOU MUST EX-PRESS YOUR OBJECTIONS IN WRITING AND THEY MUST BE SENT TO Justin Morehouse, Department of Revenue, P.O. Box 47453, Olympia, WA 98504-7453, phone 360-534-1533, fax 360-534-1606, email justinm@dor.wa.gov, AND RECEIVED BY January 22, 2024.

> November 9, 2023 Atif Aziz Rules Coordinator

AMENDATORY SECTION (Amending WSR 16-04-099, filed 2/2/16, effective 3/4/16)

WAC 458-20-15503 Digital products. This rule provides a structured approach for determining tax liability for digital products and digital codes. For purposes of this rule, a digital product includes digital goods or digital automated services, which are described in detail below. The sale or use of digital products and digital codes is generally subject to retail sales or use tax unless purchased for resale or some other exemption applies.

This rule is organized into six parts. Each part addresses a question or topic relevant to the determination of whether a person is selling or purchasing a digital product or digital code and, if so, what are the tax consequences that follow from such activity. In this respect this rule is intended to function similar to the decision tree provided in ETA 9003.2010.

- 1. Part 1: Are the products or services transferred electronically? If yes, go to Part 2.
- 2. Part 2: Does the product or service meet the general definitions of digital product or digital code? If yes, go to Part 3.
- 3. Part 3: Are there applicable exclusions from the general definitions of the digital product or digital code? If no, go to Part 4.
- 4. Part 4: Are the sales of the digital product or digital code sourced to Washington? If yes, go to Part 5.
- 5. Part 5: Are there applicable retail sales or use tax exemptions for the purchase or use of the digital product or digital code? If no, the transaction is likely taxable in Washington.
 - 6. Part 6: Miscellaneous provisions.

Examples included in this rule identify a number of facts and then state a general conclusion; they should be used only as a general guide. The tax consequences of all situations must be determined after a review of all the facts and circumstances. Additionally, each fact pattern in each example is self contained (e.g., "stands on its own") unless otherwise indicated by reference to another example. Examples concluding that sales tax applies to the transaction assume that no exclusions or exemptions apply, and the sale is sourced to Washington.

Part 1. Are the Products or Services Transferred Electronically?

- (101) Introduction. Products or services must be transferred electronically in order to be digital products. If a product is transferred by means of a tangible storage media (e.g., compact disc, magnetic tape, hard drive, etc.), it is not a digital product. Digital codes need not be transferred electronically in order to be digital codes, but may be obtained by any means, including tangible storage
- (102) Transferred electronically. Means the purchaser obtains the product by means other than tangible storage media. Generally, this means the product is transferred using the public internet, a private network, or some combination. However, it is not necessary that the product be delivered to the purchaser. As long as the purchaser may access the product, it will be considered to have been electronically transferred to the purchaser. For example, whether a digital movie is downloaded by the user or streamed by the user, it is considered to be "transferred electronically." Alternatively, the same movie purchased

on tangible media (e.g., DVD, etc.) is the purchase of tangible personal property and is not considered to be either the sale of a digital product or transferred electronically.

Part 2. Does the Product or Service Meet the General Definition of Digital Product or Digital Code?

- (201) Introduction. The term "digital product" means (1) digital goods and (2) digital automated services. Digital products transferred to an end user are generally subject to retail sales or use tax regardless of whether the purchaser's right of use is permanent, less than permanent (e.g., 24-hour period), or the purchaser is obligated to make continued payments as a condition of the sale (e.g., "subscriptions").
- (202) Digital goods. Means sounds, images, data, facts, or information, or any combination thereof, transferred electronically, with certain exclusions discussed in Part 3 of this rule. The term "digital goods" includes within it the specific term "specified digital products" (as required by the Streamline Sales and Use Tax Agreement). The sale of a digital good is generally subject to retail sales tax and retailing business and occupation (B&O) tax.
- (a) Specified digital products. Means electronically transferred digital audio-visual works, digital audio works, and digital books.
- (i) Digital audio works. These are products that result from the fixation of a series of musical, spoken, or other sounds. Digital audio works include ringtones, recorded or live music, readings of books or other written materials, speeches, and other sound recordings. For example, a music file in MP3 format accessed or downloaded through the internet is a digital audio work.
- (A) A "ringtone" is a digitized sound file that is downloaded onto a communication device (e.g., mobile phone) and may be used to alert the user to an incoming communication such as a call or text message.
- (B) A ringtone does not include "ring-back tones" or other digital audio files that are not stored on the purchaser's communication device. In other words a ring-back tone is not a "specified digital product." A ring-back tone may be a digital automated service or a digital good depending on the facts. See analysis for digital automated services in subsection (203) of this rule.
- (ii) Digital audio visual works. These products are a series of related images which, when shown in succession, impart an impression of motion, together with accompanying sounds, if any. Digital audio visual works include movies, music videos, videos of live events, and news and entertainment programs. For example, a movie downloaded or accessed via the internet is a digital audio visual work.
- (iii) Digital books. These are books in a digital format that are generally recognized in the ordinary and usual sense as books. A digital book does not include periodicals, magazines, newspapers, chat rooms, or weblogs. For example, a cookbook in a PDF format downloaded or accessed through the internet is a digital book.
- (b) Other digital goods. The following list illustrates the types of products that are also digital goods in addition to the subclass of "specified digital products" discussed above. This list is merely illustrative and not exhaustive:
- (i) A digital schematic of a lawnmower engine transferred electronically.
 - (ii) A digital car history report transferred electronically.
 - (iii) A digital picture transferred electronically.

- (iv) Digital periodicals or magazines transferred electronically.
- (v) A digital presentation that includes still photos and accompanying audio content transferred electronically.
- (c) Digital goods prior to July 26, 2009. The mere accessing or streaming of a digital good was not a retail sale before July 26, 2009. Instead, accessing or streaming a digital good was subject to the service and other activities B&O tax. The sale of a digital good to a customer who downloaded the digital good was a retail sale. See Part 6, subsection (604) of this rule for a discussion of tax amnesty for past periods.
- (203) Digital automated services. Means services transferred electronically that use one or more software applications. The sale of a digital automated service is generally subject to retail sales tax and retailing B&O tax.
- (a) Digital automated services may include. One or more software applications either prewritten or custom, as well as components that are similar to stand-alone digital goods. For example, an online information service may contain data, facts, or information the use of which is facilitated by one or more software applications that provide search capabilities and other functionality. Thus, digital automated services will include software and may include elements similar to stand alone digital goods, which operate together in an integrated fashion to provide an electronically transferred service.
- Example 1. BFC provides an online service that facilitates apartment building management. The online service lists and advertises apartment vacancies, screens applicants, routes maintenance requests, and accepts and processes rental payments. In this example the software based service facilitates and automates various administrative functions and coordinates third-party services for apartment renting. The service is a digital automated service the sale of which is generally subject to retail sales tax and retailing B&O tax.
- Example 2. QPR provides a service that uses one or more software applications to "crawl the internet" in order to identify, gather, and categorize digital information according to specified criteria. In this example software facilitates the gathering, identifying and categorizing of information acquired from the internet. The service is a digital automated service the sale of which is generally subject to retail sales tax and retailing B&O tax.
- (i) Distinguishing a digital good from digital automated services. A digital good is not a service involving one or more software applications. A digital good consists solely of images, sounds, data, facts, information or any combination thereof. Clear examples of digital goods are digital books, digital music, digital video files, and raw data.
- Example 3. XYZ provides an online service that uses one or more software applications to facilitate the use of news and information with features such as: Research history, natural and boolean searching, industry chat forums, chart creation, document and word flagging, and information organizing folders. In this example software features facilitate the search of the news or information. XYZ's service is a digital automated service the sale of which is subject to retail sales tax and retailing B&O tax.
- Example 4. Company sells digital music files (i.e., digital goods) on its website. In order to locate specific digital music files customers may use a free software based search function that is integrated into Company's website. Customers may also find the digital music file they are seeking by clicking on a series of links to get to

the desired music file. Company's software based search function associated with the sale of the digital music file does not transform the sale of the digital music file into a digital automated service. Company is selling a digital good (i.e., music file) subject to retail sales tax and retailing B&O tax.

- (ii) Distinguishing remote access prewritten software from digital automated services. Remote access prewritten software (defined in RCW 82.04.050 (6)(b)) is solely prewritten software that is made remotely accessible from the vendor's server or other third-party server for a customer. To the extent that components similar to digital goods and/or additional services are supplied with the prewritten software the sale may be the sale of a digital automated service (see also Part 3, subsection (303)(h) of this rule).
- Example 5. CFC provides an online gaming service that allows subscribers to play a game with other subscribers in a real time multiplayer environment using software accessed via the internet. In this example the gaming software is combined with additional capabilities that enable a real time multiplayer environment that is not otherwise available. The service is a digital automated service, the sale of which is generally subject to retail sales tax and retailing B&O tax.
- Example 6. Company sells prewritten word processing software that is accessed by customers but hosted on Company's computers. The software includes access to clip-art image files that can be inserted into documents created with the remotely accessed prewritten word processing software. Company is selling remote access prewritten software and not a digital automated service or digital goods. The clip art made available with the software does not transform the remotely accessed prewritten software into a digital automated service or a digital good. Company is selling remote access prewritten software subject to retail sales tax and retailing B&O tax.
- (b) Digital automated service prior to July 26, 2009. The sale of a digital automated service to consumers was not a retail sale before July 26, 2009. Generally, income earned from such sales was subject to B&O tax under the service and other activities classification.
- (204) Digital codes. These are codes that provide a purchaser with the right to obtain one or more digital products, if all of the digital products to be obtained through the use of the code have the same retail sales and use tax treatment. A digital code may be obtained by any means, including email or by tangible media regardless of its designation as song code, video code, book code, or some other term. For example, a digital code includes the sale of an alphanumeric code that, when entered online at a website, provides the customer with a digital music file for download.
- (a) Products with mixed tax treatment. Codes that provide the right to obtain one or more products that do not have the same retail sales and use tax treatment are not digital codes.
- (b) Codes that represent a stored monetary value, redeemable cards, gift cards, or gift certificates. Codes that represent a stored monetary value that is deducted from a total as it is used by the purchaser or that represent a redeemable card, gift card, or gift certificate that entitles the holder to select digital products of an indicated cash value, are not digital codes.
- **Example 7.** Calvin purchases a code at his local grocery store for use on Joe Seller's (JS) website. At check out, Calvin tells the grocery store clerk to put \$25.00 in value on the plastic card containing the code. Calvin then goes to JS's website and inputs the code from the card. The \$25.00 value of the card is stored in Calvin's "account"

and can be used on any purchase by Calvin from JS's website. Calvin then purchases five digital songs for \$5.00 from JS. At check-out from JS's website, \$5.00 is deducted from Calvin's account to pay for the songs. When the transaction is complete, Calvin has a \$20.00 balance remaining in his account on JS's website. Because the code represents a stored monetary value it is not a digital code and the sale of the code is not subject to retail sales tax or retailing B&O tax.

Part 3. Are There Applicable Exclusions from the General Definitions of Digital Product and Digital Code?

- (301) Introduction. For certain products or services transferred electronically that otherwise meet the definition of digital good or digital automated service (as discussed in Part 2) there may be a specific exclusion from the applicable definition. If an exclusion applies, then the product or service will generally not be considered a digital good or digital automated service for retail sales and use tax purposes. For example, a service that is transferred electronically and that uses one or more software applications will generally be subject to retail sales tax as a digital automated service. However, if the service is an advertising service, then an exclusion applies, and the service will not be a digital automated service subject to retail sales tax; however, the service may still be subject to B&O tax. An excluded service may also still be subject to retail sales tax under certain circumstances. For example, telecommunications services are excluded from the definition of digital automated services, but remain subject to retail sales tax under their own separate definition of retail sale.
 - (302) Exclusions from the definition of digital good are:
- (a) Telecommunications and ancillary services as defined in RCW 82.04.065. These services may be used to distribute digital goods, digital automated services, and digital codes, but are not themselves any of these products.
- (b) Computer software as defined in RCW 82.04.215 and WAC 458-20-15502. These are coded instructions designed to cause a computer or automatic data processing equipment to perform a task.
 - (c) The internet and internet access as defined in RCW 82.04.297.
- (d) Professional or personal services represented in electronic form are not a digital good. This exclusion applies where the service primarily involves the application of human effort by the service provider, and the human effort originated after the customer requested the service. For example, an electronic engineering report created at the customer's request that reflects an engineer's professional analysis, calculations, and judgment, which is sent to the customer electronically, is considered evidence of a professional service and not a digital good.

Photography. This exclusion for professional or personal services does not apply to photographers in respect to amounts received for the taking of digital photographs that are transferred electronically to the end user/customer as defined in RCW 82.04.190(11). See Example 39 for an example of a nonend user transaction involving photography that is subject to royalties B&O tax.

- (e) Exclusions listed directly below for digital automated services are also exclusions from the definition of digital good.
- (303) Exclusions from the definition of digital automated service
- (a) Services that require primarily human effort by the seller and the human effort originated after the customer requested the serv-

ice. In this context, "primarily" means greater than ((fifty)) 50 percent of the effort to perform the service involved human labor. To determine whether the ((fifty)) 50 percent or greater threshold is satisfied, the average of the time and cost factors is considered. The time factor is determined by dividing the time spent to perform the human effort portion for customers by the total time spent performing the service. The cost factor is determined by dividing the direct costs incurred to perform the human effort portion for customers by the total direct costs incurred to perform the service. Direct costs of the human effort component include salaries, employee benefits and similar direct costs. Direct costs of the automated component include the cost of software, computers, hosting services and other similar direct costs. If the average of the time and cost factors is greater than ((fifty)) 50 percent then the service requires primarily human effort and is not a digital automated service in which case the service will generally be subject to service and other activities B&O tax.

Example 8. RepuCo.com performs a reputation monitoring service on the internet for its clients. The service utilizes software and other technology that searches the internet for websites that allow posting of information that may be harmful to RepuCo.com's client's reputation ("the automated component"). If the automated component finds a website that is posting erroneous or harmful information about one of RepuCo.com's clients, then a RepuCo.com employee will contact the owner of the website by phone or email and work with the owner and the client to resolve the matter to the satisfaction of the client ("the human effort component"). If the human effort time factor is 20% and the human effort direct cost factor is 60%, then the average of the two factors is 40% (80%/2 = 40%). Accordingly, the service is performed using 40% human effort which is less than 50% and therefore the service does not require primarily human effort and is subject to retail sales tax as a digital automated service.

Alternative methods. If the time and cost factors in this rule do not fairly represent the extent to which the service is performed using primarily human effort, the taxpayer may ask in writing for, or the department may require, the employment of another reasonable method to equitably determine whether the service is performed using primarily human effort.

- (b) Loaning or transferring money or the purchase, sale, or transfer of financial instruments. For purposes of this rule, "financial instruments" include cash, accounts receivable and payable, loans and notes receivable and payable, debt securities, equity securities, as well as derivative contracts such as forward contracts, swap contracts, and options. For example, the electronic transfer of money from a savings account to a checking account, whether done for the customer by a bank teller or by an ATM machine, is excluded from the definition of digital automated service.
- (c) Dispensing cash or other physical items from a machine. Includes an ATM that dispenses cash to users.
- (d) Payment processing services, including services such as electronic credit card processing activities conducted online or in physical retail stores via electronic transmission.
- (e) Parimutuel wagering and handicapping contests as authorized by chapter 67.16 RCW.
- (f) Telecommunications services and ancillary services as those terms are defined in RCW 82.04.065. For additional information, refer to the discussion above concerning the comparable exclusion from the

definition of digital goods (see Part 3, subsection (302)(a) of this rule).

- (q) The internet and internet access as those terms are defined in RCW 82.04.297.
- (h) Remote access prewritten software. Remote access prewritten software (defined in RCW 82.04.050 (6)(b)) provided on a standalone basis is excluded from the definition of digital automated service. However, software that is used in connection with a service that is transferred electronically would generally be included in the definition of a digital automated service.

Example 9. Company sells prewritten gaming software that is identical in all substantive respects to the same software available in stores for individual use and installation on home computers except that it is hosted on Company's servers and accessed by customers. Company's sales to consumers would be treated as a sale of remote access prewritten software and therefore is excluded from the definition of digital automated services and generally subject to retail sales tax and retailing B&O tax.

Example 10. Same facts as Example 9 except that Company uses the remote access prewritten software to provide a monthly subscription service that provides a real item multiplayer environment. Company is selling a digital automated service. In this case the customers are not merely receiving the individual use of software, but instead an online gaming service facilitated by the software. Thus, the monthly subscription service is not excluded from the definition of digital automated service and is subject to retail sales tax and retailing B&O

- (i) Online education programs provided by the following:
- (i) Public or private elementary or secondary schools; or
- (ii) An institution of higher education as defined in Sections 1001 or 1002 of the federal Higher Education Act of 1965 (Title 20 U.S.C. Sections 1001 and 1002), as existing on July 1, 2009. This would include most colleges and universities. For the purposes of this rule, an online educational program must be encompassed within the institution's accreditation.

Example 11. ABC University, a qualifying institution of higher education under the federal Higher Education Act of 1965, provides an accredited online Spanish course for which it charges a quarterly access and use fee to students. The course is remotely accessed by students logging into a website and accessing a fully interactive program that includes components of video, text, and audio, as well as extensive software code. This service would generally be considered a digital automated service. However, it is specifically excluded from the definition of digital automated service as an online educational program and may be subject to service and other activities B&O tax if another exclusion, deduction, or exemption does not apply.

(j) Live presentations such as lectures, seminars, workshops, or courses, where participants are connected to other participants and presenters via the internet or other networks, allowing the participants and the presenters to provide, receive, and discuss information together in real time.

Example 12. Company provides an online seminar service for Customer. Company provides a panel of live speakers that make a presentation to Customer's employees listening to and viewing the seminar through an internet connection supplied by a third-party service provider. The seminar allows Customer's employees and panelists to ask and answer questions on a real time basis. Company's online seminar

service is transferred electronically and uses one or more software applications and therefore would generally be considered a digital automated service. However, this type of service allowing live interaction is specifically excluded from the definition of digital automated service and would generally be subject to service and other activities B&O tax.

Example 13. Same facts as Example 12 except that Company records the seminar and charges other individuals a fee for accessing the seminar from Company's website. The recorded presentation allows these customers to watch the presentation but it does not allow them to ask questions on a real time basis. Because the presentation was prerecorded there is no live interaction contemporaneous with the presentation and therefore Company is selling a digital good generally subject to retail sales tax and retailing B&O tax.

Example 14. Company provides online training courses to Steve for a fee. The training courses provide key interactive elements such as study guides, knowledge testing, and automated help, all facilitated by one or more software applications. Such courses are not live presentations and do not provide human interaction. Accordingly, Company is selling a digital automated service generally subject to retail sales tax and retailing B&O tax.

- (k) Travel agent services, including online travel services, and automated systems used by travel agents to book reservations.
- (1) Online marketplace related activities, which are services that allow the person receiving the services to make online sales of products or services, digital or otherwise, using either:
 - (i) The service provider's website; or
- (ii) The service recipient's website, but only when the service provider's technology is used either to:
 - (A) Create or host the service recipient's website; or
- (B) Process orders from customers using the service recipient's website.

Example 15. Company provides an "electronic marketplace" service to Holcomb that allows Holcomb to list and sell his coffee mugs on the internet using Company's website. This online marketplace service is excluded from the definition of digital automated services and charges for the service would generally be subject to service and other activities B&O tax.

Example 16. Same facts as Example 15, except that now Holcomb decides he no longer wants to be just another seller on Company's website. Instead, Holcomb wants his own "retailing presence" on the internet so Holcomb contracts with Company to create and host Holcomb's new coffee mug website, "HolcombsCoffeeWorld.com." This is still an online marketplace service that is excluded from the definition of digital automated services and charges for the service would generally be subject to service and other activities B&O tax.

- (iii) Exclusion limitation. The services described in this subsection do not include the underlying sale of the products or services, digital or otherwise, by the person receiving the service. For instance, in Examples 15 and 16, the sale by Holcomb of coffee mugs would still generally be subject to retail sales tax and retailing B&O tax as the sale of tangible personal property.
- (m) Advertising services means all services directly related to the creation, preparation, production, or the dissemination of advertisements. Advertising services include: Layout, art direction, graphic design, mechanical preparation, production supervision, placement, and rendering advice to a client concerning the best methods of adver-

tising that client's products or services. Advertising services also include online referrals, search engine marketing and lead generation optimization, web campaign planning, the acquisition of advertising space in the internet media, and the monitoring and evaluation of website traffic for purposes of determining the effectiveness of an advertising campaign. Advertising services do not include web hosting services and domain name registration.

Example 17. Company provides marketing services to customers wishing to promote their products using the internet. Amy sells widgets on the internet and hires Company to market her products. Company consults with Amy on her marketing needs and then creates a marketing plan for her business. Company also creates and distributes online banners, links, and targeted "email blasts" that promote Amy's business. All of the services provided by Company are advertising services excluded from the definition of digital automated services and would generally be subject to service and other activities B&O tax.

Example 18. RVP, Inc. creates "sponsored links" on its website that drive customer traffic to Amy's website. RVP is paid by Amy for each click on a sponsored link on RVP's website. The services provided by RVP are advertising services excluded from the definition of digital automated services and charges for such would generally be subject to service and other activities B&O tax.

(n) Storage, hosting, and back-up. The mere storage of digital products, digital codes, computer software, or master copies of software is excluded from the definition of digital automated services. This exclusion includes providing space on a server for web hosting or backing-up data or other information.

Example 19. Company charges Rowe a fee for 25 terabytes of storage space under its "basic storage service" offering. Company also charges Rowe an additional and optional fee for its "premium service" package offering, which involves services beyond mere storage. The "basic storage" services are mere storage services and excluded from the definition of digital automated services. These services would generally be subject to service and other activities B&O tax. However, the charges for the optional premium services are more than mere storage or hosting services. As such, the premium services are not excluded from the definition of digital automated services and would generally be subject to retail sales tax and retailing B&O tax.

(o) Data processing services means a primarily automated service provided to a business or other organization where the primary object of the service is the systematic performance of operations by the service provider on data supplied in whole or in part by the customer to: (i) Extract the required information in an appropriate form, or (ii) to convert the data to usable information. Data processing services include check processing, image processing, form processing, survey processing, payroll processing, claim processing, and similar activities. Data processing does not include remote access prewritten software used by the customer to process their own data.

Example 20. Bango Corp., in preparation for litigation, hires Company to use its automated technology to search Bango's computers and gather documents relevant to the lawsuit. Company's service also provides software tools that allow Bango to categorize, copy, store, and notate the gathered documents. Company's service is not data processing. The services performed primarily involve gathering data, and providing software tools that allow the customer to categorize, copy, store and notate documents in preparation for litigation. Accordingly, Company is selling a digital automated service generally subject to retail sales tax and retailing B&O tax.

Example 21: Company provides check processing services to Wallo Corp., a bank operating in Washington. Company accepts scanned checks provided by Wallo and then uses its software and technology to extract the check dollar amount, account number, and verify the check has been signed. Company then provides this extracted and reformatted data back to Wallo allowing it to reconcile its customer's accounts. Company provides data processing services which are excluded from the definition of digital automated services. These services would generally be subject to service and other activities B&O tax.

Example 22. Same facts as Example 21, except that Company accepts checks provided directly by Wallo's customers. Thus, check images come from both Wallo and Wallo's customers. The services provided by Company are still data processing services excluded from the definition of digital automated services even though the data does not come exclusively from Wallo. These services would generally be subject to service and other activities B&O tax.

Part 4. Are the Sales of the Digital Product or Digital Code Sourced to Washington?

- (401) Introduction. Once it is determined that a transaction involves the sale of a digital product or digital code, the sale must be sourced to Washington in order to be subject to Washington's retail sales tax and B&O tax. If the sale is sourced outside Washington it is not subject to Washington sales tax or B&O tax. Sales of digital products are sourced using the same statute that applies to other retail sales, RCW 82.32.730 as outlined below.
 - (402) Sourcing retail sales.
- (a) Business location. When a digital product or digital code is received by the buyer at a business location of the seller, the sale is sourced to that business location.
- Example 23. Frank goes to BigBox brick-and-mortar store in Washington and purchases a music file from an electronic kiosk in the store. Frank purchases and downloads the music file inside BigBox's store by connecting his digital music player to the kiosk in the store. The sale of the music file is sourced to BigBox's store location in Washington and is generally subject to retail sales tax and retailing B&O tax.
- (b) Place of receipt. If the first sourcing rule explained above in (a) of this subsection does not apply, the sale is sourced to the location where receipt takes place.
- (i) The digital product or digital code may be received by the buyer at the buyer's location or by the buyer's donee (e.g., a gift recipient) at the donee's location.
- (ii) In the context of digital products and digital codes, "receive" and "receipt" means: (A) Making first use of digital automated services; or (B) taking possession or making first use of digital goods or digital codes, whichever comes first.
- Example 24. Drogba Inc., located in Olympia, Washington, purchases a digital automated service generally subject to retail sales tax from Company. Drogba's employees access and make first use of the service at their computer workstations located in Olympia. Company knows that the digital automated service is received in Olympia and therefore will source the sale of the digital automated service to that location.

(c) Address in records. If the first two sourcing rules explained in (a) and (b) of this subsection do not apply, the sale is sourced to the location indicated by an address for the buyer that is available from the seller's business records maintained in the ordinary course of business, so long as use of this address does not constitute bad faith. For example, any address of the buyer held by the seller that reasonably estimates the receipt location will be sufficient, including an address contained in a relevant service contract or an address used for accounts receivable purpose.

Example 25. Nani Corp., located in California, purchases a digital automated service generally subject to retail sales tax and retailing B&O tax from Company located in Washington. The purchase contract between Nani and Company provides that Nani may have 5 users access the digital automated service. Company does not know where the digital automated service is actually received. However, Company has Nani's California address in its business records and will therefore source the sale to Nani's California address. Because the sale is sourced outside Washington, it is not subject to Washington's retail sales tax or retailing B&O tax. Note, to the extent that Nani Corp., receives the service at locations in Washington, it may have a use tax liability. See subsection (403) of this rule for more on use tax.

(d) Address obtained during sale. If the first three sourcing rules explained in (a), (b), and (c) of this subsection do not apply, the sale is sourced to the location indicated by an address for the buyer obtained during the consummation of the sale. For example, an address obtained during consummation of the sale would include the address of a buyer's payment instrument (e.g., billing address for a credit card), if no other address is available, so long as use of this address does not constitute bad faith.

Internet protocol (IP) address. The buyer's IP address is acceptable location information obtained at the time of sale if an address cannot otherwise be obtained during consummation of the sale.

- (e) Origin. If the first four sourcing rules explained in (a), (b), (c), or (d) of this subsection do not apply, including the circumstance where the seller is without sufficient information to apply those provisions, then the sale must be sourced to the location determined by the address from which the digital good or digital code was first available for transmission by the seller, or from which the digital automated service was provided. Any location that merely provided the digital transfer of the product sold shall be disregarded.
- (403) Sourcing for use tax purposes. The sales sourcing rules above in subsection (402) of this rule are for sourcing sales subject to retail sales tax under RCW 82.08.020 and RCW 82.32.730. What follows below is a discussion of use tax reporting obligations with respect to digital goods, digital automated services, and digital codes. Generally, use tax applies to the use of a digital product or digital code in Washington if retail sales tax has not already been paid and no exemption otherwise applies.
- (a) Digital good or digital code. "Use" means the first act within this state by which the taxpayer, as a consumer, views, accesses, downloads, possesses, stores, opens, manipulates, or otherwise uses or enjoys the digital good or digital code.
- (b) Digital automated service. "Use" means the first act within this state by which the taxpayer, as a consumer, uses, enjoys, or otherwise receives the benefit of the service.

Example 26. Company, located in New York, sells a digital automated service generally subject to retail sales tax and retailing B&O

tax to Lampard Inc., located in Washington. Lampard's employees in Washington use the internet to access Company's services using an internet web browser. However, Company does not have nexus with Washington and is therefore not required to charge and collect retail sales tax on the sale of its service to Lampard. Lampard has a use tax reporting obligation because it uses, enjoys, or otherwise receives the benefit of Company's digital automated service at its location in Washington.

Part 5. Are there Applicable Retail Sales or Use Tax Exemptions for the Purchase or Use of the Digital Product or Digital Code?

- (501) Introduction. After determining that a digital product or digital code has been sold or used and the sale or use is sourced to Washington, exemptions from retail sales or use tax should be examined. What follows is not an exhaustive list of exemptions but instead an explanation of the most common exemptions for digital products. Some exemptions may apply only with respect to certain digital products (e.g., some exemptions apply only to digital goods, not digital automated services). Exemptions may also require an exemption certificate or reseller permit.
- (502) **Resale.** The purchase of a digital product or digital code for resale with no intervening use is not subject to retail sales or use tax. Sellers should obtain from buyers a copy of the buyer's reseller permit, a properly completed "Digital Products and Remote Access Software Exemption Certificate," or otherwise comply with RCW 82.04.470 to substantiate the wholesale nature of the sale. See RCW 82.32.780.
- (503) Component of a new product. Generally, purchasing, acquiring, owning, holding, or using any digital product or digital code for purposes of incorporating it into a new product for sale will not be subject to retail sales tax. The digital product or digital code must become a component of the new product for sale. A digital code becomes a component of a new product if the digital good or digital automated service acquired through the use of the digital code becomes incorporated into a new product. RCW 82.04.190(11). This is also discussed in subsection (602) of this rule in the context of wholesale sales.

Product. For purposes of this subsection, "product" means a digital product, an article of tangible personal property, or remote access prewritten software as defined in RCW 82.04.050 (6)(b). For example, an industrial drill manufacturer and seller combines hardware, software, and data to create a new product, a "smart drill." Software embedded in the drill uses the variance data (also embedded in the drill) to control the hardware during drill operations. The data is a digital good purchased for use as a component of a new product for sale (i.e., the drill). Sellers should obtain from buyers a copy of the buyer's reseller permit, a properly completed "Digital Products and Remote Access Software Exemption Certificate," or otherwise comply with RCW 82.04.470 to substantiate the wholesale nature of the sale.

(504) Made available free to the general public. Retail sales and use tax does not apply to the purchase or use by a business or other organization of a digital product (including a digital product acquired through the use of a digital code) in order to make that digital product (1) available free of charge for the use or enjoyment of (2) the general public. Buyers claiming this exemption must provide the seller with a properly completed "Digital Products and Remote Access Software Exemption Certificate" or other exemption certificate acceptable to the department. See RCW 82.08.02082.

(a) Available for free. In order to qualify, the digital product purchased must be made available for free. In this context, "free" means that the recipient of the digital product does not need to provide anything of significant value. If the purchaser requires something of significant value from the recipient in exchange for the digital product, it is not given away for free.

Example 27. Mauro purchases 1,000 digital music files from Company to be used for a "give away" to the first 1,000 people to visit Mauro's website. When people visit Mauro's website they are required to fill out a marketing survey before they may receive a digital music file. The information gathered from the marketing survey is then sold to a marketing company by Mauro. Thus, Mauro has required that recipients provide something of significant value in exchange for the digital music file. This is not a "free" transaction and therefore, Mauro's purchase of the digital music from Company does not qualify for the exemption and would be subject to retail sales tax and retailing B&O tax. (See also Example 29.)

- (b) "General public" means all persons and is not limited or restricted to a particular class of persons, except that the general public includes:
- (\mbox{i}) Certain classes of persons defined by their residency or property ownership. The general public includes a class of persons residing or owning property within the boundaries of any state (e.g., Washington), political subdivision of a state (e.g., King County), or a municipal corporation (e.g., Seattle).

Example 28. The City of Evergreen (a municipal corporation) makes satellite images of land parcels available for free only to persons residing in Evergreen. Residents are required to enter their zip code prior to accessing the images and certify that they are a resident of the City. Accordingly, the City of Evergreen can purchase the satellite images exempt from retail sales tax.

- (ii) Library customers. With respect to libraries, the term general public includes authorized library patrons.
- (c) Buyer must have the legal rights to provide the digital product to the general public. The exemption provided in this subsection does not apply unless the purchaser has the legal right to broadcast, rebroadcast, transmit, retransmit, license, relicense, distribute, redistribute, or exhibit the digital product, in whole or in part, to the general public.

Example 29. Same facts as Example 27, except this time visitors to Mauro's website are provided free access to the digital music files and no survey information is required in exchange. Additionally, Mauro purchased the digital music files from Company with the right to distribute them to the general public. Mauro also provided the seller with an exemption certificate. Accordingly, Mauro's purchase from Company qualifies for the exemption because he has made the digital audio files available free of charge to the general public pursuant to a contract that gives him rights of distribution. Mauro only purchased 1,000 files and therefore must limit the distribution to the first 1,000 people. Most "give-aways" will have similar quantity limitations but this fact alone will not disqualify such transactions under the "general public" requirement.

- (505) Purchased solely for business purpose.
- (a) Introduction. Retail sales and use tax does not apply to the sale to or use by a business of digital goods and services rendered in respect to those digital goods, where the digital goods and services rendered in respect to digital goods are purchased solely for business

purposes. This exemption only applies to purchases of digital goods and does not apply to the purchase of digital automated services, prewritten software, or remote access prewritten software. The exemption is only available when the buyer provides the seller with an exemption certificate. Buyers may use the department's "Digital Products and Remote Access Software Exemption Certificate" to claim this exemption. See RCW 82.08.02087.

- (b) Digital codes. This exemption also applies to the sale to or use by a business of a digital code if all of the digital goods to be obtained through the use of the code will be used solely for business purposes. If the digital code purchased by a business for a business purpose provides access to both digital goods and digital automated services, the purchase of the digital code does not qualify for this exemption.
- (c) "Business purposes" means the digital good is relevant to the buyer's business needs.
- (d) **Personal or household purpose.** This exemption does not apply to the purchase for personal or household purposes.
- (e) Government entities. This exemption does not apply to purchases by a governmental entity.
- (f) Prior periods. For the period July 26, 2009, through June 30, 2010, the "business purpose" exemption applied only to "standard digital information." Standard digital information is a subset of digital

Standard digital information is a digital good that consists primarily of data, facts, and/or information that is not generated or compiled for a specific client or customer. Standard digital information does not include a digital good that is comprised primarily of sounds or images.

- (506) Purchases of standard financial information by qualifying international investment management companies. ((Effective October 1, 2013_{r})) The purchase of standard financial information by a qualifying international investment management company, or persons affiliated with a qualifying international investment management company, is exempt from retail sales and use tax. RCW 82.08.207 and 82.12.207. The exemption applies regardless of whether the standard financial information is provided in a tangible format or on a tangible storage medium or as a digital product transferred electronically. This retail sales and use tax exemption expires July 1, $((\frac{2021}{}))$ 2031.
- $((\frac{a}{a}))$ "Qualifying international investment management company" means a person((÷
- (i) Who is primarily engaged in the business of providing investment management services; and
- (ii) Who has gross income that is at least ten percent derived from providing investment management services to:
- (A) Persons or collective investment funds residing outside the United States; or
- (B) Collective investment funds with at least ten percent of their investments located outside the United States.
- (b) The definitions in RCW 82.04.293 generally apply here to this subsection (506) except as follows:
- (i) Important distinction. This definition of "qualifying international investment management company" is more narrow than the definition in RCW 82.04.293; this definition in (a)(ii)(B) of this subsection excludes "persons" and only allows for "collective investment funds" unlike RCW 82.04.293 (1) (b) (ii) which includes "persons or collective investment funds" for B&O tax purposes.

- (ii))) engaging within this state in the business of providing qualifying international investment management services as defined in RCW 82.04.293(1).
- (a) "Standard financial information" means financial data, facts, or information, or financial information services, not generated, compiled, or developed only for a single customer. Standard financial information includes, but is not limited to, financial market data, bond ratings, credit ratings, and deposit, loan, or mortgage reports. RCW 82.08.207 (4)(b)(i).
- ((((iii)))) (b) "Financial market data" means market pricing information, such as for securities, commodities, and derivatives; corporate actions for publicly and privately traded companies, such as dividend schedules and reorganizations; corporate attributes, such as domicile, currencies used, and exchanges where shares are traded; and currency information. RCW 82.08.207 (4) (b) (ii).
- (((iv))) (c) **Filing and documentation**. Sellers making tax-exempt sales should obtain a completed buyer's retail sales tax exemption certificate from the buyer. In lieu of an exemption certificate, a seller may capture the relevant data elements as allowed under the streamlined sales and use tax agreement. The seller must retain a copy of the exemption certificate or other relevant data elements for the seller's files and must meet the records requirement in subsection (607) of this rule for tax liabilities owed to this state. For sellers who electronically file their taxes, the department will provide a separate tax reporting line for exemption amounts claimed under this rule.
- (((v))) (d) Limitations on exemption. A buyer may not continue to claim the exemption once the buyer has purchased standard financial information during the current calendar year with an aggregate total selling price in excess of ((fifteen million dollars)) \$15,000,000 and an exemption has been claimed for such standard financial information. The ((fifteen million dollar)) \$15,000,000 limitation under this subsection does not apply to any other exemption that applies to standard financial information.
- (((vi))) (e) **Sellers' responsibilities.** Sellers are not responsible for ensuring a buyer's compliance with the ((fifteen million dol- $\frac{1}{2}$)) $\frac{1}{2}$ 15,000,000 limitation under this subsection. Sellers may not be assessed for uncollected sales tax on a sale to a buyer claiming an exemption under this rule after having exceeded the ((fifteen million dollar)) \$15,000,000 limitation under this subsection, except as provided in RCW 82.08.050 (4) and (5).
- (((vii))) (f) Reporting requirements for buyers. This retail sales and use tax exemption for standard financial information is subject to additional reporting requirements. Buyers must report the amount of tax preference received as directed by the department. Buyers are not required to report the amount of preference received if the tax benefit to a buyer is less than ((one thousand dollars)) \$1,000 per year; or the buyer files an annual tax return with the department.
- (507) Multiple points of use (MPU). Retail sales tax does not apply to the sale of digital products or digital codes concurrently available for use within and outside this state. See RCW ((82.12.02088 and 82.08.02088)) 82.08.0208 and 82.12.0208. Note that Washington use tax still applies to the use of the digital product or digital code used in Washington. See RCW 82.12.0208.
- (a) Requirements. A buyer is entitled to claim the MPU exemption only if:

- (i) The buyer is a business or other organization.
- (ii) The digital product purchased (or obtained by using the digital code purchased) will be concurrently available for use within and outside this state (not for personal use).
- (iii) The buyer provides the seller with a valid exemption certificate acceptable to the department claiming the MPU exemption. Buyers may use the department's "Digital Products and Remote Access Software Exemption Certificate" to claim this exemption.
- (b) Concurrently available. "Concurrently available for use within and outside this state" means that employees or other agents of the buyer may use the digital product simultaneously from one or more locations within this state and one or more locations outside this
- **Example 30.** Company sells an online patent searching service to Iniesta Corp., for simultaneous use at Iniesta's headquarters in Washington and its research and development facility in California. This service would generally be considered the sale of a digital automated service subject to retail sales tax and retailing B&O tax. In this case, the digital automated service is concurrently available for use by Iniesta's employees both within Washington and outside Washington, and therefore Iniesta may claim the MPU exemption from retail sales tax for its purchase of the digital automated service from Company. See (c) of this subsection for an explanation of how to apportion the use tax in this example.
- (c) Apportionment (allocation) of use tax. For purposes of this subsection on multiple points of use, "allocation" and "apportionment" have the same meaning. A business or other organization subject to use tax on digital products or digital codes that are concurrently available for use within and outside this state is entitled to apportion the amount of tax due this state based on users in this state compared to users everywhere. For example, in the case of Iniesta in Example 30, if we assume Iniesta had five employees in California and five employees in Washington using the service concurrently, Iniesta would allocate one-half of the purchase price to Washington because five of its ((ten)) 10 users are in Washington (e.g., 5/10 = 50%). Thus Iniesta would pay use tax to Washington based on ((fifty)) 50 percent of the value of the digital automated service. Additionally, the department may authorize or require an alternative method of allocation supported by the taxpayer's records that fairly reflects the proportion of instate to out-of-state use by the taxpayer.
- (i) Records requirement. No allocation under this rule is allowed unless the allocation method is supported by the taxpayer's records kept in the ordinary course of business in compliance with the records requirement in subsection (607) of this rule for tax liabilities owed to this state.
- (ii) "User" means an employee or agent of the taxpayer who is authorized by the taxpayer to use the digital product purchased in the performance of his or her duties as an employee or other agent of the taxpayer.
- (d) Application to digital codes. A digital code is concurrently available for use within and outside this state if users may use the digital goods or digital automated services to be obtained by the code simultaneously at one or more locations within this state and one or more locations outside this state.
- (e) Reporting. A buyer claiming an exemption under this rule must report and pay state and local use tax directly to the department. As explained in (c) of this subsection, use tax may be reported and paid

on an apportioned basis if supported by the buyer's records in compliance with the records requirement in subsection (607) of this rule for tax liabilities owed to this state.

- (508) Machinery and equipment. Generally retail sales and use tax does not apply to sales to or use by a manufacturer or processor for hire of certain machinery and equipment used directly in a manufacturing or research and development operation. This exemption is commonly referred to as the M&E exemption. (See RCW 82.08.02565 and 82.12.02565 and WAC 458-20-13601 for information regarding the M&E exemption.) Included within the definition of "machinery and equipment" for purposes of the M&E exemption are digital goods. Accordingly, digital goods acquired by manufacturers and processors for hire and used directly in a manufacturing or research and development operation are exempt from retail sales and use tax, provided all of the requirements for the M&E exemption are met.
- (509) Audio or video programming. Income received from the sale of regular audio or video programming by a radio or television broadcaster is generally subject to service and other B&O tax and therefore not subject to retail sales tax. However, the sale of audio or video programming sold on a pay per program or subscription on-demand basis is generally subject to retail sales and use tax except as provided in (d) and (e) of this subsection.
- (a) "Radio and television broadcasters" include satellite radio providers, satellite television providers, cable television providers, and providers of subscription internet television.
- (b) "Pay per program or subscription on-demand basis" means programming that the buyer pays for on a per program basis or a service that allows the buyer to access a library of programs at any time for a specific charge.
- (c) "Regular programming" is scheduled programming. The person watching cannot stop, pause, rewind, or otherwise control the broadcast of the scheduled programming, including the time that the scheduled program is broadcast.

The fact that a customer uses a recording device, such as a VCR or DVR, does not result in the broadcaster's programming being characterized as a digital good.

- (d) Cable television providers paying franchise fees. Cable television providers' sales of programming to consumers on a pay-per-program or subscription on-demand basis are not subject to retail sales and use tax if the cable television provider is subject to a franchise fee (under the authority of Title 47 U.S.C. Sec. 542(a)) on the gross revenue received from such sales. If the cable television provider is not subject to a franchise fee on the income from the sale of programming on a pay-per-program or subscription on-demand basis, then the exemption does not apply and the cable television provider must collect and remit retail sales tax on the retail sale of such programming.
- Example 31. XYZ sells video programming to customers using cable technology. XYZ does not pay a franchise fee. Customers of XYZ are charged a monthly subscription fee to receive video programming. Customers are charged additional fees to view selected movies. XYZ must charge and collect retail sales tax on the additional fees charged to view the selected movies, but not on the monthly subscription fee which would generally be subject to service and other activities B&O
- (e) Satellite television providers do not generally pay franchise fees and therefore do not qualify for the retail sales and use tax ex-

emption based on payment of franchise fees as described in (d) of this subsection.

- (510) Newspapers. Generally, retail sales and use tax does not apply to sales of newspapers transferred electronically, provided that the electronic version has a printed counterpart, and the electronic version:
 - (a) Shares content with the printed newspaper; and
- (b) Is prominently identified by the same name as the printed newspaper or otherwise conspicuously indicates that it is a complement to the printed newspaper.
- (c) "Printed newspaper" means a publication issued regularly at stated intervals at least twice a month and printed on newsprint in tabloid or broadsheet format folded loosely together without stapling, glue, or any other binding of any kind, including any supplement of a printed newspaper.
- (511) Received for free by end user. Digital products and digital codes obtained by the end user for free are not subject to use tax.
- (a) For example, a person's use of a free search engine is not subject to use tax.
- (b) For example, a person reading an online article or viewing an online picture for free is not subject to use tax.
- (512) Other use tax exemptions. Use tax does not apply to the use of digital goods that are:
- (a) Noncommercial in nature, such as personal email communications;
 - (b) Created solely for an internal audience; or
- (c) Created solely for the business needs of the person who created the digital good and is not the type of digital good that is offered for sale, including business email communications.

Example 32. Gary, an employee of Kadabbera Corp., creates a digital audio-visual presentation using presentation authoring software and his innate creative capacity. Gary distributes the presentation internally to various divisions within Kadabbera in order to train employees on changes to company policies. Gary has created and distributed an item that meets the definition of "digital good." However, the distribution and use of this digital good is not subject to use tax as long as it is used solely internally or solely for the business needs of Kadabbera.

Part 6. Miscellaneous Provisions

(601) Retail services. Washington imposes retail sales and use tax on certain enumerated services under RCW 82.04.050 ("retail services"). For example, the sale of credit bureau services is subject to retail sales tax. However, when a retail service is transferred electronically and also meets the definition of digital automated service or digital good, such service will be treated as a digital product and is eligible for all applicable digital products retail sales and use tax exemptions as described above in Part 5 of this rule. Retail services that are not transferred electronically or those retail services that are excluded from the definitions of digital good or digital automated service (e.g., telecommunications services and ancillary services) continue to be taxed as retail services.

Example 33. ABC creates a "canned" digital report on Company X's creditworthiness prepared prior to a customer request for the report. The report may be a credit bureau service and/or a digital good (if transferred electronically). The "canned" report is listed for sale on ABC's website. An employee of InvestCo, Inc. purchases and downloads a

digital copy of the "canned" credit report from ABC's website for InvestCo's business purpose. ABC is selling a digital good generally subject to retail sales tax. However, the "canned" report is purchased by InvestCo solely for a business purpose and therefore exempt from retail sales tax (see subsection (505) of this rule for more on this exemption).

Example 34. Company sells credit reports and credit research services. EPD Corp., requests that Company prepare a credit report for EPD's specialized business purposes. After receiving the request, Company's employee researches, analyzes and generates information from various digital sources to prepare the credit report for EPD. Company then sends the report electronically as a digital file to EPD. Company is not selling a digital good because the digital item supplied to EPD is merely a representation of a professional service performed by EPD's employee. Therefore, Company's services are not a "digital product." However, Company is still required to charge and collect retail sales tax because Company is still providing credit bureau services, a retail service, subject to retail sales tax.

Example 35. Company sells an online credit reporting service. The service includes access to searchable databases, digital data analysis, and digital data reporting tools. ManageCo investigates the credit worthiness of individuals and therefore purchases access to Company's online service. Company is selling a digital automated service to be used solely for a business purpose by ManageCo. However, the "used solely for a business purpose" exemption is limited to digital goods and is not applicable to digital automated services. As such, Company is required to charge and collect retail sales tax on its sale of the digital automated service to ManageCo.

(602) Royalties and wholesaling B&O tax on digital products. The sale of digital products to "nonend users" may be subject to royalties or wholesaling B&O tax depending on the type of transaction and the intangible rights provided to the purchaser. Transactions which provide the right to resell digital products (no copying rights) to consumers will generally be treated as wholesale sales. Additionally, transactions which allow the purchaser the right to incorporate a digital product into a new product for sale will also be treated as wholesale sales. See also subsection (503) of this rule. Other nonend user transactions involving digital products or digital codes will generally be treated as royalties transactions.

Example 36. Media Corp., licenses to Rerun Inc., the right to further broadcast a digital movie file on Rerun's website for a specified period of time. In this case Media Corp. provides Rerun with the right by contract to further commercially broadcast or exhibit a digital movie to its subscribers. This is a nonend user transaction subject to royalties B&O tax. Media Corp. would report its gross receipts from this transaction under the royalties B&O tax classification and not charge and collect retail sales tax on the transaction with Rerun. Rerun's charges for the subscription service provided to consumers are generally subject to retail sales tax and retailing B&O tax.

Example 37. Same facts as Example 36 except Rerun purchases individual digital movie files from Media Corp. with the right to resell those individual files to end users at retail instead of rebroadcasting or exhibiting to the public. In this case Media Corp. has provided Rerun with the right to resell individual digital movie files to end users. Media Corp. would report its gross receipts from this transaction under the wholesaling B&O tax classification and not charge and collect retail sales tax on the transaction with Rerun. Rerun's charges to consumers for the movie files are generally subject to retail sales tax and retailing B&O tax.

Example 38. Same facts as Example 37 except that Rerun purchases a single digital movie file with the right provided by contract to duplicate and sell that movie file. In this case Media Corp. has provided Rerun with the right to duplicate and sell individual digital movie files. Media Corp. would report its gross receipts from this transaction under the royalties B&O tax classification. Media Corp. would not need to charge and collect retail sales or use tax from Rerun. Rerun's charges to consumers for the movie files are generally subject to retail sales tax and retailing B&O tax.

Example 39. Jack is a photographer who creates a digital picture of Mt. Rainier. Jack licenses, by contract, to Cashman the right to duplicate and sell copies of the Mt. Rainier picture in retail stores. Cashman's payment to Jack is for the grant of an intangible right and subject to royalties B&O tax. Cashman's sale of the picture at retail to customers is subject to retail sales tax and retailing B&O tax.

- (603) Substantial nexus is not established in Washington if a business's only contact with the state of Washington is ownership of, or rights in, computer software as defined in RCW 82.04.215, including computer software used in providing a digital automated service; master copies of software; a digital goods or digital codes residing on servers in Washington. For purposes of this rule, "substantial nexus" means the requisite connection that a person must have with a state to allow the state to subject the person to the state's taxing authority, consistent with the commerce clause of the United States Constitution.
- (604) Amnesty. Before July 26, 2009, retail sales of downloaded digital goods on a permanent or nonpermanent basis were subject to retail sales tax. This did not include accessed or streamed digital goods. However, amnesty is available to those who did not collect or pay retail sales or use tax on digital goods and digital codes during that time. Sales of digital automated services and accessed or streamed digital goods were subject to service and other B&O tax before July 26, 2009, and amnesty does not extend to these transactions because they were not subject to retail sales tax during that time period.
- (a) Refunds and credits of retail sales or use tax. No refund or credit will be given for state and local retail sales and use taxes properly paid on the sale or use, before July 26, 2009, of digital goods or of installing, repairing, altering, or improving digital goods.
- (b) No B&O tax refund or credit unless sales tax was paid. If a taxpayer paid B&O tax under the service and other activities classification prior to July 26, 2009, on income received from retail sales of digital products or digital codes, the taxpayer may not receive a refund or credit for the difference between the B&O tax actually paid and the B&O tax that should have been paid under the retailing classification unless the taxpayer has remitted the retail sales tax for those sales.
- (605) Bundled transactions. A "bundled transaction" is the retail sale of two or more products, which are distinct and identifiable for one nonitemized price. Because retail sales of digital products and digital codes are subject to retail sales tax, the general rules on the taxation of bundled transactions may apply to certain transactions involving digital products and digital codes. See RCW 82.08.190 and 82.08.195 for more information on the tax treatment of bundled transactions.

- (606) Property tax. The excise tax laws relating to digital products and digital codes do not have any impact in the characterization of digital goods and digital codes as tangible or intangible personal property for purposes of property taxation and may not be used in any way in construing Title 84 RCW. See section 1201, chapter 535, Laws of 2009.
- (607) Records requirement. Every taxpayer liable for any tax collected by the department must keep and preserve, for a period of five years, suitable records to determine the amount of tax owed. Such records must include copies of all federal income tax and state tax returns and reports. All of taxpayer's books, records, and invoices must be open to examination by the department. An out-of-state taxpayer that does not keep the necessary books and records within this state may produce within this state the books and records required by the department, or permit the examination by an agent authorized or designated by the department at the place the books and records are kept.

Washington State Register, Issue 23-23

WSR 23-23-080 EXPEDITED RULES

EMPLOYMENT SECURITY DEPARTMENT

[Filed November 13, 2023, 1:37 p.m.]

Title of Rule and Other Identifying Information: WAC 192-220-070 Cross-reference cleanup.

Purpose of the Proposal and Its Anticipated Effects, Including Any Changes in Existing Rules: The purpose of this proposal is to amend WAC 192-220-070 to remove a cross reference to RCW 50.12.070 (2)(c). In 2013, the legislature amended RCW 50.12.070, and subsection (2)(c) was removed from the statute. The substance of RCW 50.12.070 (2)(c) was moved to subsection (2)(b). Thus, the cross reference in WAC 192-220-070 should be updated to reference RCW 50.12.070 (2)(b).

Reasons Supporting Proposal: The proposal is needed to update WAC 192-220-070 so that it references the correct section in RCW 50.12.070.

Statutory Authority for Adoption: RCW 50.12.010 and 50.12.040. Statute Being Implemented: RCW 50.12.070.

Rule is not necessitated by federal law, federal or state court decision.

Name of Agency Personnel Responsible for Drafting: Lawrence Larson, Olympia, Washington, 360-890-3460; Implementation and Enforcement: JR Richards, Olympia, Washington, 360-463-1079.

This notice meets the following criteria to use the expedited adoption process for these rules:

Adopts or incorporates by reference without material change federal statutes or regulations, Washington state statutes, rules of other Washington state agencies, shoreline master programs other than those programs governing shorelines of statewide significance, or, as referenced by Washington state law, national consensus codes that generally establish industry standards, if the material adopted or incorporated regulates the same subject matter and conduct as the adopting or incorporating rule.

Corrects typographical errors, makes address or name changes, or clarifies language of a rule without changing its effect.

Explanation of the Reason the Agency Believes the Expedited Rule-Making Process is Appropriate: The expedited rule-making process is appropriate because the proposed rule changes only correct a cross reference and make no substantive changes.

NOTICE

THIS RULE IS BEING PROPOSED UNDER AN EXPEDITED RULE-MAKING PROC-ESS THAT WILL ELIMINATE THE NEED FOR THE AGENCY TO HOLD PUBLIC HEAR-INGS, PREPARE A SMALL BUSINESS ECONOMIC IMPACT STATEMENT, OR PROVIDE RESPONSES TO THE CRITERIA FOR A SIGNIFICANT LEGISLATIVE RULE. IF YOU OBJECT TO THIS USE OF THE EXPEDITED RULE-MAKING PROCESS, YOU MUST EX-PRESS YOUR OBJECTIONS IN WRITING AND THEY MUST BE SENT TO Lawrence Larson, Employment Security Department, P.O. Box 9046, Olympia, WA 98507-9046, phone 360-890-3460, fax 844-652-7096, email rules@esd.wa.gov, TTY relay 711, AND RECEIVED BY January 23, 2024.

> November 14, 2023 Joy Adams Employment Security Acting Policy Director

OTS-5081.1

AMENDATORY SECTION (Amending WSR 07-23-128, filed 11/21/07, effective 1/1/08)

WAC 192-220-070 Overpayments under RCW 50.12.070 (2)(((c))) <u>(b)</u>. You are not required to repay benefits improperly paid to you because an employer failed to correctly report your wages or hours and a later correction results in a lower benefit amount or your claim becomes invalid. However, you remain liable for any overpayment assessment resulting from an eligibility decision issued before your claim became invalid that has become final.

WSR 23-23-114 EXPEDITED RULES DEPARTMENT OF NATURAL RESOURCES

[Filed November 15, 2023, 5:15 p.m.]

Title of Rule and Other Identifying Information: WAC 332-10-041 Meetings of board of natural resources.

Purpose of the Proposal and Its Anticipated Effects, Including Any Changes in Existing Rules: The scheduled days for the regular meetings of the board of natural resources will be changed to better accommodate nonlocal members. Rule will also be updated to reflect current technology for hybrid meeting capabilities.

Reasons Supporting Proposal: Multiple board members need to travel from out of the area to attend board of natural resources meetings. When a holiday occurs on the Monday before a scheduled meeting, the board members are currently having to travel on that holiday. Changes are designed to allow for board meetings to occur on Wednesday if a holiday occurs on a Monday.

New technology has changed how board members are able to attend meetings remotely. Changes are designed to reflect this.

Statutory Authority for Adoption: RCW 43.30.225.

Statute Being Implemented: RCW 43.30.225.

Rule is not necessitated by federal law, federal or state court decision.

Name of Proponent: Washington department of natural resources, governmental.

Name of Agency Personnel Responsible for Drafting: William Wells, Forks, Washington, 360-640-0181; Implementation: Tami Kellogg, Olympia, Washington, 360-902-2122; and Enforcement: Adrienne Smith, Olympia, Washington, 360-586-3204.

This notice meets the following criteria to use the expedited adoption process for these rules:

Relates only to internal governmental operations that are not subject to violation by a person.

NOTICE

THIS RULE IS BEING PROPOSED UNDER AN EXPEDITED RULE-MAKING PROC-ESS THAT WILL ELIMINATE THE NEED FOR THE AGENCY TO HOLD PUBLIC HEAR-INGS, PREPARE A SMALL BUSINESS ECONOMIC IMPACT STATEMENT, OR PROVIDE RESPONSES TO THE CRITERIA FOR A SIGNIFICANT LEGISLATIVE RULE. IF YOU OBJECT TO THIS USE OF THE EXPEDITED RULE-MAKING PROCESS, YOU MUST EXPRESS YOUR OBJECTIONS IN WRITING AND THEY MUST BE SENT TO Rochelle Goss, Department of Natural Resources, 1111 Washington Street S.E., P.O. Box 47015, Olympia, WA 98504-7015, phone 360-902-2117, email Rochelle.goss@dnr.wa.gov, AND RECEIVED BY January 22, 2024.

> November 14, 2023 Dale Mix Acting Deputy Supervisor for Todd Welker Deputy Supervisor State Uplands

AMENDATORY SECTION (Amending WSR 02-19-058, filed 9/12/02, effective 10/13/02)

- WAC 332-10-041 Meetings of board of natural resources. (1) Reqular meetings of the board of natural resources shall be held on the first Tuesday of every month except August. If a regular meeting falls on a holiday or the day after a holiday, such regular meeting shall be held on the next business day. A schedule of meetings will be published in the Washington State Register in January of each year. Changes to the schedule will be published in the Washington State Register pursuant to RCW 42.30.075. Special meetings may be held pursuant to RCW 42.30.080. Any person may obtain information about locations and meeting times by contacting the Department of Natural Resources, P.O. Box 47001, Olympia, Washington 98504-7001. The public is invited to attend and comment at all meetings.
- (2) Members of the board of natural resources may participate by telephone or other means of remote access in any regular or special meeting so long as a ((speaker phone is available at the public meeting location. To assure that all discussions comply with RCW 42.30.030,)) board discussion can be heard by those attending the public meeting and the members can hear what is stated in the meeting. If more than one board member is participating by telephone or other means of remote access, then each such board member shall ((use a separate telephone line or a separate telephone. If other electronic means of attending meetings from remote locations are available, board members may employ these means so long as board discussion can be heard by those attending the public meeting consistent with the Open Public Meetings Act RCW 42.30)) join the meeting separately, such as by a separate telephone or electronic meeting link. Board members participating by telephone or other ((electronic)) means of remote access may vote on any matter and shall be considered as part of the quorum.

WSR 23-23-125 EXPEDITED RULES DEPARTMENT OF REVENUE

[Filed November 16, 2023, 1:29 p.m.]

Title of Rule and Other Identifying Information: WAC 458-20-228 Returns, payments, penalties, extensions, interest, stays of collection.

Purpose of the Proposal and Its Anticipated Effects, Including Any Changes in Existing Rules: The department of revenue (department) intends to amend WAC 458-20-228 to update the changes enacted in 2023 legislation, chapter 374, Laws of 2023 (SHB 5565) and 2019 legislation, chapter 63, Laws of 2019 (2SHB 1059). In addition, the department will update reference to the online filing and payment system, rule names, and statute section numbers.

Reasons Supporting Proposal: The Washington legislature repealed RCW 82.27.060 Payment of tax—Remittance—Returns, and 2006 c 256 s 3, 2003 1st sp.s. c 13 s 10, 1990 c 214 s 1, & 1980 c 98 s 6. The department is updating WAC 458-20-228 to remove reference to RCW 82.27.060 in this rule. In addition, the legislature amended RCW 82.32.045 Taxes --When due and payable--Reporting periods--Verified annual returns--Relief from filing requirements, and 2010 1st sp.s. c 23 1103, which changed the due date for annual filers.

Statutory Authority for Adoption: RCW 82.01.060 and 82.32.300. Statute Being Implemented: RCW 82.27.060, 82.32.045, and 82.32.050.

Rule is not necessitated by federal law, federal or state court decision.

Name of Proponent: Department of revenue, governmental.

Name of Agency Personnel Responsible for Drafting: Patrick Watkins, 6400 Linderson Way S.W., Tumwater, WA, 360-534-1539; Implementation and Enforcement: John Ryser, 6400 Linderson Way S.W., Tumwater, WA 360-534-1605.

This notice meets the following criteria to use the expedited adoption process for these rules:

Adopts or incorporates by reference without material change federal statutes or regulations, Washington state statutes, rules of other Washington state agencies, shoreline master programs other than those programs governing shorelines of statewide significance, or, as referenced by Washington state law, national consensus codes that generally establish industry standards, if the material adopted or incorporated regulates the same subject matter and conduct as the adopting or incorporating rule.

Corrects typographical errors, makes address or name changes, or clarifies language of a rule without changing its effect.

Explanation of the Reason the Agency Believes the Expedited Rule-Making Process is Appropriate: The expedited rule-making process is appropriate for this rule update because the department is incorporating changes resulting from 2023 legislation.

NOTICE

THIS RULE IS BEING PROPOSED UNDER AN EXPEDITED RULE-MAKING PROC-ESS THAT WILL ELIMINATE THE NEED FOR THE AGENCY TO HOLD PUBLIC HEAR-INGS, PREPARE A SMALL BUSINESS ECONOMIC IMPACT STATEMENT, OR PROVIDE RESPONSES TO THE CRITERIA FOR A SIGNIFICANT LEGISLATIVE RULE. IF YOU OBJECT TO THIS USE OF THE EXPEDITED RULE-MAKING PROCESS, YOU MUST EX- PRESS YOUR OBJECTIONS IN WRITING AND THEY MUST BE SENT TO Patrick Watkins, Department of Revenue, P.O. Box 47453, phone 360-534-1539, fax 360-534-1606, email patrickw@dor.wa.gov, TTY 800-833-6384, AND RE-CEIVED BY January 22, 2024.

> November 16, 2023 Atif Aziz Rules Coordinator

OTS-5066.1

AMENDATORY SECTION (Amending WSR 23-14-002, filed 6/21/23, effective 7/22/23)

WAC 458-20-228 Returns, payments, penalties, extensions, interest, stays of collection. (1) Introduction. This rule discusses the responsibility of taxpayers to pay their tax by the appropriate due date, and the acceptable methods of payment. It discusses the interest and penalties that are imposed by law when a taxpayer fails to pay the correct amount of tax by the due date. It also discusses the circumstances under which the law allows the department of revenue (department) to waive interest or penalties.

- (a) Where can I get my questions answered, or learn more about what I owe and how to report it? Washington's tax system is based largely on voluntary compliance. Taxpayers have a legal responsibility to become informed about applicable tax laws, to register with the department, to seek instruction from the department, to file accurate returns, and to pay their tax liability in a timely manner (chapter 82.32A RCW, Taxpayer rights and responsibilities). The department has a taxpayer services program to provide taxpayers with accurate tax-reporting assistance and instructions. The department staffs local district offices, maintains a question and information phone line (360-705-6705), provides information and electronic forms on the internet (((http://))dor.wa.gov), and conducts free public workshops on tax reporting. The department also publishes notices, interpretive statements, and rules discussing important tax issues and changes.
- (b) What is ((electronic filing (or e-file),)) $\underline{\text{My DOR}}$ and how can it help me? ((E-file)) My DOR is an internet-based application ((that provides)) providing a secure and encrypted way for taxpayers to file and pay many of Washington state's ((business related)) excise taxes online. The ((e-file)) My DOR system automatically performs math calculations and checks for other types of reporting errors. Using ((efile)) My DOR to file electronically will help taxpayers avoid penalties and interest related to unintentional underpayments and delinquencies. ((E-file)) My DOR can be accessed on the department's internet site ((http://dor.wa.gov. Open the page for electronic filing. The page contains additional links to pages answering frequently asked questions, and explains the registration process for e-file)) dor.wa.gov. Taxpayers may also call the department's ((toll-free electronic filing help desk)) telephone center at 360-705-6705 for more information ((during regular business hours)).

All taxpayers are required to electronically file and electronically pay their taxes unless the department waives the requirement

((for good cause, or the taxpayer has an assigned reporting frequency that is less than quarterly. The requirement for electronic filing and payment also includes taxpayers who once met the criteria for being assigned to a monthly reporting frequency, but whom since have been authorized by the department to file and remit taxes on a less frequent basis)) in accordance with RCW 82.32.080. For more detailed information on the requirement and exceptions for ((electronic filing (e-file) and electronic payment (e-pay))) electronically filing using My DOR and submitting payment electronically, see WAC 458-20-22802 (Electronic filing and payment).

(c) Index of subjects addressed in this rule:

Topic—Description	See subsection
Where can I get my questions answered, or learn more about what I owe and how to report it? - By phone or online, the department provides a number of free and easy resources to help you find answers.	(1)(a) of this rule, (see above)
What is ((electronic filing (or e-file),)) My DOR and how can it help me? - ((E-filing)) My DOR guides you through the return and helps you avoid many common mistakes.	(1)(b) of this rule, (see above)
Do I need to file a return? - How do I access returns and file them?	(2) of this rule
What methods of payment can I use? - What can I use to pay my taxes?	(3) of this rule
When is my tax payment due? - Different reporting frequencies can have different due dates. What if the due date is a weekend or a holiday? If my payment is in the mail on the due date, am I late or on time?	(4) of this rule
Penalties - What types of penalty exist? How big are they? When do they apply?	(5) of this rule
Statutory restrictions on imposing penalties - More than one penalty can apply at the same time, but there are restrictions. Which penalties can be combined?	(6) of this rule
Interest - In most cases interest is required. What interest rates apply? How is interest applied?	(7) of this rule
Application of payment towards liability - Interest, penalties, and taxes are paid in a particular order. If my payment doesn't pay the entire liability, how can I determine what parts have been paid?	(8) of this rule
Waiver or cancellation of penalties - I think I was on time, or I had a good reason for not paying the tax when I should have. What reasons qualify me for a waiver of penalty? How can I get a penalty removed?	(9) of this rule
Waiver or cancellation of interest - Interest will only be waived in two limited situations. What are they?	(10) of this rule

Topic—Description	See subsection
Interest and penalty waiver for active duty military personnel - Is a majority owner of the business on active duty with the military? BOTH interest and penalty can be waived if all the statutory requirements are met. What are the requirements?	(11) of this rule
Stay of collection - Revenue will sometimes temporarily delay collection action on unpaid taxes. When can this happen? Can I request that revenue delay collection?	(12) of this rule
Extensions - Can I get an extension of my due date? How long does an extension last? A special extension may be available if the governor proclaims a state of emergency in your area.	(13) of this rule

- (2) Do I need to file a return? A "return" is defined as any paper or electronic document a person is required to file by the state of Washington in order to satisfy or establish a tax or fee obligation which is administered or collected by the department, and that has a statutorily defined due date. RCW ((82.32.045)) 82.32.050.
- (a) Electronic returns and payments are to be filed with the department by every person liable for any tax which the department administers and/or collects, except for the taxes imposed under chapter 82.24 RCW (Tax on cigarettes), which are collected through sales of revenue stamps. Returns must be filed through the ((electronic filing (e-file))) My DOR system (see subsection (1)(b) of this rule), or by other means if approved by the department.
- ((E-file)) Taxpayers who file with My DOR do not receive paper returns. However, ((if an e-file taxpayer specifically requests it, the department will send)) taxpayers can set up alerts in My DOR and an electronic reminder for each upcoming return as the time to file approaches.
- (b) Taxpayers whose accounts are placed on an "active nonreporting" status ((do not automatically receive)) <u>are required to timely</u> notify the department and file a tax return ((and must request a return, or register to file by e-file,)) with My DOR if they no longer qualify for this reporting status. ((+)) See WAC 458-20-101((-)) (Tax registration $((\tau))$ and tax reporting for an explanation of the active nonreporting status. ((+))
- (c) Some consumers may not be required to register with the department and obtain a tax registration endorsement. (Refer to WAC 458-20-101 for detailed information about tax registration and when it is required.) But even if they do not have to be registered, consumers may be required to pay use tax directly to the department if they have purchased items without paying Washington's sales tax. An unregistered consumer must report and pay their use tax liability directly to the department. Use tax can be reported and paid on a "Consumer Use Tax Return" or the consumer can create an online account at the department's website to conveniently report and pay use tax electronically. Consumer use tax returns are available from the department at any of the local district offices. A consumer may also call the department's telephone information center at 360-705-6705 to request a consumer use tax return by ((fax or)) mail. Finally, the consumer use tax return is available for download from the department's internet site at

((http://))dor.wa.gov, along with a number of other returns and forms which are available there.

The interest and penalty provisions of this rule may apply if use tax is not paid on time. Unregistered consumers should refer to WAC 458-20-178 (Use tax and use of tangible personal property) for an explanation of their tax reporting responsibilities.

- (3) What methods of payment can I use? The law requires taxpayers to file and pay their taxes electronically. There are two electronic payment methods: Electronic funds transfer (EFT) and credit card. The department may waive the electronic payment requirement for any taxpayer or class of taxpayers, for good cause ((or for whom the department has assigned a reporting frequency that is less than quarterly)). Waivers may be temporary or permanent, and may be made on the department's own motion. (See WAC 458-20-22802 for more information on electronic filing and payment.)
- (a) For taxpayers not required to pay electronically, payment may be made by cash, check, cashier's check, or money order.
- (b) Payment by cash should only be made at an office of the department to ensure that the payment is safely received and properly credited.
- (c) Payment may be made by uncertified bank check, but if the check is not honored by the financial institution on which it is drawn, the taxpayer remains liable for the payment of the tax, as well as any applicable interest and penalties. RCW 82.32.080. The department may refuse to accept any check which, in its opinion, would not be honored by the financial institution on which that check is drawn. If the department refuses a check for this reason the taxpayer remains liable for the tax due, as well as any applicable interest and penalties.
- (4) When is my tax payment due? RCW 82.32.045 provides that payment of the taxes due with the excise tax return must be made monthly and within 25 days after the end of the month in which taxable activities occur, unless the department assigns the taxpayer a longer reporting frequency. Payment of taxes due with returns covering a ((lonqer)) quarterly reporting frequency ((is)) are due on or before the last day of the month following the period covered by the return. (For example, payment of the tax liability for a first quarter tax return is due on April 30th.) For annual filers, tax payments, along with reports and returns are due on or before April 15th of the year immediately following the end of the period covered by the return. WAC 458-20-22801 (Tax reporting frequency((-Forms))) explains the department's procedure for assigning a quarterly or annual reporting fre-
- (a) If the date for payment of the tax due on a tax return falls upon a Saturday, Sunday, or legal holiday, the filing will be considered timely if performed on the next business day. RCW 1.12.070 and 1.16.050.
- (b) When a taxpayer is not required to electronically file and pay taxes and chooses to file or pay taxes through the U.S. Postal Service, the postmark date as shown by the post office cancellation mark stamped on the envelope will be considered conclusive evidence by the department in determining if a tax return or payment was timely filed or received. RCW 1.12.070. It is the responsibility of the taxpayer to mail the tax return or payment sufficiently in advance of the due date to assure that the postmark date is timely.

- (c) Taxpayers required to file and pay taxes electronically should refer to WAC 458-20-22802 (Electronic filing and payment) for more information regarding ((electronic filing (e-file))) My DOR, electronic payment (((e-pay))) due dates, and when electronic payments are considered received.
- (d) If a taxpayer suspects that it will not be able to file and pay by the coming due date, it may be able to obtain an extension of the due date to temporarily avoid additional penalties. Refer to subsection (12) of this rule for details on requesting an extension.
- (5) **Penalties.** Various penalties may apply as a result of the failure to correctly or accurately compute the proper tax liability, or to timely pay the tax. Separate penalties may apply and be cumulative for the same tax. Interest may also apply if any tax has not been paid when it is due, as explained in subsection (7) of this rule. (The department's ((electronic filing)) My DOR system ((e-file)) can help taxpayers avoid additional penalties and interest. See subsection (1) (b) of this rule for more information.)

The penalty types and rates addressed in this subsection are:

Penalty Type—Description	Penalty Rate	See subsection
Late payment of a return - Nine percent added when payment is not received by the due date, and increases if the tax due remains unpaid.	9/19/29%	(5)(a) of this rule
Unregistered taxpayer - Five percent added against unpaid tax when revenue discovers a taxpayer who has taxable activity but is not registered.	5%	(5)(b) of this rule
Assessment - Five percent added when a tax assessment is issued if the tax was "substantially underpaid," and increases if the tax due remains unpaid.	5/15/25% or 0/15/25%	(5)(c) of this rule
Issuance of a warrant - Ten percent added when a warrant is issued to collect unpaid tax, and does not require actual filing of a lien.	10%	(5)(d) of this rule
Disregard of specific written instructions - Ten percent added when the department has provided specific, written reporting instructions and tax is underpaid because the instructions are not followed.	10%	(5)(e) of this rule
Evasion - Fifty percent added when tax is underpaid and there is an intentional effort to hide that fact.	50%	(5)(f) of this rule
Misuse of resale certificates or a reseller permit - Fifty percent added against unpaid sales tax when a buyer uses a resale certificate or reseller permit, but should not have.	50%	(5)(g) of this rule

Penalty Type—Description	Penalty Rate	See subsection
Failure to remit sales tax to seller - Ten percent added against sales tax when the department proceeds directly against a buyer who fails to pay sales tax to the seller as part of a sales taxable retail purchase.	10%	(5)(h) of this rule
Failure to obtain the contractor's unified business identifier (UBI) number - A two hundred fifty dollar maximum penalty (does not require any tax liability) when specified businesses hire certain contractors but do not obtain and keep the contractor's UBI number.	\$250 (max)	(5)(i) of this rule
Disregarded transaction - A thirty-five percent penalty of the additional tax found to be due as a result of engaging in a disregarded transaction.	35%	(5)(j) of this rule

(a) Late payment of a return. RCW 82.32.090(1) imposes a nine percent penalty if the tax due on a taxpayer's return is not paid by the due date. A total penalty of 19 percent is imposed if the tax due is not paid on or before the last day of the month following the due date, and a total penalty of 29 percent is imposed if the tax due is still not paid on or before the last day of the second month following the due date. The minimum penalty for late payment is five dollars.

Various sets of circumstances can affect how the late payment of a return penalty is applied. See (a)(i) through (iii) of this subsection for some of the most common circumstances.

- (i) Will I avoid the penalty if I file my return without the payment? The department may refuse to accept any return that is not accompanied by payment of the tax shown to be due on the return. If the return is not accepted, the taxpayer is considered to have failed or refused to file the return. RCW 82.32.080. Failure to file the return can result in the issuance of an assessment for the actual, or an estimated, amount of unpaid tax. Any assessment issued may include an assessment penalty. (See RCW 82.32.100 and (c) of this subsection for details of when and how the assessment penalty applies.) If the tax return is accepted without payment and payment is not made by the due date, the late payment of return penalty will apply.
- (ii) What if my account is given an active nonreporting status, but I later have taxes I need to report and pay? WAC 458-20-101 provides information about the active nonreporting status available for tax reporting accounts. In general, the active nonreporting status allows persons, under certain circumstances, to engage in business activities subject to the Revenue Act without filing excise tax returns. Persons placed on an active nonreporting status by the department are required to timely notify the department if their business activities no longer meet the conditions to be in active nonreporting status. One of the conditions is that the person is not required to collect or pay a tax the department is authorized to collect. The late payment of return penalty will be imposed if a person on active nonreporting status

incurs a tax liability that is not paid by the due date for taxpayers that are on an annual reporting basis (((i.e., the last day of January next succeeding the year in which the tax liability accrued))).

- (iii) I ((didn't)) did not register my business with the department when I started it, and now I think I was supposed to be paying taxes! What should I do? You should fill out ((and send in)) a business license application to get your business registered. It is important for you to register before the department identifies you as an unregistered taxpayer and contacts you about your business activities. (WAC 458-20-101 provides information about registering your business.) Except as noted below, if a person engages in taxable activities while unregistered, but then registers prior to being contacted by the department, the registration is considered voluntary. When a person voluntarily registers, the late payment of return penalty does not apply to those specific tax-reporting periods representing the time during which the person was unregistered.
- (A) However, even if the person has voluntarily registered as explained above, the late payment of return penalty will apply if the person:
- (I) Collected retail sales tax from customers and failed to remit it to the department; or
- (II) Engaged in evasion or misrepresentation with respect to reporting tax liabilities or other tax requirements; or
- (III) Engaged in taxable business activities during a period of time in which the person's previously open tax reporting account had been closed.
- (B) Even though other circumstances may warrant retention of the late payment of return penalty, if a person has voluntarily registered, the unregistered taxpayer penalty (see (b) of this subsection) will not be due.
- (b) Unregistered taxpayer. RCW 82.32.090(4) imposes a five percent penalty on the tax due for any period of time where a person engages in a taxable activity and does not voluntarily register prior to being contacted by the department. "Voluntarily register" means to properly complete and submit a master application to any agency or entity participating in the unified business identifier (UBI) program for the purpose of obtaining a UBI number, all of which is done before any contact from the department. For example, if a person properly completes and submits a business license application to the department of labor and industries for the purpose of obtaining a UBI number, and this is done prior to any contact from the department of revenue, the department considers that person to have voluntarily registered. A person has not voluntarily registered if a UBI number is obtained by any means other than submitting a properly completed business license application. WAC 458-20-101 (Tax registration and tax reporting) provides additional information regarding the UBI program.
- (c) Assessment. If the department issues an assessment for substantially underpaid tax, a five percent penalty will be added to the assessment when it is issued. If any tax included in the assessment is not paid by the due date, or by any extended due date, the penalty will increase to a total of 15 percent against the amount of tax that remains unpaid. If any tax included in the assessment is not paid within 30 days of the original or extended due date, the penalty will further increase to a total of 25 percent against the amount of tax that remains unpaid. The minimum for this penalty is five dollars. RCW 82.32.090(2).
 - (i) As used in this rule, "substantially underpaid" means that:

- (A) The taxpayer has paid less than 80 percent of the amount of tax determined by the department to be due for all of the types of taxes included in, and for the entire period of time covered by, the department's examination; and
- (B) The amount of underpayment is at least \$1,000. If both of these conditions are true when an assessment is issued, it will include the initial five percent assessment penalty. If factual adjustments are made after issuance of an assessment, and those adjustments change whether a taxpayer paid less than 80 percent of the tax due, the department will reevaluate imposition of the original five percent penalty.
- (ii) If the initial five percent assessment penalty is included with an assessment when it is issued, the penalty is calculated against the total amount of tax that was not paid when originally due and payable (see RCW 82.32.045). Audit payments made prior to issuance of an assessment will be applied to the assessment after calculation of the initial five percent assessment penalty. At the discretion of the department, preexisting credits or amendments paid prior to an audit or unrelated to the scope of the assessment may be applied before the five percent assessment penalty is calculated, reducing the amount of the penalty. Additional assessment penalty is assessed against the amount of tax that remains unpaid at that particular time, after payments are applied to the assessment.
- (d) Issuance of a warrant. If the department issues a tax warrant for the collection of any fee, tax, increase, or penalty, an additional penalty will immediately be added in the amount of 10 percent of the amount of the tax due, but not less than \$10.00. RCW 82.32.090(3). Refer to WAC 458-20-217 for additional information on the application of warrants and tax liens.
- (e) Disregard of specific written instructions. If the department finds that all or any part of a deficiency resulted from the disregard of specific written instructions as to reporting of tax liabilities, an additional penalty of 10 percent of the additional tax found due will be imposed because of the failure to follow the instructions. RCW 82.32.090(5).
- (i) What is "disregard of specific written instructions"? A taxpayer is considered to have received specific written instructions when the department has informed the taxpayer in writing of its tax obligations and specifically advised the taxpayer that failure to act in accordance with those instructions may result in this penalty being imposed. The specific written instructions may be given as a part of a tax assessment, audit, determination, or closing agreement. The penalty applies when a taxpayer does not follow the specific written instructions, resulting in underpayment of the tax due. The penalty may be applied only against the taxpayer given the specific written instructions. However, the taxpayer will not be considered to have disregarded the instructions if the taxpayer has appealed the subject matter of the instructions and the department has not issued its final instructions or decision.
- $(ext{ii})$ What if I try to follow the written instructions, but I still don't get it quite right? The penalty will not be applied if the taxpayer has made a good faith effort to comply with specific written instructions.
- (f) Evasion. If the department finds that all or any part of the deficiency resulted from an intent to evade the tax due, a penalty of 50 percent of the additional tax found to be due will be added. RCW 82.32.090(7). The evasion penalty is imposed when a taxpayer knows a

tax liability is due but attempts to escape detection or payment of the tax liability through deceit, fraud, or other intentional wrongdoing. An intent to evade does not exist where a deficiency is the result of an honest mistake, miscommunication, or the lack of knowledge regarding proper accounting methods. The department has the burden of showing the existence of an intent to evade a tax liability through clear, cogent and convincing evidence.

- (i) Evasion penalty only applies to the specific taxes that a taxpayer intended to evade. To the extent that the evasion involved only specific taxes, the evasion penalty will be added only to those taxes. The evasion penalty will not be applied to those taxes which were inadvertently underpaid. For example, if the department finds that the taxpayer intentionally understated the purchase price of equipment in reporting use tax and also inadvertently failed to collect or remit the sales tax at the correct rate on retail sales of merchandise, the evasion penalty will be added only to the use tax deficiency and not the sales tax.
- (ii) What actions may establish an intent to evade? The following is a nonexclusive list of actions that are generally considered to establish an intent to evade a tax liability. This list should only be used as a general guide. A determination of whether an intent to evade exists may be ascertained only after a review of all the facts and circumstances.
- (A) The use of an out-of-state address by a Washington resident to register property to avoid a Washington excise or use tax, when at the time of registration the taxpayer does not reside at the out-ofstate address on a more than temporary basis. Examples of such an address include, but are not limited to, the residence of a relative, mail forwarding or post office box location, motel, campground, or vacation property;
- (B) The willful failure of a seller to remit retail sales taxes collected from customers to the department; and
- (C) The alteration of a purchase invoice or misrepresentation of the price paid for property (e.g., a used vehicle) to reduce the amount of tax owing.
- (q) Misuse of resale certificates, reseller permits, and other documents. Any buyer who uses a resale certificate, a reseller permit, or other documentation authorized under RCW 82.04.470, to purchase items or retail services without payment of sales tax, and who is not entitled to use the certificate, permit, or other documentation for the purchase, will be assessed a penalty of 50 percent of the tax due. RCW 82.32.291. The penalty can apply even if there was no intent to evade the payment of the tax. For more information concerning this penalty or the proper use of resale certificates, reseller permits, and other documentation, refer to WAC 458-20-102 (((Resale certificates)) Reseller permits).
- (h) Failure to remit sales tax to seller. The department may assert an additional 10 percent penalty against a buyer who has failed to pay the seller the retail sales tax on taxable purchases, if the department proceeds directly against the buyer for the payment of the tax. This penalty is in addition to any other penalties or interest prescribed by law. RCW 82.08.050.
- (i) Failure to obtain the contractor's unified business identifier (UBI) number. If a person who is liable for any fee or tax imposed by chapters 82.04 through 82.27 RCW contracts with another person or entity for work subject to chapter 18.27 RCW (Registration of contractors) or chapter 19.28 RCW (Electricians and electrical installa-

- tions), that person must obtain and preserve a record of the UBI number of the person or entity performing the work. A person failing to do so is subject to the public works contracting restrictions in RCW 39.06.010 (Contracts with unregistered or unlicensed contractors prohibited), and a penalty determined by the director, but not to exceed \$250. RCW 82.32.070(2).
- (j) Engaging in disregarded transactions. RCW 82.32.090 imposes a 35 percent penalty for engaging in a disregarded transaction as defined in RCW 82.32.655(3). See RCW 82.32.090(6), 82.32.655, and 82.32.660.
- (6) Statutory restrictions on imposing penalties. Depending on the circumstances, the law may impose more than one type of penalty on the same tax liability. However, those penalties are subject to the following restrictions:
- (a) The penalties imposed for the late payment of a return, unregistered taxpayer, assessment, and issuance of a warrant (see subsection (5)(a) through (d) of this rule) may be applied against the same tax concurrently, each unaffected by the others, up to their combined maximum rates. Application of one or any combination of these penalties does not prohibit or restrict full application of other penalties authorized by law, even when they are applied against the same tax. RCW 82.32.090(8).
- (b) The department may impose either the evasion penalty (subsection (5)(f) of this rule) or the penalty for disregarding specific written instructions (subsection (5)(e) of this rule), but may not impose both penalties on the same tax. RCW 82.32.090(9). The department also will not impose the penalty for the misuse of a resale certificate (subsection (5)(g) of this rule) in combination with either the evasion penalty or the penalty for disregarding specific written instructions on the same tax.
- (c) The penalty provided in subsection (5)(j) of this rule may be assessed together with any other applicable penalties provided in this rule on the same tax found to be due, except for the evasion penalty provided in subsection (5)(f) of this rule.
- (7) **Interest.** The department is required by law to add interest to assessments for tax deficiencies and overpayments. RCW 82.32.050 and 82.32.060. Interest accrued against an underpayment only applies to underpaid tax. (Refer to WAC 458-20-229 for a discussion of interest as it relates to refunds and WAC 458-20-230 for a discussion of the statute of limitations as applied to interest.)
- (a) For interest imposed after December 31, 1998, interest will be added from the last day of the month following each calendar year included in a notice, or the last day of the month following the final month included in a notice if not the end of the calendar year, until the due date of the notice. However, for 1998 taxes only, interest may not begin to accrue any earlier than February 1, 1999, even if the last period included in the notice is not at the end of calendar year 1998. If payment in full is not made by the due date of the notice, additional interest will be due until the date of payment. The rate of interest continues at the annual variable interest rates described below in (c) of this subsection.
- (b) How is interest applied to an assessment that includes underpaid tax from multiple years? The following is an example of how the interest provisions apply. Assume that a tax assessment is issued with a due date of June 30, 2010. The assessment includes periods from January 1, 2008, through September 30, 2009.

- (i) For calendar year 2008 tax, interest begins February 1, 2009, (from the last day of the month following the end of the calendar year). When the assessment is issued interest is computed through June 30, 2010, (the due date).
- (ii) For the 2009 tax period ending with September 30, 2009, interest begins November 1, 2009, (from the last day of the month following the last month included in the assessment period). When the assessment is issued interest is computed through June 30, 2010, (the due date).
- (iii) Interest will continue to accrue on any portion of the assessed taxes which remain unpaid after the due date, until the date those taxes are paid.
- (c) How is each year's interest rate determined? The annual variable interest rate will be an average of the federal short-term rate as defined in 26 U.S.C. Sec. 1274(d) plus two percentage points. The rate for each new year will be computed by taking an arithmetical average to the nearest percentage point of the federal short-term rate, compounded annually. The average is calculated using the federal short-term rates from January, April, July of the calendar year immediately preceding the new year, and October of the previous preceding year, as published by the United States Secretary of the Treasury. The interest rate will be adjusted on the first day of January of each year.
- (d) How is the interest applied if an assessment includes some years that are underpaid and some that are overpaid? If the assessment contains tax deficiencies in some years and overpayments in other years with the net difference being a tax deficiency, the interest rate for tax deficiencies will also be applied to the overpayments. (Refer to WAC 458-20-229 for interest on refunds.)
- (8) Application of payment towards liability. The department will apply taxpayer payments in the following order:
 - Interest;
 - Penalties;
 - Fees;
 - Other nontax amounts;
 - Tax, except spirits tax;
 - Spirits tax;

without regard to any direction of the taxpayer. RCW 82.32.080.

In applying a partial payment to a tax assessment, the payment will first be applied against the oldest tax liability. For purposes of RCW 82.32.145 (Limited liability business entity - Terminated, dissolved, abandoned, insolvent - Collection of unpaid trust fund taxes), it will be assumed that any payments applied to the tax liability will be first applied against any retail sales tax liability, and then to other trust fund tax liabilities. For example, an audit assessment is issued covering a period of two years, which will be referred to as "YEAR 1" (the earlier year) and "YEAR 2" (the most recent year). The tax assessment includes total interest and penalties for YEAR 1 and YEAR 2 of \$500, retail sales tax of \$400 for YEAR 1, \$600 retail sales tax for YEAR 2, \$2,000 of other taxes for YEAR 1, and \$7,000 of other taxes for YEAR 2. The order of application of any payments will be first against the \$500 of total interest and penalties, second against the \$400 retail sales tax in YEAR 1, third against the \$2,000 of other taxes in YEAR 1, fourth against the \$600 retail sales tax of YEAR 2, and finally against the \$7,000 of other taxes in YEAR 2.

- (9) Waiver or cancellation of penalties. RCW 82.32.105 authorizes the department to waive or cancel penalties under limited circumstan-
- (a) Circumstances beyond the control of the taxpayer. The department will waive or cancel the penalties imposed under chapter 82.32 RCW upon finding that the underpayment of the tax, or the failure to pay any tax by the due date, was the result of circumstances beyond the control of the taxpayer. It is possible that a taxpayer will qualify for a waiver of one type of penalty, without obtaining a waiver for all penalties associated with a particular tax liability. Circumstances determined to be beyond the control of the taxpayer when considering a waiver of one type of penalty are not necessarily pertinent when considering a waiver of a different penalty type. For example, circumstances that qualify for waiver of a late payment of return penalty do not necessarily also justify waiver of the substantial underpayment assessment penalty. Refer to WAC 458-20-102 (Reseller permits) for examples of circumstances which are beyond the control of the taxpayer specifically regarding the penalty for misuse of a reseller permit found in RCW 82.32.291.
- (i) A request for a waiver or cancellation of penalties should contain all pertinent facts and be accompanied by such proof as may be available. The taxpayer bears the burden of establishing that the circumstances were beyond its control and directly caused the late payment. The request should be made in the form of a letter; however, verbal requests may be accepted and considered at the discretion of the department. Any petition for correction of assessment submitted to the department's administrative review and hearings division for waiver of penalties must be made within the period for filing under RCW 82.32.160 (within 30 days after the issuance of the original notice of the amount owed or within the period covered by any extension of the due date granted by the department), and must be in writing, as explained in WAC 458-20-100 (Informal administrative reviews). Refund requests must be made within the statutory limitation period.
- (ii) The circumstances beyond the control of the taxpayer must actually cause the late payment. Circumstances beyond the control of the taxpayer are generally those which are immediate, unexpected, or in the nature of an emergency. Such circumstances result in the taxpayer not having reasonable time or opportunity to obtain an extension of the due date or otherwise timely file and pay. Circumstances beyond the control of the taxpayer include, but are not necessarily limited to, the following.
- (A) The return payment was mailed on time but inadvertently sent to another agency.
- (B) Erroneous written information given to the taxpayer by a department officer or employee caused the delinquency. A penalty generally will not be waived when it is claimed that erroneous oral information was given by a department employee. The reason for not canceling the penalty in cases of oral information is because of the uncertainty of the facts presented, the uncertainty of the instructions or information imparted by the department employee, and the uncertainty that the taxpayer fully understood the information given. Reliance by the taxpayer on incorrect advice received from the taxpayer's legal or accounting representative is not a basis for cancellation of a penalty.
- (C) The delinquency was directly caused by death or serious illness of the taxpayer, or a member of the taxpayer's immediate family. The same circumstances apply to the taxpayer's accountant or other tax

preparer, or their immediate family. This situation is not intended to have an indefinite application. A death or serious illness which denies a taxpayer reasonable time or opportunity to obtain an extension or to otherwise arrange timely filing and payment is a circumstance eligible for penalty waiver.

- (D) The delinquency was caused by the unavoidable absence of the taxpayer or key employee, prior to the filing date. "Unavoidable absence of the taxpayer" does not include absences because of business trips, vacations, personnel turnover, or terminations.
- (E) The delinquency was caused by the destruction by fire or other casualty of the taxpayer's place of business or business records.
- (F) The delinquency was caused by an act of fraud, embezzlement, theft, or conversion on the part of the taxpayer's employee or other persons contracted with the taxpayer, which the taxpayer could not immediately detect or prevent, provided that reasonable safeguards or internal controls were in place. See (a)(iii)(E) of this subsection.
- (G) The department does not respond to the taxpayer's request for a tax return (or other forms necessary to compute the tax) within a reasonable period of time, which directly causes delinquent filing and payment on the part of the taxpayer. This assumes that, given the same situation, if the department had provided the requested form(s) within a reasonable period of time, the taxpayer would have been able to meet its obligation for timely payment of the tax. In any case, the taxpayer has responsibility to insure that its return is filed in a timely manner (e.g., by keeping track of pending due dates) and must anticipatively request a return for that purpose, if one is not received. (Note: Tax returns and other forms are available at no cost from the department's website, dor.wa.gov. When good cause exists, taxpayers are advised to contact the department and request an extension of the due date for filing, before the due date of concern has passed. See subsection (12) of this rule. Taxpayers who have registered to file electronically with ((e-file)) My DOR will avoid potential penalties relating to paper returns not received. See subsection (1)(b) of this rule.)
- (iii) The following are examples of circumstances that are generally not considered to be beyond the control of the taxpayer and will not qualify for a waiver or cancellation of penalty:
 - (A) Financial hardship;
 - (B) A misunderstanding or lack of knowledge of a tax liability;
- (C) The failure of the taxpayer to receive a tax return form, Ex-CEPT where the taxpayer timely requested the form and it was still not furnished in reasonable time to mail the return and payment by the due date, as described in (a)(ii)(G) of this subsection;
- (D) Registration of an account that is not considered a voluntary registration, as described in subsection (5)(a)(iii) and (b) of this
- (E) Mistakes or misconduct on the part of employees or other persons contracted with the taxpayer (not including conduct covered in (a) (ii) (F) of this subsection); and
- (F) Reliance upon unpublished, written information from the department that was issued to and specifically addresses the circumstances of some other taxpayer.
- (b) Waiver of the late payment of return penalty. The late payment of return penalty (see subsection (5)(a) of this rule) may be waived either as a result of circumstances beyond the control of the taxpayer (RCW 82.32.105 (1) and (a) of this subsection) or after a 24 month review of the taxpayer's reporting history, as described below.

- (i) If the late payment of return penalty is assessed on a return but is not the result of circumstances beyond the control of the taxpayer, the penalty will still be waived or canceled if the following two circumstances are satisfied:
- (A) The taxpayer requests the penalty waiver for a tax return which was required to be filed under RCW 82.32.045 (taxes reported on the combined excise tax return), RCW 82.23B.020 (oil spill response tax), ((RCW 82.27.060 (tax on enhanced food fish),)) RCW 82.29A.050 (leasehold excise tax), RCW 84.33.086 (timber and forest lands), RCW 82.14B.030 (tax on telephone access line use); and
- (B) The taxpayer has timely filed and paid all tax returns due for that specific tax program for a period of 24 months immediately preceding the period covered by the return for which the waiver is being requested. RCW 82.32.105(2).
- If a taxpayer has obtained a tax registration endorsement with the department prior to engaging in business within the state and has engaged in business activities for a period less than 24 months, the taxpayer is eligible for the waiver if the taxpayer had no delinquent tax returns for periods prior to the period covered by the return for which the waiver is being requested. As a result, the taxpayer's very first return due can qualify for a waiver under the 24 month review provision. (See also WAC 458-20-101 for more information regarding the tax registration and tax reporting requirements.) This is the only situation under which the department will consider a waiver when the taxpayer has not timely filed and paid tax returns covering an immediately preceding 24 month period.
- (ii) A return will be considered timely for purpose of the waiver if there is no tax liability on it when it is filed. Also, a return will be considered timely if any late payment penalties assessed on it were waived or canceled due to circumstances beyond the control of the taxpayer (see (a) of this subsection). The number of times penalty has been waived due to circumstances beyond the control of the taxpayer does not influence whether the waiver in this subsection will be granted. A taxpayer may receive more than one of the waivers in this subsection within a 24 month period if returns for more than one of the listed tax programs are filed, but no more than one waiver can be applied to any one tax program in a 24 month period.

For example, a taxpayer files combined excise tax returns as required under RCW 82.32.045, and timber tax returns as required under RCW 84.33.086. This taxpayer may qualify for two waivers of the late payment of return penalty during the same 24 month period, one for each tax program. If this taxpayer had an unwaived late payment of return penalty for the combined excise tax return during the previous 24 month period, the taxpayer may still qualify for a penalty waiver for the timber tax program.

(iii) The 24 month period reviewed for this waiver is not affected by the due date of the return for which the penalty waiver is requested, even if that due date has been extended beyond the original due date.

For example, assume a taxpayer's September 2012 return has had the original due date of October 25th extended to November 25th. The return and payment are received after the November 25th extended due date. A penalty waiver is requested. Since the delinquent return represented the month of September 2012, the 24 months which will be reviewed begin on September 1, 2010, and end with August 31, 2012, (the 24 months prior to September 2012). All of the returns representing that period of time will be included in the review. The extension of

the original due date has no effect on the 24 month period under review.

- (iv) A 24 month review is only valid when considering waiver of the late payment of return penalty described in subsection (5)(a) of this rule. The 24 month review process cannot be used as justification for a waiver of interest, assessment penalty, or any penalty other than the late payment of return penalty.
- (10) Waiver or cancellation of interest. The department will waive or cancel interest imposed under chapter 82.32 RCW only in the following situations:
- (a) The failure to pay the tax prior to issuance of the assessment was the direct result of written instructions given the taxpayer by the department; or
- (b) The extension of the due date for payment of an assessment was not at the request of the taxpayer and was for the sole convenience of the department. RCW 82.32.105(3).
- (11) Interest and penalty waiver for active duty military personnel. RCW 82.32.055 provides a waiver of BOTH interest and penalty imposed under chapter 82.32 RCW when:
 - (a) The majority owner of the business is:
 - (i) On active duty in the military;
 - (ii) Participating in an armed conflict;
- (iii) Assigned to a location outside the territorial boundaries of the United States; and
- (b) The gross income of the business is \$1,000,000 or less for the calendar year immediately prior to the year in which the majority owner is initially deployed outside the United States for the armed conflict.

Interest and penalty may not be waived or canceled for a period longer than 24 months. The waiver applies to interest or penalty based on the date they are imposed, which must be within the 24 month waiver period.

To receive a waiver or cancellation of interest and penalty under this subsection, the taxpayer must submit a copy of the majority owner's deployment orders for deployment outside the territorial boundaries of the United States.

- (12) Stay of collection. RCW 82.32.190 allows the department to initiate a stay of collection, without the request of the taxpayer and without requiring any bond, for certain tax liabilities when they may be affected by the outcome of a question pending before the courts (see (a) of this subsection). RCW 82.32.200 provides conditions under which the department, at its discretion, may allow a taxpayer to file a bond in order to obtain a stay of collection on a tax assessment (see (b) of this subsection). The department will grant a taxpayer's stay of collection request, as described in RCW 82.32.200, only when the department determines that a stay is in the best interests of the state.
- (a) Circumstances under which the department may consider initiating a stay of collection without requiring a bond (RCW 82.32.190) include, but are not necessarily limited to, the existence of the following:
- (i) A constitutional issue to be litigated by the taxpayer, the resolution of which is uncertain;
- (ii) A matter of first impression for which the department has little precedent in administrative practice; or
- (iii) An issue affecting other similarly situated taxpayers for whom the department would be willing to stay collection of the tax.

- (b) The department will give consideration to a request for a stay of collection of an assessment (RCW 82.32.200) if:
- (i) A written request for the stay is made prior to the due date for payment of the assessment; and
- (ii) Payment of any unprotested portion of the assessment and other taxes due is made timely; and
- (iii) The request is accompanied by an offer of a cash bond, or a security bond that is guaranteed by a specified authorized surety insurer. The amount of the bond will generally be equal to the total amount of the assessment, including any penalties and interest. However, where appropriate, the department may require a bond in an increased amount not to exceed twice the amount for which the stay is requested.
- (c) Claims of financial hardship or threat of litigation are not grounds that justify the granting of a stay of collection. However, the department will consider a claim of significant financial hardship as grounds for staying collection procedures, but this will be done only if a partial payment agreement is executed and kept in accordance with the department's procedures and with such security as the department deems necessary.
- (d) If the department grants a stay of collection, the stay will be for a period of no longer than two calendar years from the date of acceptance of the taxpayer request, or 30 days following a decision not appealed from by a tribunal or court of competent jurisdiction upholding the validity of the tax assessed, whichever date occurs first. The department may extend the period of a stay originally granted, but only for good cause shown.
- (e) Interest will continue to accrue against the unpaid tax portion of a liability under stay of collection.
- (13) Extensions. The department, for good cause, may extend the due date for filing any return.
- (a) Any permanent extension more than 10 days beyond the due date, and any temporary extension in excess of 30 days, must be conditional upon deposit by the taxpayer with the department of an amount equal to the estimated tax liability for the reporting period or periods for which the extension is granted. This deposit is credited to the taxpayer's account and may be applied to the taxpayer's liability upon cancellation of the permanent extension or upon reporting of the tax liability where a temporary extension of more than 30 days has been granted.

The amount of the deposit is subject to departmental approval. The amount will be reviewed from time to time, and a change may be required at any time that the department concludes that such amount does not approximate the tax liability for the reporting period or periods for which the extension was granted.

(b) RCW 82.32.080 allows department of revenue to grant extensions of the due date for any taxes due to department of revenue when the governor has proclaimed a state of emergency under RCW ((43.06.040)) 43.06.010. In general, the bill gives department of revenue the authority to provide extensions on its own initiative, or at the specific request of any taxpayers affected by the emergency. The specific details of how, where, and to whom any extensions are granted will depend on the type and scope of each unique emergency and will be determined when an emergency is declared.

WSR 23-23-127 EXPEDITED RULES DEPARTMENT OF REVENUE

[Filed November 16, 2023, 3:13 p.m.]

Title of Rule and Other Identifying Information: WAC 458-20-265 Sales and use tax exemption—Airplane maintenance repair stations.

Purpose of the Proposal and Its Anticipated Effects, Including Any Changes in Existing Rules: The department of revenue (department) intends to update this rule to reflect changes based on 2023 legislation, SHB 1318.

Reasons Supporting Proposal: The Washington legislature enacted statutory amendments to RCW 82.08.025661 and 82.12.025661 and the department is updating this rule to reflect these changes.

Statutory Authority for Adoption: RCW 82.32.300, 82.01.060. Statute Being Implemented: RCW 82.08.025661 and 82.12.025661.

Rule is not necessitated by federal law, federal or state court decision.

Name of Proponent: Governmental.

Name of Agency Personnel Responsible for Drafting: Chelsea Brenegan, 6400 Linderson Way S.W., Tumwater, WA, 360-534-1530; Implementation and Enforcement: John Ryser, 6400 Linderson Way S.W., Tumwater, WA, 360-534-1605.

This notice meets the following criteria to use the expedited adoption process for these rules:

Adopts or incorporates by reference without material change federal statutes or regulations, Washington state statutes, rules of other Washington state agencies, shoreline master programs other than those programs governing shorelines of statewide significance, or, as referenced by Washington state law, national consensus codes that generally establish industry standards, if the material adopted or incorporated regulates the same subject matter and conduct as the adopting or incorporating rule.

Corrects typographical errors, makes address or name changes, or clarifies language of a rule without changing its effect.

Explanation of the Reason the Agency Believes the Expedited Rule-Making Process is Appropriate: The expedited rule-making process is appropriate for this rule because the department is incorporating the changes resulting from 2023 legislation and fixing typographical errors that do not change the effect of the rule.

NOTICE

THIS RULE IS BEING PROPOSED UNDER AN EXPEDITED RULE-MAKING PROC-ESS THAT WILL ELIMINATE THE NEED FOR THE AGENCY TO HOLD PUBLIC HEAR-INGS, PREPARE A SMALL BUSINESS ECONOMIC IMPACT STATEMENT, OR PROVIDE RESPONSES TO THE CRITERIA FOR A SIGNIFICANT LEGISLATIVE RULE. IF YOU OBJECT TO THIS USE OF THE EXPEDITED RULE-MAKING PROCESS, YOU MUST EXPRESS YOUR OBJECTIONS IN WRITING AND THEY MUST BE SENT TO Chelsea Brenegan, Department of Revenue, P.O. Box 47453, Olympia, WA 98504-7453, phone 360-534-1530, fax 360-534-1606, email ChelseaB@dor.wa.gov, AND RECEIVED BY January 22, 2024.

> November 16, 2023 Atif Aziz Rules Coordinator

AMENDATORY SECTION (Amending WSR 22-24-103, filed 12/6/22, effective 1/6/23)

- WAC 458-20-265 Sales and use tax exemption—Airplane maintenance repair stations. (1)(a) Introduction. This rule explains the retail sales and use tax exemption, as described in RCW 82.08.025661 and 82.12.025661, for the construction of airplane maintenance repair stations operated by an eligible maintenance repair operator.
- (b) Other rules that may apply. Readers may also want to refer to additional rules for further information, including the following:
 - (i) WAC 458-20-229 Refunds.
- (ii) WAC 458-20-267 Annual tax performance reports for certain tax preferences.
- (2) **Definitions**. For the purposes of this rule, the following definitions apply:
- (a) "Airplane maintenance repair station" has the same meaning as "repair station" adopted by the National Air Transportation Association and is a maintenance facility that has a certificate issued by the Federal Aviation Administration under Title 14 of the Code of Federal Regulations (14 C.F.R.) Part 145 that is engaged in the maintenance, preventive maintenance, inspection, alteration of airplanes, and alteration of airplane products.
- (b) "Commercial airplane," as defined in RCW 82.32.550(1), is an airplane certified by the Federal Aviation Administration for transporting persons or property, and any military derivative of such an airplane.
- (c) "Component," as defined in RCW 82.32.550(2), means a part or system certified by the Federal Aviation Administration for installation or assembly into a commercial airplane.
- (d) "Eligible maintenance repair operator" means a person classified by the Federal Aviation Administration as qualified to operate a Federal Aviation Regulation Part 145 certified repair station that is located in ((an international)) a commercial services airport owned by a county with a population ((greater than 1,500,000)) less than 1,000,000 or a commercial services airport jointly owned by a city and county.
- (e) "Operationally complete" means constructed to the point of being functionally capable of hosting the repair and maintenance of airplanes.
 - (3) Retail sales or use tax exemption.
- (a) Subject to the requirements of RCW 82.08.025661 and this rule, state and local retail sales and use taxes do not apply to the items and services as described in (b) of this subsection that are charged or sold to, or purchased or used by:
- (i) An eligible maintenance repair operator engaged in the maintenance of airplanes; or
- (ii) A port district, political subdivision, or municipal corporation, if the new airplane maintenance repair station is to be leased to an eligible maintenance repair operator engaged in the maintenance of airplanes.
 - (b) The exempt items and services include:
- (i) Labor and services to construct a new airplane maintenance repair station;

- (ii) Tangible personal property that will be incorporated as an ingredient or component during the course of constructing the new airplane maintenance repair station; and
- (iii) Labor and services to install, during the course of constructing the new airplane maintenance repair station, building fixtures not otherwise eligible for the exemption under RCW 82.08.02565.
- (c) To qualify for the exemption described in this rule, the port district, political subdivision, or municipal corporation must have first entered into an agreement with an eligible maintenance repair operator to build the new facility, prior to starting construction of the new facility.
- (4) Remittance application. The exemption described in this rule is a remittance.
- (a) A business claiming the state and local retail sales or use tax exemption must first pay all applicable state and local retail sales or use taxes on all purchases qualifying for the exemption under subsection (3)(b) of this rule.
- (b) The business may then file a quarterly remittance application with the department for the previously paid retail sales or use tax that is determined by the department to qualify for the exemption. The remittance form may be sent electronically to the department or to the mailing address found in (b)(ii) of this subsection.
- (i) The remittance application must specify and separately identify the amount of the exempted state and local retail sales and use taxes claimed and the qualifying purchases or acquisitions for which the exemption is claimed, along with any supporting documents required by the department. Refer to the department's website at ((https:// www.))dor.wa.gov for documentation requirements.
- (ii) The application for remittance is titled "Application for Refund or Credit" and is available on the department's website at ((https://www.))dor.wa.gov. You may also contact the telephone information center at 360-705-6705 or write to the following address:

Attn: New Construction for FAR Part 145 Repair Station Refunds Taxpayer Account Administration Division Department of Revenue P.O. Box 47476 Olympia, WA 98504-7476

- (c) Local retail sales and use taxes that qualify for this exemption are eligible for remittance ((beginning on the exemption's effective date of July 1, 2016)) the first quarter after construction com-
- (d) State retail sales and use taxes that qualify for this exemption are eligible for remittance the later of either:
- (i) The date on which the airplane maintenance and repair station has been operationally complete for four years; or (ii) December 1, 2021.
- (e) The business must provide written notice to the department when the maintenance and repair station is operationally complete as defined in subsection (2) (e) of this rule. The notice should be sent electronically to the department or to the mailing address found in (b) (ii) of this subsection.
- (f) The state and local retail sales and use taxes described in this rule are not eligible for remittance on purchases of items or services under subsection (3)(b) of this rule that occur on or after the exemption's expiration date of January 1, $((\frac{2027}{1}))$ 2031.

- (5) Department must determine eligibility.
- (a) The department must determine eligibility for the exemption based on information provided by the business and through audit and other administrative records.
- (b) The business must retain, in adequate detail to enable the department to determine whether the equipment or construction meets the criteria under this rule, construction invoices and documents including, but not limited to, invoices, proof of tax paid, and documents describing the location and size of new structures.
- (c) By the end of the calendar quarter that follows the quarter in which the refund application was submitted, the department will remit qualified exempted amounts to a qualifying business, in accordance with subsection (4)(c) and (d) of this rule, for local and state retail sales and use taxes.
- (d) The department may not remit the state portion of the retail sales and use taxes paid if the business did not report at least 100 average employment positions with an average annualized wage of \$80,000 to the employment security department for ((October 1, 2020, through September 30, 2021, with an average annualized wage of \$80,000)) four consecutive calendar quarters, beginning with the first calendar quarter after the date the facility is issued an occupancy permit by the local permit issuing authority. The business must provide the department with the unemployment insurance number provided to the employment security department for verification of employment ((levels)) requirements.
- If a new airplane maintenance repair station owned by a port district, political subdivision, or municipal corporation is leased to an eligible maintenance repair operator engaged in the maintenance of airplanes, only the business lessee, and not the lessor, must meet the employment requirement described in (d) of this subsection.
- (6) Annual tax performance report. An eliqible maintenance repair operator receiving a remittance under this rule must electronically file an annual report with the department in accordance with RCW 82.32.534. For more information about filing an annual report, see WAC 458-20-267 and visit the department's website at ((https:// www.))dor.wa.gov or contact the telephone information center at 360-705-6705.

WSR 23-23-153 EXPEDITED RULES DEPARTMENT OF HEALTH

(Pharmacy Quality Assurance Commission) [Filed November 20, 2023, 5:41 p.m.]

Title of Rule and Other Identifying Information: Citation and technical changes to pharmacy rules in chapter 246-945 WAC. The pharmacy quality assurance commission (commission) is considering amending WAC 246-945-001, 246-945-011, 246-945-014, 246-945-018, 246-945-063, 246-945-156, 246-945-170, 246-945-173, 246-945-175, 246-945-200, 246-945-217, 246-945-230, 246-945-417, and 246-945-590 to remove and replace citations to rules that have been repealed and make general grammatical and technical corrections.

Purpose of the Proposal and Its Anticipated Effects, Including Any Changes in Existing Rules: The commission completed rule making in 2020, consolidating multiple chapters of rules that regulate the practice of pharmacy into one chapter, chapter 246-945 WAC. This proposal will remove citations to repealed WAC chapters, update citations to the current governing WAC chapter or specific rule(s), and make general grammatical corrections without making any material changes.

Reasons Supporting Proposal: Following the rules consolidation project that resulted in the creation of chapter 246-945 WAC in 2020, the commission discovered a number of cross-references that are now outdated. The secretary also finalized updated fees for commission licensees since that time and updates are now needed to correct all fee rule references.

Statutory Authority for Adoption: RCW 18.64.005.

Statute Being Implemented: RCW 18.64.005.

Rule is not necessitated by federal law, federal or state court decision.

Name of Proponent: Washington state pharmacy quality assurance commission, governmental.

Name of Agency Personnel Responsible for Drafting, Implementation, and Enforcement: Joshua Munroe, 111 Israel Road S.E., Tumwater, WA 98501, 360-502-5058.

This notice meets the following criteria to use the expedited adoption process for these rules:

Corrects typographical errors, makes address or name changes, or clarifies language of a rule without changing its effect.

Explanation of the Reason the Agency Believes the Expedited Rule-Making Process is Appropriate: The commission believes the expedited rule-making process is appropriate as the proposed rules fall under RCW 34.05.353 (1)(c) since they "correct typographical errors ... and clarify language of a rule without changing its effect." The updating of references from older repealed sections of rule to the current active sections of rule does not represent a substantive change to any regulation under the commission's jurisdiction.

NOTICE

THIS RULE IS BEING PROPOSED UNDER AN EXPEDITED RULE-MAKING PROC-ESS THAT WILL ELIMINATE THE NEED FOR THE AGENCY TO HOLD PUBLIC HEAR-INGS, PREPARE A SMALL BUSINESS ECONOMIC IMPACT STATEMENT, OR PROVIDE RESPONSES TO THE CRITERIA FOR A SIGNIFICANT LEGISLATIVE RULE. IF YOU OBJECT TO THIS USE OF THE EXPEDITED RULE-MAKING PROCESS, YOU MUST EX-PRESS YOUR OBJECTIONS IN WRITING AND THEY MUST BE SENT TO Joshua Munroe, Pharmacy Quality Assurance Commission, P.O. Box 47852, Olympia,

WA 98504-7852, phone 360-502-5058, email PharmacyRules@doh.wa.gov, https://fortress.wa.gov/doh/policyreview, AND RECEIVED BY January 22, 2024.

> November 20, 2023 Kenneth Kenyon, PharmD, BCPS, Chair Pharmacy Quality Assurance Commission

OTS-4837.4

AMENDATORY SECTION (Amending WSR 20-12-072, filed 6/1/20, effective 7/1/20)

WAC 246-945-001 Definitions. The definitions in chapters 18.64 and 18.64A RCW and those in this section apply throughout this chapter unless otherwise stated.

- (1) "ACPE" means accreditation council for pharmacy education.
- (2) "Active ingredient" means any component that is intended to furnish pharmacological activity or other direct effect in the diagnosis, cure, mitigation, treatment, or prevention of disease, or to affect the structure or any function of the body of humans or other animals. The term includes those components that may undergo chemical change in the manufacture of the drug product and be present in that drug product in a modified form intended to furnish the specified activity or effect.
- (3) "Adulterated" refers to a drug that was produced and the methods used in, or the facilities or controls used for, its manufacture, processing, packing, or holding do not conform to or are not operated or administered in conformity with WAC 246-945-550 as to safety and has the identity and strength, and meets the quality and purity characteristics, which it purports or is represented to possess.
- (4) "Animal control agency" means any agency authorized by law to euthanize or destroy animals; to sedate animals prior to euthanasia or to engage in chemical capture of animals.
- (5) "Approved legend drugs" means any legend drug approved by the commission for use by registered humane societies or animal control agencies for the sole purpose of sedating animals prior to euthanasia, when necessary, and for use in chemical capture programs.
- (6) "Audit trail" means all materials and documents required for the entire process of filling a prescription, which shall be sufficient to document or reconstruct the origin of the prescription, and authorization of subsequent modifications of that prescription.
- (7) "Blood" means whole blood collected from a single donor and processed either for transfusion or further manufacturing.
- (8) "Blood component" means that part of the blood separated by physical or mechanical means.
- (9) "Central fill pharmacy" means a pharmacy contracting with an originating pharmacy, or having the same owner as an originating pharmacy, that provides centralized prescription filling on behalf of the originating pharmacy pursuant to these rules.
- (10) "Chemical capture program" means wildlife management programs registered under RCW 69.41.080 and 69.50.320 to use approved legend drugs and controlled substance for chemical capture. Chemical

capture includes immobilization of individual animals in order for the animals to be moved, treated, examined, or for other legitimate purpo-

- (11) "Collaborative drug therapy agreement" or "CDTA" means a written quideline or protocol previously established and approved by a practitioner authorized to prescribe drugs that enables a pharmacist to exercise prescriptive authority.
- (12) "Controlled substances" has the same meaning as RCW 69.50.101.
- (13) "Controlled substance wholesaler" means a wholesaler licensed under RCW 18.64.046 to possess and sell controlled substances to a licensed pharmacy or other legally licensed or authorized person.
- (14) "Commission" means the pharmacy quality assurance commission.
- (15) "Counterfeit" means a drug which, or the container or labeling of which, without authorization, bears the trademark, trade name, or other identifying mark, imprint, or device, or any likeness thereof, of a drug manufacturer, processor, packer, or distributor other than the person or persons who in fact manufactured, processed, packed, or distributed such drug and which thereby falsely purports or is represented to be the product of, or to have been packed or distributed by, such other drug manufacturer, processor, packer, or distributor.
- (16) "CPE" means continuing pharmacy education accredited by the ACPE.
 - (17) "Consultation" means:
- (a) A communication or deliberation between a pharmacist and a patient, a patient's agent, or a patient's health care provider in which the pharmacist uses professional judgment to provide advice about drug therapy.
- (b) A method by which the pharmacist meets patient information requirements as set forth in WAC 246-945-325.
- (18) "Credential" means a license, certification, or registration under the chapters specified in RCW 18.130.040 issued to a person to practice a regulated health care profession. Whether the credential is a license, certification, or registration is determined by the law regulating the profession.
- (19) "DEA" means the United States Drug Enforcement Administration.
- (20) "Delegated tasks" means tasks that are performed pursuant to a pharmacist's direction, without the exercise of the pharmacy ancillary personnel's own judgment and discretion, and which do not require the pharmacy ancillary personnel's to exercise the independent professional judgment that is the foundation of the practice of the profession of pharmacy.
- (21) "Department" means the Washington state department of health.
- (22) "Dose" means the amount of drug to be administered at one time.
- (23) "Drug(s) of concern" are those drugs identified by the commission as demonstrating a potential for abuse by all professionals licensed to prescribe, dispense, or administer such substances in this
- (24) "Drug price advertising" means the dissemination of nonpromotional information pertaining to the prices of legend or prescription drugs.

- (25) "Drug product" means a finished dosage form (e.g., tablet, capsule, solution) that contains an active drug ingredient generally, but not necessarily, in association with inactive ingredients. The term also includes a finished dosage form that does not contain an active ingredient but is intended to be used as a placebo.
- (26) "Drug sample" means a unit of prescription drug that is not intended to be sold and is intended to promote the sale of the drug.
- (27) "Drug standard and information sources" means industry recognized reference and resources.
- (28) "Drug storage area" means an area where legend drugs, controlled substances, or other restricted items are stored, compounded, or dispensed.
- (29) "Drug utilization review" includes, but is not limited to, the following activities:
- (a) Evaluation of prescriptions and patient records for known allergies, rational therapy-contraindications, appropriate dose, and route of administration and appropriate directions for use;
- (b) Evaluation of prescriptions and patient records for duplication of therapy;
- (c) Evaluation of prescriptions and patient records for interactions between drug-drug, drug-food, drug-disease, and adverse drug re-
- (d) Evaluation of prescriptions and patient records for proper utilization, including over- or under-utilization, and optimum therapeutic outcomes.
- (30) "Electronic means" means an electronic device used to send, receive, ((and/)) or store prescription information, including computers, facsimile machines, etc.
- (31) "Electronic signature" means an electronic sound, symbol, or process attached to or logically associated with a contract or other record and executed or adopted by a person with the intent to sign the record.
- (32) "Enrolled student" means a student who has accepted an offer of admission in writing and the student has made the appropriate deposit securing admission to an accredited school or college of pharma-Cy.
- (33) "Equivalent manager" means an individual authorized to act on behalf of a pharmaceutical firm not licensed as a pharmacy to serve as the primary contact for the department and is responsible for managing the facility operations which includes, but is not limited to, actively involved in and aware of the daily operations of the facility.
- (34) "Export wholesaler" means any wholesaler authorized by the commission to export legend drugs and nonprescription (OTC) drugs to foreign countries.
 - (35) "FDA" United States Food and Drug Administration.
- (36) "Full-line wholesaler" means a drug wholesale distributor that is licensed under RCW 18.64.046 to possess and sell legend drugs, controlled substance and nonprescription drugs to a licensed pharmacy or other legally licensed or authorized person.
- (37) "FPGEC" means foreign pharmacy graduate examination committee.
- (38) "FPGEE" means foreign pharmacy graduate equivalency examination.
- (39) "Generic substitution" means the act of switching between a branded drug and its therapeutically equivalent generic version.

- (40) "HIPAA" means Health Insurance Portability and Accountability Act.
- (41) "Hospital" means any institution licensed under chapter 70.41 or 71.12 RCW or designated under RCW 72.23.020.
- (42) "Hospital pharmacy" means that portion of a hospital licensed under RCW 18.64.043 which is engaged in the manufacture, production, preparation, dispensing, sale, or distribution of drugs, components, biologicals, chemicals, devices and other materials used in the diagnosis and treatment of injury, illness and diseases.
- (43) "Hospital pharmacy associated clinic" or "HPAC" means an individual practitioner's office or multipractitioner clinic owned, operated, or under common control of a parent hospital or health system, where the physical address of the office or clinic is identified on a hospital pharmacy license.
- (44) "Immediate supervision" means supervision by a pharmacist who is immediately available at all times the delegated tasks are being performed; who is aware of delegated tasks being performed; and who provides personal assistance, direction and approval throughout the time the delegated tasks are being performed.
- (a) "Immediately available" means the pharmacist and pharmacy ancillary personnel or interns are on the same physical premises, or if not, technology is used to enable real time, two-way communications between the pharmacist and ((technician(s))) pharmacy ancillary personnel and interns.
- (b) Use of technology: A pharmacist, as an adjunct to assist in the immediate supervision of the pharmacy ancillary personnel or intern, may employ technological means to communicate with or observe the pharmacy ancillary personnel or intern. A pharmacist shall make certain all applicable state and federal laws including, but not limited to, confidentiality, are fully observed when employing technological means of communication and observation. If technology is being used to provide immediate supervision of pharmacy ancillary personnel or intern such technology shall be sufficient to provide the personal assistance, direction and approval required to meet the standard of practice for the delegated tasks.
- (45) "Inoperable" means a credential status indicating that an individual cannot practice because he or she is not actively participating or enrolled in a required training program when this condition is a requirement of the credential. Inoperable status is not the result of enforcement action. The health care professional can resume practice when appropriately enrolled in a required training program and the credential is reactivated.
- (46) "Internal test assessment" means, but is not limited to, conducting those tests of quality assurance necessary to ensure the integrity of the test.
- (47) "Investigational drug" means any article drug that has an investigational drug application (INDA) that has been approved by the
- (48) "Key party" means immediate family members and others who would be reasonably expected to play a significant role in the health care decisions of the patient or client and includes, but is not limited to, the spouse, domestic partner, sibling, parent, child, guardian and person authorized to make health care decisions of the patient or client.
- (49) "Law enforcement" means any general or limited authority Washington peace officer or federal law enforcement officer or tribal officer.

- (50) "License transfer" means the process used by licensed pharmacists to transfer their existing pharmacist license to Washington using NABP's Electronic Licensure Transfer Program \mathbb{R} (e-LTP $^{\text{TM}}$).
- (51) "Lot" means a batch or a specific identified portion of a batch having uniform character and quality within specified limits, or in the case of a drug product produced by continuous process, it is a specific identified amount produced in a unit of time or quantity in a manner that assures it is having uniform character and quality within specified limits.
 - (52) "Manual signature" means a printed or wet signature.
- (53) "Misbranded" applies to all drugs the package or label of which bears any statement, design or device regarding such article or the ingredients or substances contained therein which is false or misleading in any particular way, and drug product which is falsely branded as to the state, territory or country in which it is manufactured or produced.
 - (54) "NABP" means the National Association of Boards of Pharmacy.
 - (55) "NDC" means National Drug Code.
- (56) "Nuclear pharmacy" means a pharmacy providing radiopharmaceutical services.
- (57) "Nuclear pharmacist" means a pharmacist licensed under RCW 18.64.080 who holds an endorsement that meets the requirements of WAC 246-945-180.
- (58) "Originating pharmacy" means a pharmacy that receives a prescription from a patient, the patient's agent, or a prescriber, outsources prescription filling or processing functions to another pharmacy, and ultimately dispenses the prescription drug or device to the patient or the patient's agent. This does not include pharmacies engaged in shared pharmacy services in accordance with RCW 18.64.570.
- (59) "Over-the-counter drugs" or "OTC" means "nonlegend" or "nonprescription" drugs, and any drugs which may be lawfully sold without a prescription.
- (60) "Over-the-counter only wholesaler" means any wholesaler licensed under RCW 18.64.046 to possess and sell OTC drugs to any outlets credentialed for resale.
- (61) "Pharmaceutical firm" means a business engaged in the dispensing, delivering, distributing, manufacturing, or wholesaling of prescription drugs or devices within or into Washington state.
- (62) "Pharmacy intern" means a person who is registered with the commission under RCW 18.64.080(3) as a pharmacy intern.
- (63) "Pharmacy services" means any services provided that meet
- the definition of the practice of pharmacy, RCW 18.64.011. (64) "Plan of correction" is a proposal devised by the applicant or credential holder that includes specific corrective actions that must be taken to correct identified unresolved deficiencies with time frames to complete them.
 - (65) "Precursor drugs" as defined in chapter 69.43 RCW.
- (66) "Prescription drug" means any drug, including any biological product required by federal statute or regulation to be dispensed only by a prescription, including finished dosage forms and bulk drug substances subject to section 503(b) of the Federal Food, Drug, and Cosmetic Act.
- (67) "Protocol" means a written set of procedures, steps or guidance.
 - (68) "Radiopharmaceutical service" means, but is not limited to:

- (a) The preparing, compounding, dispensing, labeling, and delivery of radiopharmaceuticals;
- (b) The participation in radiopharmaceutical selection and radiopharmaceutical utilization reviews;
- (c) The proper and safe storage and distribution of radiopharmaceuticals;
 - (d) The maintenance of radiopharmaceutical quality assurance;
- (e) The responsibility for advising, where necessary or where regulated, of therapeutic values, hazards and use of radiopharmaceuticals; or
- (f) The offering or performing of those acts, services, operations or transactions necessary in the conduct, operation management and control of a nuclear pharmacy.
- (69) "Radiopharmaceutical" means any substance defined as a drug in section 201(g)(1) of the Federal Food, Drug, and Cosmetic Act which exhibits spontaneous disintegration of unstable nuclei with the emission of nuclear particles or photons and includes any nonradioactive reagent kit or nuclide generator which is intended to be used in the preparation of any such substance but does not include drugs such as carbon-containing compounds or potassium-containing salts which contain trace quantities of naturally occurring radionuclides. The term "radioactive drug" includes a "radioactive biological product."
- (70) "Radiopharmaceutical quality assurance" means, but is not limited to, the performance of appropriate chemical, biological and physical tests on radiopharmaceuticals and the interpretation of the resulting data to determine their suitability for use in humans and animals, including internal test assessment authentication of product history and the keeping of proper records.
- (71) "Readily retrievable" means a record that is kept by automatic data processing systems or other electronic, mechanized, or written recordkeeping systems in such a manner that it can be separated out from all other records in a reasonable time.
- (72) "Reverse distributor" means a pharmaceutical wholesaler that receives drugs for destruction, return credit, or otherwise disposes of drugs received from a registrant that holds a credential to dispense or possess drugs.
- (73) "Secretary" means the secretary of the Washington state department of health.
 - (74) "Strength" means:
 - (a) The concentration of the drug product; ((and/)) or
- (b) The potency, that is, the therapeutic activity of the drug product as indicated by appropriate laboratory tests or by adequately developed and controlled clinical data.
- (75) "U.S. jurisdiction" means a state of the United States, the District of Columbia, the Commonwealth of Puerto Rico, or a territory or insular possession subject to the jurisdiction of the United States.
 - (76) "USP" means the United States Pharmacopeia.
- (77) "Therapeutic substitution" means the act of dispensing an alternative drug that is believed to be therapeutically similar but may be chemically different, in a different category, or with different pharmacokinetic properties. This substitution is based on the premise that the substituted drug will provide similar clinical efficacy, desired outcome, and safety profile.
- (78) "TOEFL iBT" means an internet based test which measures the ability to use and understand English. It evaluates the combined use of reading, listening, speaking and writing skills.

- (79) "Virtual manufacturer" means an individual or facility that sells his or her own prescription drugs, but never physically possesses the drugs.
- (80) "Virtual wholesaler" means an individual or facility that sells a prescription drug ((and/)) or device, but never physically possesses the product.
- (81) "Wholesale distribution" means distribution of prescription drugs to persons other than a consumer or patient, but does not include:
- (a) The sale, purchase, or trade of a drug, an offer to sell, purchase or trade a drug, or the dispensing of a drug pursuant to a prescription;
- (b) The lawful distribution of drug samples by manufacturers' representatives or distributors' representatives;
- (c) The sale, purchase, or trade of blood and blood components intended for transfusion;
- (d) Intracompany sales, being defined as any transaction or transfer between any division, subsidiary, parent ((and/)) or affiliated, or related company under the common ownership and control of a corporate entity, unless such transfer occurs between a wholesale distributor and a health care entity or practitioner; or
- (e) The sale, purchase, or trade of a drug or an offer to sell, purchase, or trade a drug for emergency medical reasons, for purposes of this section, "emergency medical reasons" includes transfers of prescription drugs by retail pharmacy to another retail pharmacy or practitioner to alleviate a temporary shortage, except that the gross dollar value of such transfers shall not exceed five percent of the total prescription drug sale revenue of either the transferor or transferee pharmacy during any ((twelve)) 12 consecutive month period.

AMENDATORY SECTION (Amending WSR 20-12-072, filed 6/1/20, effective 7/1/20)

- WAC 246-945-011 Prescription validity. (1) Prior to dispensing and delivering a prescription, a pharmacist shall verify its validity.
 - (2) A prescription shall be considered invalid if:
- (a) At the time of presentation, the prescription shows evidence of alteration, erasure, or addition by any person other than the person who wrote it;
- (b) The prescription does not contain the required information as provided in WAC 246-945-010;
 - (c) The prescription is expired; or
- (d) The prescription is for a controlled substance and does not comply with the requirements in RCW 69.50.308.
 - (3) A prescription is considered expired when:
- (a) The prescription is for a controlled substance listed in Schedule II through V and the date of dispensing is more than six months after the prescription's date of issue.
- (b) The prescription is for a noncontrolled legend drug or ((OTC's)) OTC and the date of dispensing is more than ((twelve)) 12 months after the prescription's date of issue.

AMENDATORY SECTION (Amending WSR 21-17-062, filed 8/11/21, effective 9/11/21)

- WAC 246-945-014 Electronic prescribing mandate waiver. (1) A practitioner may submit an attestation to the department for a waiver from the electronic prescribing mandate in RCW 69.50.312, if the practitioner is experiencing an economic hardship, technological limitations not reasonably in the control of the practitioner, or other exceptional circumstance. A practitioner does not need to submit a waiver if exempted from the mandate under RCW 69.50.312 (2)(a) through (j). A practitioner must submit an attestation for the waiver using forms provided by the department. The department shall deem the waiver granted upon submission of an attestation and the practitioner will be deemed exempt under RCW 69.50.312 (2)(k).
- (2) A practitioner who has submitted an attestation for a waiver from the mandate in RCW 69.50.312 is exempt from the electronic prescribing mandate for the calendar year in which the attestation is signed, beginning with the effective date of this section.
- (a) For economic hardship and ((technical)) technological limitations, a practitioner may attest to the need for a waiver up to three times, giving the practitioner three years to come into compliance with the mandate.
- (b) There is no limit on the number of other exceptional circumstance waivers under subsection (3)(c) of this section that a practitioner can submit.
- (3) A practitioner required to electronically prescribe under RCW 69.50.312 may submit an attestation for a waiver from this mandate due to:
 - (a) Economic hardship in the following circumstances:
- (i) A bankruptcy in the previous year or submitted an attestation for a waiver under this chapter due to a bankruptcy in the previous year;
 - (ii) Opening a new practice after January 1, 2020;
- (iii) Intent to discontinue operating in Washington prior to December 31, 2022; or
- (iv) Operating a low-income clinic, that is defined as a clinic serving a minimum of ((thirty)) 30 percent medicaid patients.
- (b) Technological limitations outside the control of the practitioner if the practitioner is in the process of transitioning to an electronic prescription system.
 - (c) Other exceptional circumstances include:
 - (i) The practitioner is providing services at a free clinic;
- (ii) The practitioner generates fewer than (($\frac{100}{100}$)) $\frac{100}{100}$ prescriptions of Schedules II through V drugs in a one-year period, including both new and refill prescriptions;
- (iii) The practitioner is located in an area without sufficient internet access to comply with the e-prescribing mandate; or
- (iv) Unforeseen circumstances that stress the practitioner or health care system in such a way that compliance is not possible. Examples may include, but are not limited to, natural disasters, widespread health care emergencies, unforeseeable barriers to electronic prescribing, or unforeseen events that result in a statewide emergen-Cy.
- (4) The department may audit waiver attestations submitted by a practitioner to determine compliance with this chapter. Knowingly submitting a false attestation is grounds for disciplinary action against

a practitioner's license by the appropriate disciplinary authority as well as fines pursuant to RCW 69.50.312(5).

AMENDATORY SECTION (Amending WSR 20-12-072, filed 6/1/20, effective 7/1/20)

- WAC 246-945-018 Prescriptions—Labeling—Prepackage medications. Prepackage medications dispensed pursuant to RCW 70.41.480, medications dispensed in unit dose form, and medications dispensed by a pharmacy to a long-term care facility must include a label with the following information:
 - (1) Drug name;
 - (2) Drug strength;
 - (3) Expiration date in accordance with WAC 246-945-016(3);
- (4) The manufacturer's name and lot number, if not maintained in a separate record; and
- (5) The identity of the pharmacist or provider responsible for the prepackaging, if not maintained in a separate record.

AMENDATORY SECTION (Amending WSR 20-12-072, filed 6/1/20, effective 7/1/20)

- WAC 246-945-063 Precursor definitions. The definitions in this section apply to WAC 246-945-065 through 246-945-088.
- (1) "((Registered)) Restricted product" means any nonprescription product containing any detectable quantity of ephedrine, pseudoephedrine, and phenylpropanolamine or their salts or isomers, or salts of isomers.
- (2) "Retailer" means a pharmacy licensed by, or shopkeeper or itinerant vendor registered with, the department of health under chapter 18.64 RCW that sells, dispenses, or otherwise provides restricted products to purchasers.
- (3) "Sale" means the transfer, selling, or otherwise furnishing of any restricted product to any person.

AMENDATORY SECTION (Amending WSR 20-12-072, filed 6/1/20, effective 7/1/20)

- WAC 246-945-156 Pharmacy intern—Temporary practice permit. (1) An individual that holds a pharmacy intern registration in another U.S jurisdiction, that has registration standards substantially equivalent to Washington, may request a temporary practice permit if:
- (a) The applicant is not subject to denial of a credential or issuance of a conditional or restricted credential in any state;
 - (b) Does not have a criminal record in Washington state;
- (c) The applicant's fingerprint-based national background check results are pending; and
 - (d) The applicant meets WAC 246-945-155 (1)(a) or (b).
- (2) To request a temporary practice permit, the pharmacy intern applicant shall submit a written request for a temporary practice per-

mit, and any applicable fees in accordance with ((chapter 246-907)) WAC 246-945-990 through 246-945-992.

- (3) A temporary practice permit expires:
- (a) When the pharmacy intern registration is issued;
- (b) When a notice of decision on the pharmacy intern registration application is mailed to the applicant; or
- (c) Ninety days after the temporary practice permit is issued. The applicant may obtain a one-time extension of up to ((ninety)) 90 days with approval of the commission.

AMENDATORY SECTION (Amending WSR 20-12-072, filed 6/1/20, effective 7/1/20)

WAC 246-945-170 Pharmacist licensure by license transfer—Temporary practice permits. (1) An individual who holds an active pharmacist license, in good standing, issued by another U.S. jurisdiction may apply for a pharmacist license in Washington by license transfer. In addition to the completion of the commission's application, the applicant must:

- (a) File for license transfer using the NABP eLTP process; and
- (b) Take and pass the approved jurisprudence examination.
- (2) A temporary practice permit to practice pharmacy may be issued to an applicant for a pharmacist license by license transfer if the applicant meets all of the requirements and qualifications in subsection (1) of this section, and the following criteria are met:
- (a) The applicant is not subject to denial of a credential or issuance of a conditional or restricted credential in any U.S. jurisdiction;
 - (b) Does not have a criminal record in Washington state;
- (c) The applicant's fingerprint-based national background check results are pending; and
- (d) To request a temporary practice permit, the applicant shall submit a written request for a temporary practice permit, and pay the applicable fees in accordance with (($\frac{\text{chapter } 246-907}{\text{chapter } 246-907}$)) WAC $\frac{246-945-990}{\text{chapter } 246-907}$ through 246-945-992.
 - (3) A temporary practice permit expires:
 - (a) When the pharmacist license is issued;
- (b) When a notice of decision on the pharmacist license application is mailed to the applicant; or
- (c) One hundred eighty days after the temporary practice permit is issued. The applicant may obtain a one-time extension of ((one hundred eighty)) 180 days with approval of the commission.
- (4) A temporary practice permit holder cannot qualify as a responsible pharmacy manager.

AMENDATORY SECTION (Amending WSR 20-12-072, filed 6/1/20, effective 7/1/20)

WAC 246-945-173 Expired pharmacist license. To return to active status a pharmacist with an expired license shall pay the applicable fees in accordance with ((chapter 246-907)) WAC 246-945-990 through 246-945-992 and:

- (1) If the pharmacist license has been expired for less than three years the pharmacist shall meet the requirements of ((chapter 246-12 WAC, Part 2)) WAC 246-12-040 and ((fifteen)) 15 CPE hours per year the license has been expired.
- (2) If the pharmacist license has been expired for three years or more, and the pharmacist holds an active credential in another U.S. jurisdiction, and is in good standing, the pharmacist shall:
- (a) Meet the requirements ((in chapter 246-12 WAC, Part 2)) of WAC 246-12-040;
- (b) Provide certification of an active pharmacist license which includes:
 - (i) Name and license number;
 - (ii) Issue and expiration date; and
- (iii) Verification that the license has not been the subject of final or pending disciplinary action.
- (c) Submit verification of current active pharmacy practice from another U.S. jurisdiction; and
- (d) Take and pass the commission approved jurisprudence examination.
- (3) If a pharmacist license has been expired for three years or more, and the pharmacist has not been in active practice in another U.S. jurisdiction, the pharmacist shall:
- (a) Meet the requirements of ((chapter 246-12 WAC, Part 2)) <u>WAC</u> 246-12-040;
- (b) Serve an internship of ((three hundred)) 300 hours in compliance with WAC 246-945-163; and
- (c) Take and pass the commission approved jurisprudence and licensure examinations.

AMENDATORY SECTION (Amending WSR 20-12-072, filed 6/1/20, effective 7/1/20)

- WAC 246-945-175 Inactive pharmacist license. (1) A pharmacist may obtain an inactive license by meeting the requirements of WAC 246-12-090 and RCW 18.64.140.
- (2) An inactive license can be renewed in accordance with ((chapter 246-12)) WAC 246-12-100 and by paying the applicable fees in accordance with WAC 246-945-990 through 246-945-992.
- (3) If a license is inactive for three years or less, to return to active status a pharmacist shall meet the requirements of ((chapter 246-12 WAC, Part 4)) WAC 246-12-110.
- (4) If a license is inactive for more than three years, and the pharmacist has been in active practice in another U.S. jurisdiction, to return to active status the pharmacist must:
- (a) Provide certification of an active pharmacist license which includes:
 - (i) Name and license number;
 - (ii) Issue and expiration date; and
- (iii) Verification that the license has not been the subject of final or pending disciplinary action.
- (b) Submit verification of current active pharmacy from another U.S. jurisdiction;
- (c) Meet the requirements of ((chapter 246-12 WAC, Part 4)) WAC 246-12-110; and

- (d) Take and pass the commission approved jurisprudence examination.
- (5) If a pharmacist license has been inactive for more than three years, and the pharmacist has not been in active practice in another U.S. jurisdiction, to return to active status, the pharmacist shall comply with the requirements of WAC 246-945-173(3).

AMENDATORY SECTION (Amending WSR 23-09-062, filed 4/18/23, effective 5/19/23)

- WAC 246-945-200 Pharmacy assistants. (1) To become registered as a pharmacy assistant an applicant shall submit an application to the commission that meets the requirements of WAC 246-12-020.
- (2) The supervising pharmacist, shall instruct the pharmacy assistant regarding their scope of practice.
- (3) To renew a registration a pharmacy assistant shall submit an application to the commission with the applicable fees in accordance with WAC 246-945-990 through 246-945-992.

AMENDATORY SECTION (Amending WSR 20-12-072, filed 6/1/20, effective 7/1/20)

- WAC 246-945-217 Expired pharmacy technician certification. To return to active status a pharmacy technician with an expired certification shall pay the applicable fees in accordance with ((chapter 246-907)) WAC 246-945-990 through 246-945-992, and:
- (1) If a pharmacy technician's certification has expired for five years or less, the pharmacy technician shall meet the requirements of ((chapter 246-12 WAC, Part 2)) WAC 246-12-040.
- (2) If the pharmacy technician's certification has expired for over five years and they have not been in active practice in another U.S. jurisdiction, the pharmacy technician shall:
- (a) Complete the requirements for certification under WAC 246-945-205; and
- (b) Meet the requirements of ((chapter 246-12 WAC, Part 2)) <u>WAC</u> 246-12-040.
- (3) If the pharmacy technician's certification has expired for over five years and they have been in an active practice in another U.S. jurisdiction with duties that are substantially equivalent to a pharmacy technician in Washington state, the pharmacy technician shall:
- (a) Submit verification of current active pharmacy practice in another U.S. jurisdiction; and
- (b) Meet the requirements of ((chapter 246-12 WAC, Part 2)) WAC 246-12-040.

AMENDATORY SECTION (Amending WSR 20-12-072, filed 6/1/20, effective 7/1/20)

WAC 246-945-230 General information, change of location, ownership or new construction. (1) The definitions in this subsection apply throughout WAC 246-945-230 through 246-945-247 unless otherwise specified:

- (a) "License" includes "licensing," "licensure," "certificate," "certification," and "registration."
- (b) "Facility" includes pharmacies, nonresident pharmacies, health care entities, hospital pharmacy associated clinics, wholesalers, and manufacturers.
 - (2) The commission shall license a facility that:
- (a) Submits a completed application for the license applied for on forms provided by the commission;
- (b) Pays the applicable fees in accordance with ((chapter 246-907)) WAC 246-945-990 through 246-945-992. This fee will not be prorated under any circumstances;
- (c) Undergoes an inspection by the commission if the facility is located in Washington pursuant to WAC 246-945-005 that results in either no deficiencies or an approved plan of correction; and
- (d) Obtains a controlled substances registration from the commission and is registered with the DEA if the facility intends to possess or distribute controlled substances.
 - (3) Once an initial license is issued, a licensed facility must:
- (a) Notify the commission and pay a facility inspection fee in lieu of paying an ((original)) initial license fee for modifications or remodels. A modification or remodel of a pharmacy location includes changes to a previously approved area, room or pharmacy building which result in changes in the pharmacy that affects security, square footage, access to drugs, compounding or necessitates temporary relocation of pharmacy services.
- (b) Submit a new application on forms provided by the commission and pay the (($\frac{\text{original}}{\text{original}}$)) initial license fee as established in (($\frac{\text{chap-ter }246-907}{\text{chap-907}}$)) WAC $\frac{246-945-990}{\text{original}}$ through $\frac{246-945-992}{\text{original}}$ if the facility changes location to a different address. If located in Washington, a facility may not relocate prior to the inspection of the new premises.
- (c) Notify the commission and pay the ((original)) initial license fee in accordance with ((chapter 246-907)) WAC 246-945-990through 246-945-992 whenever there is a change of ownership. Change in ownership includes changes in business or organizational structure such as a change from sole proprietorship to a corporation, or a change of more than ((fifty)) <u>50</u> percent ownership in a corporation.
- (i) Upon receipt of a change of ownership application and fees, the purchaser may begin operations prior to the issuance of a new pharmacy license only when the purchaser and seller have a written power of attorney agreement. This agreement shall delineate that violations during the pending application process shall be the sole responsibility of the seller.
- (ii) This agreement shall be provided to the commission upon request.
- (d) Notify the commission within ((thirty)) 30 days of any changes to the information provided on their application.
- (e) Notify the commission of any changes in their responsible pharmacy manager in accordance with WAC 246-945-480, if a responsible pharmacy manager is required for initial licensure.
- (f) Renew their license in accordance with ((chapter 246-907)) WAC 246-945-990 through 246-945-992.
 - (4) A license is issued to a location and is not transferable.

AMENDATORY SECTION (Amending WSR 20-12-072, filed 6/1/20, effective 7/1/20)

- WAC 246-945-417 Electronic systems for patient medication records, prescriptions, chart orders, and controlled substance records. (1) A pharmacy shall use an electronic recordkeeping system to establish and store patient medication records, including patient aller-
- gies, idiosyncrasies or chronic conditions, and prescription, refill, transfer information, and other information necessary to provide safe and appropriate patient care.
- (a) Systems must prevent auto-population of user identification information.
- (b) Pharmacies that provide off-site pharmacy services without a pharmacist for product fulfillment or prescription processing must track the identity of each individual involved in each step of the off-site pharmacy services.
- (2) The electronic recordkeeping system must be capable of realtime retrieval of information pertaining to the ordering, verification, and processing of the prescription where possible.
- (3) The electronic recordkeeping system must include security features to protect the confidentiality and integrity of patient records including:
- (a) Safeguards designed to prevent and detect unauthorized access, modification, or manipulation of prescription information and patient medication records; and
- (b) Functionality that documents any alteration of prescription information after a prescription is dispensed, including the identification of the individual responsible for the alteration.
- (4) The pharmacy shall have policies and procedures in place for system downtime.
- (a) The procedure shall provide for the maintenance of all patient recordkeeping information as required by this chapter.
- (b) Upon restoration of operation of the electronic recordkeeping system the information placed in the auxiliary recordkeeping procedure shall be entered in each patient's records within two working days, after which the auxiliary records may be destroyed.
- (c) This section does not require that a permanent dual recordkeeping system be maintained.
- (5) The pharmacy shall maintain records in accordance with WAC 246-945-020.
- (6) Electronic prescriptions for prescription drugs must be maintained by the pharmacy in a system that meets the requirements of 21 C.F.R. Sec. 1311.
- (7) HCEs or HPACs that maintain an electronic record system must be done in accordance with subsections $((\frac{(2)}{2}))$ (1) through $((\frac{(7)}{2}))$ (6) of this section.

AMENDATORY SECTION (Amending WSR 20-12-072, filed 6/1/20, effective 7/1/20)

WAC 246-945-590 Wholesaler—Policies and procedures. Wholesalers shall establish, maintain, and adhere to written policies and procedures, which shall be followed for the receipt, security, storage, inventory, transport, and shipping and wholesale distribution of

drugs, including policies and procedures for identifying, recording, and reporting losses or thefts and for correcting all errors and inaccuracies in inventories. Wholesalers shall include the following in their written policies and procedures:

- (1) A procedure to be followed for handling recalls and withdrawals of drugs. Such procedure shall be adequate to deal with recalls and withdrawals due to:
- (a) Any action initiated at the request of FDA or any other federal, state, or local law enforcement or other government agency, including the commission; or
- (b) Any volunteer action by the manufacturer to remove defective or potentially defective drugs from the market.
- (2) A procedure to ensure that wholesalers prepare for, protect against, and handle any crisis that affects security or operation of any facility in the event of a strike, fire, flood, or other natural disaster, or other situations of local, state, or national emergency.
- (3) A procedure to ensure that any outdated drugs shall be segregated from other drugs and either returned to the manufacturer or destroyed in accordance with federal and state laws, including all necessary documentation and the appropriate witnessing. This procedure shall provide for written documentation of the disposition of outdated
- (4) A procedure for the destruction of outdated drugs in accordance with federal and state laws.
- (5) A procedure for the disposing and destruction of containers, labels, and packaging to ensure that the containers, labels, and packaging cannot be used in counterfeiting activities, including all necessary documentation, and the appropriate witnessing of the destruction of any labels, packaging, immediate containers, or containers in accordance with all applicable federal and state requirements.
- (6) A procedure for identifying, investigating, and reporting significant drug inventory discrepancies involving counterfeit, suspect of being counterfeit, contraband, or suspect of being contraband, in the inventory and reporting of such discrepancies ((as required to the FDA, commission and/or appropriate federal or state agency)) to the FDA, commission, and, as applicable, the DEA upon discovery of such discrepancies.
- (7) A procedure for reporting criminal or suspected criminal activities involving the inventory of drug(s) as required to the commission, FDA, and if applicable, DEA.
 - (8) Procedures addressing:
- (a) The design and operation of the suspicious order monitoring and reporting system;
- (b) Mandatory annual training for staff responsible for identifying and reporting suspicious orders and potential diversion activities. Such training must include the following:
 - (i) The wholesaler's suspicious order monitoring system;
- (ii) The process to collect all relevant information on customers in accordance with WAC ((246-960-330)) 246-945-585; and
- (iii) The requirement and process for submission of suspicious order and information on customers who engage in potential diversion activities.
- (9) A procedure for timely responding to customers who submit purchase orders for patients with emergent needs.

Washington State Register, Issue 23-23

WSR 23-23-159 EXPEDITED RULES DEPARTMENT OF REVENUE

[Filed November 21, 2023, 10:25 a.m.]

Title of Rule and Other Identifying Information: WAC 458-20-134 Commercial or industrial use and 458-20-178 Use tax and the use of tangible personal property.

Purpose of the Proposal and Its Anticipated Effects, Including Any Changes in Existing Rules: The department is updating WAC 458-20-134 and 458-20-178 due to changes from recent legislation, SHB 1764 (2023). The legislation amended RCW 82.04.450 to provide a valuation method for asphalt and aggregate used in public road construction.

Additionally, the update includes several grammar and punctuation changes to enhance clarity and readability.

Reasons Supporting Proposal: The update is to conform the rule to SHB 1764, which passed during the 2023 legislative session.

Statutory Authority for Adoption: RCW 82.32.300, 82.01.060.

Statute Being Implemented: RCW 82.12.010 and 82.04.450.

Rule is not necessitated by federal law, federal or state court decision.

Name of Proponent: Department of revenue, governmental.

Name of Agency Personnel Responsible for Drafting: Brett Grannemann, 6400 Linderson Way S.W., Tumwater, WA, 360-534-1532; Implementation and Enforcement: John Ryser, 6400 Linderson Way S.W., Tumwater, WA, 360-534-1605.

This notice meets the following criteria to use the expedited adoption process for these rules:

Adopts or incorporates by reference without material change federal statutes or regulations, Washington state statutes, rules of other Washington state agencies, shoreline master programs other than those programs governing shorelines of statewide significance, or, as referenced by Washington state law, national consensus codes that generally establish industry standards, if the material adopted or incorporated regulates the same subject matter and conduct as the adopting or incorporating rule.

Corrects typographical errors, makes address or name changes, or clarifies language of a rule without changing its effect.

Explanation of the Reason the Agency Believes the Expedited Rule-Making Process is Appropriate: The expedited rule-making process is appropriate for this rule update because the department is incorporating changes resulting from 2023 legislation.

NOTICE

THIS RULE IS BEING PROPOSED UNDER AN EXPEDITED RULE-MAKING PROC-ESS THAT WILL ELIMINATE THE NEED FOR THE AGENCY TO HOLD PUBLIC HEAR-INGS, PREPARE A SMALL BUSINESS ECONOMIC IMPACT STATEMENT, OR PROVIDE RESPONSES TO THE CRITERIA FOR A SIGNIFICANT LEGISLATIVE RULE. IF YOU OBJECT TO THIS USE OF THE EXPEDITED RULE-MAKING PROCESS, YOU MUST EX-PRESS YOUR OBJECTIONS IN WRITING AND THEY MUST BE SENT TO Brett Grannemann, Department of Revenue, P.O. Box 47453, Olympia, WA 98504-7453, phone 360-534-1532, fax 360-534-1606, email brettg@dor.wa.gov, AND RE-CEIVED BY January 22, 2024.

> November 21, 2023 Atif Aziz

OTS-5089.1

AMENDATORY SECTION (Amending WSR 22-04-026, filed 1/24/22, effective 2/24/22)

WAC 458-20-134 Commercial or industrial use. (1) Definitions.

- (a) The term "commercial or industrial use" means the following uses of products, including by-products, by the same person that extracted or manufactured them:
 - (i) Any use as a consumer; and
- (ii) ((The)) Manufacturing of articles, substances, or commodities. (RCW 82.04.130.)
- (b) The term "biomass fuel" means wood waste and other wood residuals, including forest derived biomass, but does not include firewood or wood pellets. "Biomass fuel" also includes partially organic by-products of pulp, paper, and wood manufacturing processes.
- (2) Examples of commercial or industrial use. The following are examples of commercial or industrial use:
- (a) The use of lumber by the manufacturer of that lumber to build a shed for its own use.
- (b) The use of a motor truck by the manufacturer of that truck as a service truck for itself.
- (c) The use by a boat manufacturer of patterns, jigs, and dies ((which)) that it has manufactured.
- (d) The use by a contractor building or improving a publicly owned road of crushed rock or pit run gravel ((which)) that it has extracted.
- (3) Business and occupation tax. Persons manufacturing or extracting tangible personal property for commercial or industrial use are subject to tax under the manufacturing or extracting B&O tax classifications, as the case may be. The tax is measured by the value of the product manufactured or extracted and used. See WAC 458-20-112 Value of products, for additional information.
- (4) Use tax. Persons manufacturing or extracting tangible personal property for commercial or industrial use are subject to use tax on the value of the article used, unless a specific exemption is provided. See WAC 458-20-178 Use tax and the use of tangible personal property, for further explanation of use tax and the definition of "value of the article used."
- (5) Exemptions. The following uses of articles produced for commercial or industrial use are expressly exempt from use tax.
- (a) RCW 82.12.0263 exempts from the use tax the use of biomass fuel by the same person that extracted or manufactured that biomass fuel when it is used directly in the operation of the particular extractive operation or manufacturing plant ((which)) that produced or manufactured the same biomass fuel.
- (b) Property produced for use in manufacturing ferrosilicon, which is subsequently used to make magnesium for sale, is exempt from use tax if the primary purpose is to create a chemical reaction directly through contact with an ingredient of ferrosilicon. RCW 82.04.190(1).

- (c) Hog fuel used to produce electricity, steam, heat, or biofuel is exempt from use tax. RCW 82.12.956. For the purposes of this exemption, "hog fuel" means wood waste and other wood residuals including forest derived biomass, but not including firewood or wood pellets. "Biofuel" means a liquid or gaseous fuel derived from organic matter intended for use as a transportation fuel including, but not limited to, biodiesel, renewable diesel, ethanol, renewable natural gas, and renewable propane.
- (6) Special provisions regarding value of article used. RCW 82.12.010 provides the following special valuation provisions to persons manufacturing products for commercial or industrial use:
- (a) In the case of articles manufactured or produced by the user and used in the manufacture or production of products sold or to be sold to the United States Department of Defense, the value of the articles used is determined according to the value of the ingredients of those articles.
- (b) In the case of an article manufactured or produced for purposes of serving as a prototype for the development of a new or improved product, the value of the article used is determined by:
- (i) The retail selling price of such new or improved product when first offered for sale; or
- (ii) The value of materials incorporated into the prototype in cases in which the new or improved product is not offered for sale.
- (c) In the case of a person manufacturing or extracting asphalt or aggregates used in providing services taxable under RCW 82.04.280 (1) (b), the value of the asphalt or aggregates used is based on cost. Specifically, the value of the asphalt or aggregates equals the sum of all direct and indirect costs attributable to the asphalt or aggregates used, plus a public road construction market adjustment of five percent of those costs.

OTS-5090.1

AMENDATORY SECTION (Amending WSR 23-14-002, filed 6/21/23, effective 7/22/23)

- WAC 458-20-178 Use tax and the use of tangible personal property. (1) Introduction. This rule provides general use tax-reporting information for consumers. It discusses who is responsible for remitting use tax, and when and how to remit the tax. The rule also explains the imposition of use tax as it applies to the use of tangible personal property within this state when the acquisition of the tangible personal property was not subject to retail sales or deferred sales tax.
- (a) **Examples.** Examples found in this rule identify a number of facts and then state a conclusion. These examples should be used only as a general guide. The tax results of other situations must be determined after a review of all of the facts and circumstances.
- (b) Additional information available. For information on use tax exemptions please refer to chapter 82.12 RCW. When appropriate, this rule refers the reader to applicable statutes and rules. In addition, the reader may wish to refer to the following:

- (i) WAC ((458-20-145, Local sales and use tax, provides information on sourcing local sales and use taxes.
- (ii))) 458-20-112, Value of products, provides information on the measure of tax for certain sales.
- (ii) WAC 458-20-145, Local sales and use tax, provides information on sourcing local sales and use taxes.
- (iii) WAC 458-20-15503, Digital products, provides information on sales and use tax liability on digital products such as: Digital goods, including digital audio works, digital audio-visual works, and digital books; digital automated services; digital codes used to obtain digital goods or digital automated services; and remote-access software.
- (((iii))) (iv) WAC 458-20-169, Nonprofit organizations, provides information on a use tax exemption for donated items to a nonprofit charitable organization.
- $((\frac{1}{2}))$ (v) WAC 458-20-17803, Use tax on promotional material, provides information about the use tax reporting responsibilities of persons who distribute or cause the distribution of promotional material, except newspapers, the primary purpose of which is to promote the sale of products or services in Washington.
- $((\frac{(v)}{(v)}))$ <u>(vi)</u> WAC 458-20-190, Sales to and by the United States— Doing business on federal reservations—Sales to foreign governments, provides tax reporting information for businesses doing business with the United States.
- $((\frac{(vi)}{(vi)}))$ (vii) WAC 458-20-192, Indians—Indian country, provides information on use tax pertaining to Indians and Indian tribes and use tax pertaining to non-Indians in Indian country.
- (((vii))) <u>(viii)</u> WAC 458-20-257, Warranties and service contracts, provides information on tax responsibilities of persons selling or performing services covered by warranties, service contracts, and mixed agreements for tangible personal property.
- (2) What is use tax? Use tax complements the retail sales tax, and in most cases mirrors the retail sales tax. Articles of tangible personal property used or certain services purchased in Washington are subject to use tax when the state's retail sales tax has not been paid, or where an exemption is not available. Tangible personal property or services used or purchased by the user in any manner are taxable including, but not limited to:
 - Purchases directly from out-of-state sellers;
 - Purchases through the internet, telemarketing, mail order; or
 - Acquisitions at casual or isolated sales.
- (a) Example 1. ABC Company (ABC) orders office supplies from outof-state vendors and also through catalogs. In addition, ABC pays annual subscriptions for magazines for their own use. None of these vendors is required to collect Washington's retail sales tax. Use tax is due on all taxable items ordered including the annual subscriptions.
- (b) Example 2. Mary is a music instructor that teaches adults how to play the piano. Mary does not charge her students retail sales tax on the costs of the weekly piano lessons. Use tax is not due on the lessons, as the lessons are not a retail sales taxable service. See WAC 458-20-224, Service and other business activities.
- (3) "Use" defined. For purposes of this rule, "use," "used," "using," or "put to use" have their ordinary meaning and include the first act by which a person takes or assumes dominion or control over the article (as a consumer). It includes installation, storage, withdrawal from storage, or any other act preparatory to subsequent actual

use or consumption within the state. (See RCW 82.12.010.) Multiple uses of the same article by the same person do not generally result in multiple use tax liabilities.

- (4) Measure of tax Value of article used. Use tax generally is levied and collected on an amount equal to the value of the article used by the taxpayer. RCW 82.12.010 defines this value to generally be the purchase price of the article. There are a number of specific situations where this value may be different than the amount of consideration paid or given by the buyer to the seller. See subsection (7) of this rule for exceptions.
- (a) When the value is the purchase price. The term "purchase price" has the same meaning as "selling price." The selling price is the total amount of consideration, except trade-in property of like kind, including cash, credit, property, and services, for which tangible personal property is sold, leased, or rented, valued in money, whether received in money or otherwise. The selling price, and therefore the "value of the article used" also includes delivery charges. Delivery charges are charges made by the seller for preparing and delivering tangible personal property to a location designated by the buyer and include, but is not limited to, charges for transportation, shipping, postage, handling, crating, and packing. (See RCW 82.08.010 and 82.12.010.)
- (b) When the purchase price does not represent true value. When an article is sold under conditions in which the purchase price does not represent the true value, the "value of the article used" is to be determined as nearly as possible according to the retail selling price at place of use of similar products of like quality and character. (See RCW 82.12.010.) This is frequently referred to as the fair market value of the property. For additional information regarding the measure of tax for articles in these situations, refer to WAC 458-20-112, Value of products. Refer to subsection (4)(i)(i) of this rule for determining use tax when there is no similar article of like quality and character.

A comparison ((+)) and examination of arm's length sales transactions is required when determining the value of the article used on the basis of the retail selling price of similar products. An arm's length sale generally involves a transaction negotiated by unrelated parties, each acting in his or her own self-interest.

- (i) In an arm's length sales transaction, the value placed on the property by the parties to the transaction may be persuasive evidence of the true value of the property. Where there is a conflict regarding the true value of tangible personal property between sales documents, entries in the accounting records (($\frac{and}{}$)), or value reported for use tax purposes, the department often looks to the person's accounting records as an indication of the minimum value of capitalized property. Neither the department nor the taxpayer is necessarily bound by this value if it is established that the entry in the books of account does not fairly represent the true value of the article used.
- (ii) Some arm's length sales transactions involve multiple pieces of property or different types of property (such as when both real and personal property are sold). While the total sales price may represent a true value for the property in total, the values allocated to the specific components may not in and of themselves represent true values for those components. This is especially apparent when the values assigned by the parties to the sales transaction vary from those entered into the accounting records ((and/)) or reported for use tax purposes. In such cases, the value of the article used for the purpose of the

use tax must be determined as nearly as possible according to the retail selling price, at the place of use, of similar products of like quality and character.

- (c) Property acquired and used outside Washington before use occurs in Washington. The purchase price of property acquired and used outside Washington before being used in this state may not represent the property's true value. Under these circumstances, the value of article used is the retail selling price at place of use of similar products of like quality and character as of the time the article is first used in Washington.
- (d) Imported property. When property is imported from outside the United States for use in Washington state, the value of the article used includes any amount of tariff or duty paid with respect to importation.
- (e) Articles produced for commercial or industrial use. A person who extracts or manufactures products or by-products for commercial or industrial use is subject to use tax and the business and occupation (B&O) tax on the value of products or by-products used. "Commercial or industrial use" is the use of products, including by-products, as a consumer by the person who extracted or manufactured the products or by-products. See WAC 458-20-134, Commercial or industrial use and WAC 458-20-136, Manufacturing, processing for hire, fabricating.

Tax applies even if the person is not generally in the business of extracting, producing, or manufacturing the products, or the extracting or manufacturing activity is incidental to the person's primary business activity. Thus, a clothing retailer who manufactures signs or other materials for display purposes incurs a liability even though the clothing retailer is not otherwise in the business of manufacturing signs and other display materials for sale.

- (i) The extractor or manufacturer is responsible for remitting retail sales or use tax on all materials used while developing or producing an article for commercial or industrial use. This includes materials that are not components of the completed article.
- (ii) The value of the extracted or manufactured article is subject to use tax when the article is completed and used. The measure of use tax due for the completed article may be reduced by the value of any materials actually incorporated into that article if the manufacturer or extractor previously paid sales or use tax on the materials. See subsection (4)(g) of this rule for an explanation of the measure of tax for a completed prototype.
- (f) Bailment. For property acquired by bailment, the "value of the article used" for the bailee is an amount representing a reasonable rental for the use of the bailed article, determined as nearly as possible according to the value of such use at the places of use of similar products of like quality and character. (See RCW 82.12.010.) If the nature of the article is such that it can only be used once, the reasonable rental value is the full value of the article used. See also WAC 458-20-211, Leases or rentals of tangible personal property, bailments.
- (g) Prototypes. The value of the article used with respect to an article manufactured or produced for purposes of serving as a prototype for the development of a new or improved product is:
- The retail selling price of such new or improved product when first offered for sale; or
- The value of materials incorporated into the prototype in cases where the new or improved product is not offered for sale. (See RCW 82.12.010.)

- (h) Articles manufactured and used in the production of products for the department of defense. When articles are manufactured and used in the production of products for the department of defense, use tax is due except where there is an exemption. The value of the article used with respect to an article manufactured or produced by the user and used in the manufacture or production of products sold or to be sold to the department of defense of the United States is the value of the ingredients of the manufactured or produced article. (See RCW 82.12.010.) However, refer to WAC 458-20-13601, Manufacturers and processors for hire—Sales and use tax exemption for machinery and equipment_L to determine if such articles qualify for exemption under RCW 82.12.02565.
- (i) Property temporarily brought into Washington for business use. In the case of articles owned by a user engaged in business outside the state which are brought into the state for no more than 180 days in any period of 365 consecutive days and which are temporarily used for business purposes by the person in this state, the value of the article used must be an amount representing a reasonable rental for the use of the articles, unless the person has paid tax under chapter 82.08 or 82.12 RCW upon the full value of the article used.

However, this measure of "value of article used" is a separate provision from RCW 82.12.0251 use tax exemption. The use tax exemption is provided to nonresidents bringing property into Washington for his or her use or enjoyment while temporarily within the state, unless the property is used in conducting a nontransitory business activity. The term "nontransitory business activity," for the purposes of this exemption, means and includes the business of extracting, manufacturing, selling tangible and intangible property, printing, publishing, and performing contracts for ((the)) constructing or improving ((the)) real or personal property. It does not include the business of conducting a circus or other form of amusement when the personnel and property of such business regularly moves from one state into another, nor does it include casual or incidental business done by a nonresident lawyer, doctor or accountant.

- (i) Reasonable rental value. A reasonable rental value is normally determined by the rental price or using the fair market rental value of similar products of like quality and character if rental price is not reasonable. If a reasonable rental value cannot be determined because of the nature of property, such as it may not be possible to find similar products of like quality and character, monthly reasonable rental value may be determined based on depreciation plus one percent (per month) of the purchase price. For the purpose of this computation, depreciation should be computed on a straight-line basis with an assumption that there is no salvage value. The life of the asset must be based on "book" life rather than an accelerated life that might be used for federal tax purposes. This calculation applies even if the asset is fully depreciated.
- (ii) **Example.** A piece of equipment that originally cost \$100,000 and has a book life of 48 months results in a monthly rental value of $\$3,083 ((100,000/48) + (100,000 \times .01))$. This monthly value applies even if the asset is fully depreciated or is greater or less than the actual depreciation used for federal tax purposes. A lesser value can be used if the taxpayer retains documentation supporting the lesser value and that value is based on rental values.
- (j) Special provisions for vessel dealers and manufacturers. The value of an article used for a vessel held in inventory and used by a

vessel dealer or vessel manufacturer for personal use is the reasonable rental value of the vessel used. This value applies only if the vessel dealer or manufacturer can show that the vessel is truly held for sale and that the dealer or manufacturer is and has been making good faith efforts to sell the vessel. (See RCW 82.12.802.) This may result in a vessel manufacturer incurring multiple use tax liabilities with respect to multiple uses of the same vessel.

The use of a vessel by a vessel dealer or vessel manufacturer for certain purposes is not subject to use tax. For specific information on these exemptions see RCW 82.12.800 and 82.12.801.

- (k) Special provision for asphalt and aggregate. In the case of a person manufacturing or extracting asphalt or aggregates used in providing services taxable under RCW 82.04.280 (1) (b), the value of the asphalt or aggregates used is based on cost. Specifically, the value of the asphalt or aggregates equals the sum of all direct and indirect costs attributable to the asphalt or aggregates used, plus a public road construction market adjustment of five percent of those costs.
- (5) Who is liable for the tax? RCW 82.12.020 imposes use tax upon every person using tangible personal property or certain retail services as a consumer in the state of Washington. The law does not distinguish between persons using property (or certain retail services) for business or personal use. Thus, a Washington resident purchasing personal items via the internet or through a mail-order catalog has the same legal responsibility to report and remit use tax as does a corporation purchasing office supplies. The rate of the use tax is the same as the retail sales tax rate in the location where the property is used. Refer to WAC 458-20-145, Local sales and use tax, for further discussion about determining where use occurs.
- (a) When tax liability arises. Use tax is owed at the time the tangible personal property is first put to use in this state, unless an exemption is available.
 - (b) Reporting and remitting payment to the department of revenue.
- (i) Registered taxpayers. Persons registered with the department under RCW 82.32.030 to do business in Washington should use their excise tax return to report and remit use tax.
- (ii) Unregistered persons. Persons not required to be registered with the department should use a Consumer Use Tax Return to report and remit use tax. The Consumer Use Tax Return is available by:
 - (A) Using the department's website at dor.wa.gov;
- (B) Calling the department's telephone information center at 360-705-6705; or
- (C) Requesting the form at any of the department's local field offices.

The completed Consumer Use Tax Return, with payment, is due on or before the 25th day of the month following the month in which the tax liability occurs. For example, a person acquires clothing without payment of the retail sales tax during August. The Consumer Use Tax Return and the tax are due by September 25th.

The return and payment can be submitted electronically using the department's online system at dor.wa.gov, mailed, or delivered to any of the department's local field offices.

(6) How does use tax differ from the retail sales tax? There are circumstances where the law does not provide a use tax exemption to complement a retail sales tax exemption. Where there is no complementary use tax exemption, the buyer ((\neq)) or user is still responsible for remitting use tax on his or her use of the purchased property.

For instance, there is no complementary use tax exemption to the retail sales tax exemption in RCW 82.08.0251. This exemption provides a retail sales tax exemption for articles acquired in casual sales transactions, if the seller is not required to be registered with the department. Because there is no complementary use tax exemption, the buyer((\neq)) or user is responsible for remitting the use tax on his or her use of the purchased property. For example, if a person purchases furniture through a classified ad from a homeowner, the buyer is responsible for reporting and paying the use tax although the sale is exempt from retail sales tax.

- (7) Exceptions. The law provides certain exceptions to the imposition of tax on a single event. These exceptions occur when the law provides a method of determining the measure of tax different than the full value of the article being used.
- (a) Destroyed property. The mere destruction or discarding of tangible personal property as unusable or worthless is usually not considered a taxable "use." The following examples identify a number of facts and then state a conclusion.
- (i) Example 4. AA Computer Software (AA) has some obsolete inventory that will no longer sell as an updated version of the software is now available for purchase. AA decides to throw away this inventory even though it has never been used. As the software was never used, use tax is not owed on the destroyed inventory.
- (ii) Example 5. WW Dealer purchases a used vehicle for resale. WW Dealer publicizes an upcoming sale by airing a television commercial in which WW Dealer destroys the vehicle. WW Dealer's destruction of the vehicle for publicity purposes is considered use by a consumer. The vehicle is subject to use tax sourced at the location where WW Dealer destroys the vehicle.
- (b) Tangible personal property acquired by gift or donation. ((The use of)) Using property acquired by gift or donation is subject to the use tax, unless the person ((gifting)) giving or donating the property previously paid or remitted Washington retail sales or use tax on the purchase or use of the property. (See RCW 82.12.020.) However, a credit for tax paid in another jurisdiction is available if documentation of tax paid is provided. See subsection (8) of this rule for additional information.

Use tax does not apply when the same property is ((qifted)) given or donated back to the original ((giftor)) giver or donor if the original ((giftor)) giver or donor previously paid the retail sales tax or use tax.

- Example 6. John purchases a vehicle, pays retail sales tax on the purchase, and ((gifts)) gives the vehicle to Mary. Mary's use of the vehicle is not subject to use tax because John paid sales tax when he purchased the vehicle. After two years, Mary returns the vehicle to John. John's use of the vehicle is not subject to use tax because he paid sales tax when he originally purchased the vehicle. However, use tax is due if Mary ((gifts)) gives or donates the vehicle to a person other than John because Mary has not previously paid retail sales or use tax.
- (c) Tangible personal property put to both an exempt and taxable use. If property is first used for an exempt or nontaxable purpose and is later used for a nonexempt or taxable purpose, use tax is due on the value of the property when first used for the nonexempt or taxable purpose. For instance, RCW 82.12.0251 provides a use tax exemption for the temporary use within Washington of watercraft brought in by certain nonresidents. (See WAC 458-20-238, Sales of watercraft to nonres-

idents—Use of watercraft in Washington by nonresidents, for a detailed explanation of the exemption requirements.) However, use tax is due if the nonresident exceeds the temporary use threshold or the nonresident subsequently becomes a Washington resident.

(d) Intervening use of property purchased for resale. Persons purchasing tangible personal property for resale in the regular course of business may purchase the property at wholesale without paying retail sales tax provided the property is not put to intervening use, and the buyer provides the seller with a completed reseller permit. (See RCW 82.04.050 and 82.04.060.)

A buyer who purchases taxable property at wholesale and subsequently puts the property to intervening use is subject to either the retail sales tax (commonly referred to as "deferred retail sales tax") or use tax, unless a specific use tax exemption applies to the intervening use. The tax applies even if the property is at all times held out for sale and is in fact later sold. Tax is due even if the intervening use is the result of an unforeseen circumstance, such as when property is purchased for resale, the customer fails to satisfy the terms of the sales agreement, and the property is used until another customer is found. See WAC 458-20-102 Reseller permits regarding taxreporting requirements when a person purchases property for both resale and consumption.

(e) Using inventory to promote sales. Intervening use does not include the use of inventory for floor or window display purposes if that merchandise is subsequently sold as new merchandise. Likewise, intervening use does not include the use of inventory for demonstration purposes occurring with efforts to sell the same merchandise if that merchandise is subsequently sold as new merchandise. The fact that the selling price may be discounted because the property is shop worn from display or demonstration is not, by itself, controlling for the purposes of determining whether intervening use has occurred.

Evidence that property has been put to intervening use includes, but is not limited to, the following:

- (i) Property not sold as new merchandise. Intervening use occurs if, after use of the property for display or demonstration purposes, the property can no longer be sold as new merchandise. An indication that intervening use has occurred is if property is without a new model warranty if the sale of the property normally includes such a warranty.
- (ii) Capitalizing demonstrator or display property. The capitalization and depreciation of property is evidence of intervening use. Thus, there is a rebuttable presumption that intervening use occurs if the accounting records identify the property as a demonstrator or as display merchandise. The burden is on the person making such entries in the accounting records to substantiate any claims the property was not put to intervening use.
- (iii) Loaning property to promote sales. Intervening use includes loaning property to a customer or potential customer for the purpose of promoting sales of other products. For example, intervening use occurs if a coffee manufacturer ((and/)) or distributor loans brewing equipment to a customer to promote coffee sales, even if the equipment is subsequently sold to the same or different customer. In this example, the coffee manufacturer ((and/)) or distributor loaning the equipment would owe use tax on the full value of the equipment. If the manufacturer ((and/)) or distributor had not paid use tax, the customer would owe use tax on the reasonable rental value as this is a bail-

ment situation. See subsection (4)(f) of this section for the measure of tax on bailed articles.

- (f) Effect of the trade-in exclusion. The exclusion for the value of trade-in property from the measure of tax applies only if the trade-in property is of the same general type or classification as the property for which it was traded-in. There is no requirement that Washington's retail sales or use tax be previously paid on the tradein property. There is also no requirement that the property subject to use tax be acquired in Washington, or that the user be a Washington resident at the time he or she acquired the property. For additional information refer to WAC 458-20-247, Trade-ins, selling price, sellers' tax measure.
- (8) Credit for taxes paid in other jurisdictions. RCW 82.12.035 provides a credit against Washington's use tax for legally imposed retail sales or use taxes paid by the purchaser to: Any other state, possession, territory, or commonwealth of the United States, or any political subdivision of a state, the District of Columbia, or any foreign country or political subdivision of a foreign country. (See RCW 82.56.010.)
- (a) This use tax credit is available only if the present user, or his or her bailor or donor, has documentation that shows the retail sales or use tax was paid with respect to such property, extended warranty, digital products, digital codes, or service defined as a retail sale in RCW 82.04.050 to the other taxing jurisdiction.
- (b) This credit is not available for other types of taxes such as, but not limited to, value-added taxes (VATs).
- (c) For the purposes of allocating state and local use taxes, the department first applies the credit against the amount of any use tax due the state. Any unused portion of the credit is then applied against the amount of any use tax due to local jurisdictions. RCW 82.56.010, Multistate Tax Compact, Article V. Elements of Sales and Use Tax Laws.
- (9) No apportionment of use tax liability. Unless specifically provided by law, the value of the article or use tax liability may not be apportioned even though the user may use the property both within and without Washington, or use the property for both taxable and exempt purposes.
- (a) Example 7. A construction company using an airplane for traveling to and from its Washington office and out-of-state job sites must remit use tax on the full value of the airplane, even if the airplane was purchased and delivery taken outside Washington. There is no apportionment of this value even though the airplane is used both within and outside of Washington.
- (b) Exemption. For an exemption pertaining to use tax liability, see WAC 458-20-17401, Use tax liability for motor vehicles, trailers, and parts used by motor carriers operating in interstate or foreign commerce.

WSR 23-23-160 EXPEDITED RULES DEPARTMENT OF REVENUE

[Filed November 21, 2023, 10:29 a.m.]

Title of Rule and Other Identifying Information: WAC 458-20-210 Sales of tangible personal property for farming—Sales of agricultural products by farmers and 458-20-176 Commercial deep sea fishing—Commercial passenger fishing—Diesel fuel.

Purpose of the Proposal and Its Anticipated Effects, Including Any Changes in Existing Rules: These drafts show updates to reflect 2023 legislation, HB 1573, including updates to expiration dates for RCW 82.04.4266 and 82.04.4268. Additionally, out-of-date provisions have been struck and updated.

Reasons Supporting Proposal: This is an expedited update to the rule to incorporate changes made in 2023 legislation, HB 1573.

Statutory Authority for Adoption: RCW 82.32.300 and 82.01.060(2).

Statute Being Implemented: RCW 82.04.4266, 82.04.4268, 82.04.4269, 82.08.900, and 82.12.900.

Rule is not necessitated by federal law, federal or state court decision.

Name of Proponent: Department of revenue, governmental.

Name of Agency Personnel Responsible for Drafting: Tim Danforth, 6400 Linderson Way S.W., Tumwater, WA, 360-534-1538; Implementation and Enforcement: John Ryser, 6400 Linderson Way S.W., Tumwater, WA, 360-534-1605.

This notice meets the following criteria to use the expedited adoption process for these rules:

Adopts or incorporates by reference without material change federal statutes or regulations, Washington state statutes, rules of other Washington state agencies, shoreline master programs other than those programs governing shorelines of statewide significance, or, as referenced by Washington state law, national consensus codes that generally establish industry standards, if the material adopted or incorporated regulates the same subject matter and conduct as the adopting or incorporating rule.

Corrects typographical errors, makes address or name changes, or clarifies language of a rule without changing its effect.

Explanation of the Reason the Agency Believes the Expedited Rule-Making Process is Appropriate: The expedited rule-making process is appropriate for this rule because the department is incorporating the changes resulting from 2023 legislation, updating out-of-date provisions, and making clarifications that do not change the effect of the rule.

NOTICE

THIS RULE IS BEING PROPOSED UNDER AN EXPEDITED RULE-MAKING PROC-ESS THAT WILL ELIMINATE THE NEED FOR THE AGENCY TO HOLD PUBLIC HEAR-INGS, PREPARE A SMALL BUSINESS ECONOMIC IMPACT STATEMENT, OR PROVIDE RESPONSES TO THE CRITERIA FOR A SIGNIFICANT LEGISLATIVE RULE. IF YOU OBJECT TO THIS USE OF THE EXPEDITED RULE-MAKING PROCESS, YOU MUST EX-PRESS YOUR OBJECTIONS IN WRITING AND THEY MUST BE SENT TO Tim Danforth, Department of Revenue, P.O. Box 47453, Olympia, WA 98504-7453, phone 360-534-1538, fax 360-534-1606, email timd@dor.wa.gov, AND RE-CEIVED BY January 22, 2024.

November 21, 2023 Atif Aziz Rules Coordinator

OTS-5094.1

AMENDATORY SECTION (Amending WSR 18-13-094, filed 6/19/18, effective 7/20/18)

WAC 458-20-176 Commercial deep sea fishing—Commercial passenger fishing—Diesel fuel. (1) Introduction. This rule explains the business and occupation (B&O) tax, sales tax and use tax responsibilities of those engaged in commercial deep sea fishing, and suppliers selling to those persons.

Other rules that may apply. Readers may want to refer to other rules for additional information, including those in the following list:

- (a) WAC 458-20-119 Sales by caterers and food service contractors;
 - (b) WAC 458-20-135 Extracting natural products;
- (c) WAC 458-20-178 Use tax and the use of tangible personal prop-
- (d) WAC 458-20-193 Interstate sales of tangible personal property;
 - (e) WAC 458-20-244 Food and food ingredients.
 - (2) **Definitions**. The following definitions apply to this rule.
- (a) Commercial deep sea fishing. "Commercial deep sea fishing" means fishing done for profit outside the territorial waters of the state of Washington. It does not include sport fishing or the operation of charter boats for sport fishing. Nor does the phrase include the operation or purchase of watercraft for kelping, purse seining, or gill netting, because such fishing methods can be legally performed in Washington only within the territorial waters of the state (the threemile limit). Therefore, watercraft rigged for fishing by any of these methods will be deemed for use in other than commercial deep sea fishing unless proof, including documentation to be retained by sellers, is furnished that said watercraft will be used for these purposes exclusively outside the Washington territorial limit.
- (b) Commercial passenger fishing. "Commercial passenger fishing" means that done from charter boats for sport outside the territorial waters of the state of Washington.
- (c) Component part. "Component part" includes all tangible personal property that is attached to and a part of a watercraft. It includes dories, gurdies and accessories, bait tanks, baiting tables and turntables. It also includes spare parts that are designed for ultimate attachment to a watercraft. The term "component part" does not include equipment or furnishings of any kind that are not attached to a watercraft, nor does it include consumable supplies. Thus, it does not include, among other things, bedding, table and kitchen wares, fishing nets, hooks, lines, floats, hand tools, ice, fuel or lubricants.

- (d) Watercraft. "Watercraft" means every type of floating equipment that is designed for carrying fishing gear, fish catch or fishing crews, and used primarily in commercial deep sea fishing operations.
 - (3) Business and occupation tax.
- (a) Persons engaged in commercial deep sea fishing are not taxable under the extracting classification with respect to catches obtained outside the territorial waters of this state.
- (b) Such persons are taxable under either the retailing or the wholesaling classification with respect to sales made within this state, unless entitled to exemption by reason of the commerce clauses of the federal constitution.
- (c) Such persons may qualify for a B&O tax exemption under RCW 82.04.4269. This exemption pertains to the value of products or the gross proceeds of sales derived from:
- (i) Manufacturing seafood products that remain in a raw, raw frozen, or raw salted state at the completion of the manufacturing by that person; or
- (ii) In the ordinary course of business, manufactured seafood products that remain in a raw, raw frozen or raw salted state to buyers that transport the goods out of the state of Washington. A person taking an exemption must keep and preserve records for the period required by RCW 82.32.070 establishing that the goods were transported by the buyer in the ordinary course of business out of the state of Washington.
- (d) Persons claiming the exemption in (c) of this subsection must file a completed annual tax performance report with the department under RCW 82.32.534. In addition, persons claiming this tax preference must report the amount of the exemption on their monthly or quarterly excise tax return. For more information on reporting requirements for this tax preference see RCW 82.32.808.
- (e) The exemption provided by RCW 82.04.4269 is scheduled to expire on July 1, ((2025)) 2035.
 - (4) Retail sales tax.
- (a) Under RCW 82.08.0262, the retail sales tax does not apply to sales of watercraft (including component parts thereof) which are primarily for use in conducting commercial deep sea fishing operations, nor does retail sales tax apply to sales of or charges made for labor and services rendered in respect to the constructing, repairing, cleaning, altering or improving of such property.
- (b) The retail sales tax applies to sales made to persons engaged in commercial deep sea fishing of every type of tangible personal property (except only sales of watercraft and component parts thereof) and to sales of or charges made for labor and services rendered in respect to the construction, repairing, cleaning, altering or improving of such types of property. Thus, the retail sales tax applies to sales to such persons of such things as fishing nets, hooks, lines, floats and bait; table and kitchen wares; hand tools, ice, fuel except diesel fuel as noted in subsection (7) of this rule, and lubricants for use or consumption. For sales of food and food ingredients see WAC 458-20-119 and 458-20-244.
 - (5) Exemption certificates required.
- (a) Persons selling watercraft or component parts thereof to persons engaged in commercial deep sea fishing or performing services with respect to such craft or parts, are required to obtain from the buyer a certificate evidencing the exempt nature of the transaction.
- (b) Buyers claiming the exemption may use the department's Buyers' Retail Sales Tax Exemption Certificate. The certificate can be

found on the department's website at dor.wa.gov. Sellers must retain certificates in its records as evidence of the exempt nature of the sales to eligible buyers.

- (c) Fishing boats used primarily in commercial deep sea fishing operations that are incidentally used within the waters of this state are still eligible for the exemption from retail sales tax.
- (d) Sales of fishing boats, that are the types used in the waters of Puget Sound or the Columbia River and the tributaries thereof, and are not practical for use in deep sea fishing, are subject to the retail sales tax including sales of component parts thereof and on charges made for the repair of the same.
- (e) It is a gross misdemeanor for a buyer to make a false certificate of exemption for the purpose of avoiding the tax.
 - (6) Use tax.
- (a) The use tax does not apply to the use of watercraft or component parts thereof. RCW 82.12.0254.
- (b) The use tax applies to the actual use within this state of all other types of tangible personal property purchased at retail where the sales tax has not been paid and no exemption exists.
 - (7) Diesel fuel.
- (a) RCW 82.08.0298 and 82.12.0298 provide sales and use tax exemptions on diesel fuel for both commercial passenger fishing (charter boats for sport fishing) and commercial deep sea fishing operations.
- (b) Neither retail sales nor use tax applies with respect to sales or use of diesel fuel in the operation of watercraft in commercial deep sea fishing operations or commercial passenger fishing operations by persons who are regularly engaged in the business of such operations outside the territorial waters (three-mile limit) of this state. For purposes of this exemption, a person is not regularly engaged in either business if the person has gross receipts from the extra territorial operations of less than five thousand dollars a year. For persons involved in both commercial deep sea fishing operations and commercial passenger fishing operations, the receipts from both will be added together to determine eligibility for this exemption.
- (c) If a person qualifies for the exemptions by virtue of operating a deep sea fishing vessel, and has the requisite amount of gross receipts from that activity, all diesel fuel purchases and uses by such person for such vessel are tax exempt. It is not required that all the diesel fuel purchased be used outside the territorial waters of this state.
- (d) Diesel fuel exemption certificates required. Persons selling diesel fuel to such persons are required to obtain from the buyer a certificate evidencing the exempt nature of the transaction. This certificate must identify the buyer by name and address, and by the registered name and number of the watercraft with respect to which the purchase is made. Blanket certificates covering all diesel fuel purchases for specified watercraft may be used, where appropriate. A seller of diesel fuel who accepts such a certificate is not liable for sales tax on the diesel fuel sold. Certificates must be retained by the sellers in their permanent records as evidence of the exempt nature of diesel sales to eligible buyers. It is a gross misdemeanor for a buyer to make a false certificate of exemption for the purpose of avoiding the tax. Buyers may use the Buyers' Retail Sales Tax Exemption Certificate found on the department's website at dor.wa.gov.

AMENDATORY SECTION (Amending WSR 23-14-002, filed 6/21/23, effective 7/22/23)

- WAC 458-20-210 Sales of tangible personal property for farming— Sales of agricultural products by farmers. (1) Introduction. This rule explains the application of business and occupation (B&O), retail sales, and use taxes to the sale and/or use of feed, seed, fertilizer, spray materials, and other tangible personal property for farming. This rule also explains the application of B&O, retail sales, and litter taxes to the sale of agricultural products by farmers. Farmers should refer to WAC 458-20-101 (Tax registration and tax reporting) to determine whether they must obtain a tax registration endorsement or a temporary registration certificate from the department of revenue (department).
- (a) Examples. This rule contains examples that identify a number of facts and then state a conclusion. These examples should be used only as a general guide. The tax results of other situations must be determined after a review of all facts and circumstances.
- (b) Other rules that may be relevant. Farmers and persons making sales to farmers may also want to refer to rules in the following list for additional information:
- (i) WAC 458-20-178 Use tax and the use of tangible personal property;
- (ii) WAC 458-20-209 Farming for hire and horticultural services performed for farmers;
 - (iii) WAC 458-20-222 Veterinarians;
- (iv) WAC 458-20-239 Sales to nonresidents of farm machinery or implements, and related services;
 - (v) WAC 458-20-243 Litter tax; and
- (vi) WAC 458-20-262 Retail sales and use tax exemptions for ((agricultural employee)) farmworker housing.
- (2) Who is a farmer? A "farmer" is any person engaged in the business of growing, raising, or producing, on the person's own lands or on the lands in which the person has a present right of possession, any agricultural product to be sold. Effective July 1, 2015, a "farmer" also includes eligible apiarists that grow, raise, or produce honey bee products for sale, or provide bee pollination services. A "farmer" does not include a person growing, raising, or producing agricultural products for the person's own consumption; a person selling any animal or substance obtained therefrom in connection with the person's business of operating a stockyard, slaughterhouse, or packing house; or a person in respect to the business of taking, cultivating, or raising timber. RCW 82.04.213.
- (3) What is an agricultural product? An "agricultural product" is any product of plant cultivation or animal husbandry including, but not limited to: A product of horticulture, grain cultivation, vermiculture, viticulture, or aquaculture as defined in RCW 15.85.020; plantation Christmas trees; short-rotation hardwoods as defined in RCW 84.33.035; turf; or any animal, including, but not limited to, an animal that is a private sector cultured aquatic product as defined in RCW 15.85.020, a bird, an insect, or the substances obtained from such animals. Effective July 1, 2015, "agricultural product" includes honey bee products. An "agricultural product" does not include animals de-

fined under RCW 16.70.020 as "pet animals." Effective June 12, 2014, RCW 82.04.213 excludes cannabis from the definition of "agricultural product." Cannabis is any product with a THC concentration greater than .3 percent. RCW 82.04.213.

- (4) Who is an eligible apiarist? An "eligible apiarist" is a person who owns or keeps one or more bee colonies and who grows, raises, or produces honey bee products for sale at wholesale and is registered under RCW 15.60.021.
- (5) What are honey bee products? "Honey bee products" are queen honey bees, packaged honey bees, honey, pollen, bees wax, propolis, or other substances obtained from honey bees. "Honey bee products" do not include manufactured substances or articles. RCW 82.04.213.
- (6) What is cannabis? "Cannabis" is any product with a THC concentration greater than .3 percent. For additional information on cannabis see RCW 69.50.101.
- (7) Sales to farmers. Persons making sales of tangible personal property to farmers are generally subject to wholesaling or retailing B&O tax, as the case may be, on the gross proceeds of sales. Sales of some services performed for farmers, such as installing or repairing tangible personal property, are retail sales and subject to retailing B&O tax on the gross proceeds of such sales. Persons making retail sales must collect retail sales tax from the buyer, unless the sale is specifically exempt by law. Refer to subsection (9) of this rule for information about specific sales tax exemptions available for sales to farmers.
- (a) Documenting wholesale sales. A seller must take and retain from the buyer a copy of the buyer's reseller permit, or a completed "Farmers' Certificate for Wholesale Purchases and Sales Tax Exemptions" to document the wholesale nature of any transaction.
- (b) Buyer's responsibility when the seller does not collect retail sales tax on a retail sale. If the seller does not collect retail sales tax on a retail sale, the buyer must pay the retail sales tax (commonly referred to as "deferred sales tax") or use tax directly to the department, unless the sale is specifically exempt by law. The excise tax return does not have a separate line for reporting deferred sales tax. Consequently, deferred sales tax liability should be reported on the use tax line of the buyer's excise tax return. If a deferred sales tax or use tax liability is incurred by a farmer who is not required to obtain a tax registration endorsement from the department, the farmer must report the tax on a "Consumer Use Tax Return" and remit the appropriate tax to the department. For detailed information regarding use tax see WAC 458-20-178.

The Consumer Use Tax Return may be obtained by calling the department's telephone information center at 360-705-6705. The return may also be obtained from the department's website at dor.wa.gov.

(c) Feed, seed, seedlings, fertilizer, spray materials, and agents for enhanced pollination. Sales to farmers of feed, seed, seedlings, fertilizer, spray materials, and agents for enhanced pollination, including insects such as bees, to be used for the purpose of producing an agricultural product, whether for wholesale or retail sale, are wholesale sales.

However, when these items are sold to consumers for purposes other than producing agricultural products for sale, the sales are retail sales. For example, sales of feed to riding clubs, racetrack operators, boarders, or similar persons who do not resell the feed at a specific charge are retail sales. Sales of feed for feeding pets or work animals, or for raising animals for the purpose of producing agricultural products for personal consumption are also retail sales. Sales of seed, fertilizer, and spray materials for use on lawns and gardens, or for any other personal use, are likewise retail sales.

(i) What is feed? "Feed" is any substance used as food to sustain or improve animals, birds, fish, bees, or other insects, including whole and processed grains or mixtures thereof, hay and forages or meals made therefrom, mill feeds and feeding concentrates, stock salt, hay salt, sugar, pollen patties, bone meal, fish meal, cod liver oil, double purpose limestone grit, oyster shell, and other similar substances. Food additives that are given for their beneficial growth or weight effects are "feed."

Hormones or similar products that do not make a direct nutritional or energy contribution to the body are not "feed," nor are products used as medicines.

- (ii) What is seed? "Seed" is the propagative portions of plants commonly used for seeding or planting whether true seed, bulbs, plants, seed-like fruits, seedlings, or tubers. For purposes of this rule, "seed" does not include seeds or propagative portions of plants used to grow cannabis.
- (iii) What is fertilizer? "Fertilizer" is any substance containing one or more recognized plant nutrients and is used for its plant nutrient content and/or is designated for use in promoting plant growth. "Fertilizer" includes limes, gypsum, and manipulated animal and vegetable manures. There is no requirement that fertilizers be applied directly to the soil.
- (iv) What are spray materials? "Spray materials" are any substance or mixture of substances in liquid, powder, granular, dry flowable, or gaseous form, which is intended to prevent, destroy, control, repel, or mitigate any insect, rodent, nematode, mite, mollusk, fungus, weed, and any other form of plant or animal life normally considered to be a pest. The term includes treated materials, such as grains, that are intended to destroy, control, or repel such pests. "Spray materials" also include substances that act as plant regulators, defoliants, desiccants, or spray adjuvants.
 - (V) Examples.
- (A) **Example 1.** Sue grows vegetables for retail sale at a local market. Sue purchases fertilizers and spray materials that she applies to the vegetable plants. She also purchases feed for poultry that she raises to produce eggs for her personal consumption. Because the vegetables are an agricultural product produced for sale, retail sales tax does not apply to Sue's purchases of fertilizers and spray materials, provided she gives the seller a copy of her reseller permit, or a completed Farmers' Certificate for Wholesale Purchases and Sales Tax Exemptions. Retail sales tax applies to her purchases of poultry feed, as the poultry is raised to produce eggs for Sue's personal consump-
- (B) Example 2. WG Vineyards (WG) grows grapes that it uses to manufacture wine for sale. WG purchases pesticides and fertilizers that are applied to its vineyards. WG may purchase these pesticides and fertilizers at wholesale, provided WG gives the seller a copy of their reseller permit, or a completed Farmers' Certificate for Wholesale Purchases and Sales Tax Exemptions.
- (C) **Example 3.** Seed Co. contracts with farmers to raise seed. Seed Co. provides the seed and agrees to purchase the crop if it meets specified standards. The contracts provide that ownership of the crop is retained by Seed Co., and the risk of crop loss is borne by the farmers. The farmers must pay for the seed whether or not the crop

meets the specified standard. The transfer of the possession of the seed to each farmer is a wholesale sale, provided Seed Co. obtains a copy of their reseller permit, or a completed Farmers' Certificate for Wholesale Purchases and Sales Tax Exemptions from that farmer.

- (d) Chemical sprays or washes. Sales of chemical sprays or washes, whether to farmers or other persons, for the purpose of postharvest treatment of fruit for the prevention of scald, fungus, mold, or decay are wholesale sales.
- (e) Farming equipment. Sales to farmers of farming equipment such as machinery, machinery parts and repair, tools, and cleaning materials are retail sales and subject to retailing B&O and retail sales taxes, unless specifically exempt by law. Refer to subsections (7)(i) and (9) of this rule for information about sales tax exemptions available to farmers.
- (f) Packing materials and containers. Sales of packing materials and containers, or tangible personal property that will become part of a container, to a farmer who will sell the property to be contained therein are wholesale sales, provided the packing materials and containers are not put to intervening use by the farmer. Thus, sales to farmers of binder twine for binding bales of hay that will be sold or wrappers for fruit and vegetables to be sold are subject to wholesaling B&O tax. However, sales of packing materials and containers to a farmer who will use the items as a consumer are retail sales and subject to retailing B&O and retail sales taxes. Thus, sales of binder twine to a farmer for binding bales of hay that will be used to feed the farmer's livestock are retail sales.
- (q) Purchases for dual purposes. A buyer normally engaged in both consuming and reselling certain types of tangible personal property who is unable to determine at the time of purchase whether the particular property purchased will be consumed or resold must purchase according to the general nature of his or her business. RCW 82.08.130. If the buyer principally consumes the articles in question, the buyer should not give a copy of its reseller permit for any part of the purchase. If the buyer principally resells the articles, the buyer may provide a copy of its reseller permit for the entire purchase. For the purposes of this subsection, the term "principally" means greater than 50 percent.

If a buyer makes a purchase for dual purposes and does not give a copy of their reseller permit for any of the purchase and thereafter resells some of the articles purchased, the buyer may claim a "taxable amount for tax paid at source" deduction. For additional information regarding purchases for dual purposes and the "taxable amount for tax paid at source" deduction see WAC 458-20-102.

- (i) Potential deferred sales tax liability. If the buyer gives a copy of its reseller permit for all purchases and thereafter consumes some of the articles purchased, the buyer is liable for deferred sales tax and must remit the tax directly to the department. Refer to (b) of this subsection, WAC 458-20-102 and 458-20-178 for more information regarding deferred sales tax and use tax.
- (ii) Example 4. A farmer purchases binder twine for binding bales of hay. Some of the hay will be sold and some will be used to feed the farmer's livestock. More than 50 percent of the binder twine is used for binding bales of hay that will be sold. Because the farmer principally uses the binder twine for binding bales of hay that will be sold, the farmer may provide a copy of their reseller permit, or a completed Farmers' Certificate for Wholesale Purchases and Sales Tax Exemptions to the seller for the entire purchase. The farmer is liable

for deferred sales tax on the binder twine used for binding bales of hay that are used to feed the farmer's livestock and must remit the tax directly to the department.

- (h) "Fruit bin rentals" by fruit packers. Fruit packers often itemize their charges to farmers for various services related to the packing and storage of fruit. An example is a charge for the bins that the packer uses in the receiving, sorting, inspecting, and storing of fruit (commonly referred to as "bin rentals"). The packer delivers the bins to the grower, who fills them with fruit for eventual storage in the packer's warehouse. Charges by fruit packers to farmers for such bin rentals do not constitute the rental of tangible personal property to the farmer where the bins are under the control of the packer for use in the receiving, sorting, inspecting, and storing of fruit. These charges are income to the packer related to the receipt or storage of fruit. The packer, as the consumer of the bins, is subject to retail sales or use tax on the purchase or use of the bins. For information regarding the taxability of fruit packing by cooperative marketing associations and independent dealers acting as agents for others in the sales of fruit and produce see WAC 458-20-214.
- (i) Machinery and equipment used directly in a manufacturing operation. Machinery and equipment used directly in a manufacturing operation by a manufacturer or processor for hire is exempt from sales and use taxes provided that all requirements for the exemptions are met. RCW 82.08.02565 and 82.12.02565. These exemptions are commonly referred to as the M&E exemption. Farmers who use agricultural products that they have grown, raised, or produced as ingredients in a manufacturing process may be entitled to the M&E exemption on the acquisition of machinery and equipment used directly in their manufacturing operation. For more information on the M&E exemption see WAC 458-20-13601.
- (8) Sales by farmers. Farmers are not subject to B&O tax on wholesale sales of agricultural products. Effective July 1, 2015, bee pollination services provided to farmers by eligible apiarists also qualify for the exemption provided by RCW 82.04.330. Farmers who manufacture products using agricultural products that they have grown, raised, or produced should refer to (b) of this subsection for tax-reporting information.

Farmers are subject to retailing B&O tax on retail sales of agricultural products and retailing or wholesaling B&O tax on sales of nonagricultural products, as the case may be, unless specifically exempt by law. Also, B&O tax applies to sales of agricultural products that the seller has not grown, raised, or produced on the seller's own land or on land in which the seller has a present right of possession, whether these products are sold at wholesale or retail. Likewise, B&O tax applies to sales of animals or substances derived from animals in connection with the business of operating a stockyard, slaughterhouse, or packing house. Farmers may be eligible to claim a small business B&O tax credit if the amount of B&O tax liability in a reporting period is under a certain amount. For more information about the small business B&O tax credit see WAC 458-20-104.

(a) Litter tax. The gross proceeds of sales of certain products, including food for human or pet consumption, are subject to litter tax. RCW 82.19.020. Litter tax does not apply to sales of agricultural products that are exempt from B&O tax under RCW 82.04.330. RCW 82.19.050. Thus, farmers are not subject to litter tax on wholesale sales of agricultural products but are liable for litter tax on the gross proceeds of retail sales of agricultural products that constitute food for human or pet consumption. In addition, farmers that manufacture products for use and consumption within this state (e.g., a farmer who produces wine from grapes that the farmer has grown) may be liable for litter tax measured by the value of the products manufactured. For more information about the litter tax see chapter 82.19 RCW and WAC 458-20-243.

- **Example 5.** RD Orchards (RD) grows apples at its orchards. Most apples are sold at wholesale, but RD operates a seasonal roadside fruit stand from which it sells apples at retail. The wholesale sales of apples are exempt from both B&O and litter taxes. The retail sales of apples are subject to retailing B&O and litter taxes but are exempt from sales tax because the apples are sold as a food product for human consumption. Refer to subsection (9)(d) of this rule for more information about the retail sales tax exemption applicable to sales of food products for human consumption.
- (b) Farmers using agricultural products in a manufacturing process. The B&O tax exemption provided by RCW 82.04.330 does not apply to any person selling manufactured substances or articles. Thus, farmers who manufacture products using agricultural products that they have grown, raised, or produced are subject to manufacturing B&O tax on the value of products manufactured. Farmers who sell their manufactured products at retail or wholesale in the state of Washington are also generally subject to the retailing or wholesaling B&O tax, as the case may be. In such cases, a multiple activities tax credit (MATC) may be available. Refer to WAC 458-20-136 (Manufacturing, processing for hire, fabricating) and WAC 458-20-19301 (Multiple activities tax credits), respectively, for more information about the manufacturing B&O tax and the MATC.
- (i) Manufacturing fresh fruits and vegetables. RCW 82.04.4266 provides a B&O tax exemption to persons manufacturing fresh fruits or vegetables by canning, preserving, freezing, processing, or dehydrating fresh fruits or vegetables. For purposes of this rule, "fruits" and "vegetables" does not include cannabis.

Wholesale sales of fresh fruits or vegetables canned, preserved, frozen, processed, or dehydrated by the seller and sold to purchasers who transport the goods out of this state in the ordinary course of business are also eligible for this exemption. A seller must keep and preserve records for the period required by RCW 82.32.070 establishing that the purchaser transported the goods out of Washington state.

- (A) A person claiming the exemption must file a complete annual tax performance report with the department under RCW 82.32.534. In addition, persons claiming this tax preference must report the amount of the exemption on their monthly or quarterly excise tax return. For more information on reporting requirements for this tax preference see RCW 82.32.808.
- (B) RCW 82.04.4266 is scheduled to expire July 1, ((2025)) 2035, at which time the preferential B&O tax rate under RCW 82.04.260 will apply.
- (ii) Manufacturing dairy products. RCW 82.04.4268 provides a B&O tax exemption to persons manufacturing dairy products, not including any cannabis-infused product, that as of September 20, 2001, are identified in 21 C.F.R., chapter 1, parts 131, 133, and 135. These products include milk, buttermilk, cream, yogurt, cheese, and ice cream, and also include by-products from the manufacturing of dairy products such as whey and casein.

The exemption also applies to persons selling manufactured dairy products to purchasers who transport the goods out of Washington state in the ordinary course of business and, until July 1, 2025, to purchasers who use such dairy products as an ingredient or component in the manufacturing of a dairy product in Washington state. Unlike the exemption for certain wholesale sales of fresh fruits or vegetables (see (b)(i) of this subsection), the exemption for sales of qualifying dairy products does not require that the sales be made at wholesale.

A seller must keep and preserve records for the period required by RCW 82.32.070 establishing that the purchaser transported the goods out of Washington state or the goods were sold to a manufacturer for use as an ingredient or component in the manufacturing of a dairy product in Washington state.

- (A) A person claiming the exemption must file a complete annual tax performance report with the department under RCW 82.32.534. In addition, persons claiming this tax preference must report the amount of the exemption on their monthly or quarterly excise tax return. For more information on reporting requirements for this tax preference see RCW 82.32.808.
- (B) RCW 82.04.4268 is generally scheduled to expire July 1, ((2025)) 2035, at which time the preferential B&O tax rate under RCW 82.04.260 will apply. The exemption for sales of dairy products to purchasers who use such dairy products as an ingredient or component in the manufacturing of a dairy product in Washington state expires July 1, 2025.
- ((C) Effective October 1, 2013, the exemption provided by RCW 82.04.4268 expanded to include wholesale sales by a dairy product manufacturer to a purchaser who uses the dairy products as an ingredient or component in the manufacturing in Washington of another dairy product. The definition of dairy products was expanded to include products comprised of not less than 70 percent dairy products measured by weight or volume.))
- (c) Raising cattle for wholesale sale. RCW 82.04.330 provides a B&O tax exemption to persons who raise cattle for wholesale sale provided that the cattle are held for at least 60 days prior to the sale. Persons who hold cattle for fewer than 60 days before reselling the cattle are not considered to be engaging in the normal activities of growing, raising, or producing livestock for sale.
- **Example 6.** A feedlot operation purchases cattle and feeds them until they attain a good market condition. The cattle are then sold at wholesale. The feedlot operator is exempt from B&O tax on wholesale sales of cattle if it held the cattle for at least 60 days while they were prepared for market. However, the feedlot operator is subject to wholesaling B&O tax on wholesale sales of cattle held for fewer than 60 days prior to the sale.
- (d) B&O tax exemptions available to farmers. In addition to the exemption for wholesale sales of agricultural products, several other B&O tax exemptions available to farmers are discussed in this subsection.
- (i) Growing, raising, or producing agricultural products owned by other persons. RCW 82.04.330 exempts amounts received by a farmer for growing, raising, or producing agricultural products owned by others, such as custom feed operations.
- Example 7. A farmer is engaged in the business of raising cattle owned by others (commonly referred to as "custom feeding"). After the cattle attain a good market condition, the owner sells them. Amounts received by the farmer for custom feeding are exempt from B&O tax under RCW 82.04.330, provided that the farmer held the cattle for at least 60 days. Farmers are not considered to be engaging in the activ-

ity of raising cattle for sale unless the cattle are held for at least 60 days while the cattle are prepared for market. (See (c) of this subsection.)

- (ii) Processed hops shipped outside Washington for first use. RCW 82.04.337 exempts amounts received by hop growers or dealers for hops shipped outside the state of Washington for first use, if those hops have been processed into extract, pellets, or powder in this state. However, the processor or warehouser of such products is not exempt on amounts charged for processing or warehousing such products.
- (iii) Sales of hatching eggs or poultry. RCW 82.04.410 exempts amounts received for the sale of hatching eggs or poultry by farmers producing hatching eggs or poultry, when these agricultural products are for use in the production for sale of poultry or poultry products.
- (9) Retail sales tax and use tax exemptions. This subsection provides information about a number of retail sales tax and corresponding use tax exemptions available to farmers and persons buying tangible personal property at retail from farmers. Some exemptions require the buyer to provide the seller with an exemption certificate. Refer to subsection (10) of this rule for additional information regarding exemption certificates.
- (a) Pollen. RCW 82.08.0277 and 82.12.0273 exempt the sale and use of pollen from retail sales and use taxes.
- (b) **Semen.** RCW 82.08.0272 and 82.12.0267 exempt the sale and use of semen used in the artificial insemination of livestock from retail sales and use taxes.
- (c) Feed for livestock at public livestock markets. RCW 82.08.0296 and 82.12.0296 exempt the sale and use of feed to be consumed by livestock at a public livestock market from retail sales and use taxes.
- (d) Food products. RCW 82.08.0293 and 82.12.0293 exempt the sale and use of food products for human consumption from retail sales and use taxes. These exemptions also apply to the sale or use of livestock for personal consumption as food. For more information about food products that qualify for this exemption see WAC 458-20-244.
- (e) Auction sales of farm property. RCW 82.08.0257 and 82.12.0258 exempt from retail sales and use taxes tangible personal property, including household goods, which has been used in conducting a farm activity, if the property is purchased from a farmer, as defined in RCW 82.04.213, at an auction sale held or conducted by an auctioneer on a farm. Effective June 12, 2014, these exemptions do not apply to personal property used by a person in the production of cannabis.
- (f) Poultry. RCW 82.08.0267 and 82.12.0262 exempt from retail sales and use taxes the sale and use of poultry used in the production for sale of poultry or poultry products.
- Example 8. A poultry hatchery produces poultry from eggs. The resulting poultry are sold to egg producers. These sales are exempt from retail sales tax under RCW 82.08.0267. (They are also exempt from B&O tax. See subsection (8)(d)(iii) of this rule.)
- (q) Leases of irrigation equipment. RCW 82.08.0288 and 82.12.0283 exempt the lease or use of irrigation equipment from retail sales and use taxes, but only if:
- (i) The lessor purchased the irrigation equipment for the purpose of irrigating land controlled by the lessor;
- (ii) The lessor has paid retail sales or use tax upon the irrigation equipment;
- (iii) The irrigation equipment is attached to the land in whole or in part;

- (iv) Effective June 12, 2014, the irrigation equipment is not used in the production of cannabis; and
- (v) The irrigation equipment is leased to the lessee as an incidental part of the lease of the underlying land and is used solely on such land.
- (h) Beef and dairy cattle. RCW 82.08.0259 and 82.12.0261 exempt the sale and use of beef and dairy cattle, to be used by a farmer in producing an agricultural product, from retail sales and use taxes.
- Example 9. John operates a farm where he raises beef and dairy cattle for sale. He also raises other livestock for sale including hogs, sheep, and goats. John's sales of beef and dairy cattle for use on a farm are exempt from retail sales tax. However, John must collect retail sales tax on all retail sales of sheep, goats, and hogs unless the sales qualify for either the food products exemption described in (d) of this subsection, or the exemption for sales of livestock for breeding purposes described in this subsection (9)(i) of this rule.
- (i) Livestock for breeding purposes. RCW 82.08.0259 and 82.12.0261 exempt the sale or use of livestock, as defined in RCW 16.36.005, for breeding purposes where the animals are registered in a nationally recognized breed association from retail sales and use taxes.
- Example 10. ABC Farms raises and sells quarter horses registered in the American Quarter Horse Association (AQHA). Quarter horses are generally recognized as a definite breed of horse, and the AQHA is a nationally recognized breed association. Therefore, ABC Farms is not required to collect sales tax on retail sales of quarter horses for breeding purposes, provided it receives and retains a completed exemption certificate from the buyer.
- (j) Bedding materials for chickens. RCW 82.08.920 and 82.12.920 exempt from retail sales and use taxes the sale to and use of bedding materials by farmers to accumulate and facilitate the removal of chicken manure, provided the farmer is raising chickens that are sold as agricultural products.
- (i) What are bedding materials? "Bedding materials" are wood shavings, straw, sawdust, shredded paper, and other similar materials.
- (ii) Example 11. Farmer raises chickens for use in producing eggs for sale. When the chickens are no longer useful for producing eggs, Farmer sells them to food processors for soup and stew meat. Farmer purchases bedding materials used to accumulate and facilitate the removal of chicken manure. The purchases of bedding materials by Farmer are exempt from retail sales tax as long as Farmer provides the seller with a completed Farmers' Certificate for Wholesale Purchases and Sales Tax Exemptions. See subsection (10) of this rule for where to find an exemption certificate. The seller must retain a copy of the exemption certificate for its records.

The exemption merely requires that the chickens be sold as agricultural products. It is immaterial that Farmer primarily raises the chickens to produce eggs.

- (k) Propane or natural gas used to heat structures housing chickens. RCW 82.08.910 and 82.12.910 exempt from retail sales and use taxes the sale to and use of propane or natural gas by farmers to heat structures used to house chickens. The propane or natural gas must be used exclusively to heat the structures, and the structures must be used exclusively to house chickens that are sold as agricultural prod-
- (i) What are "structures"? "Structures" are barns, sheds, and other similar buildings in which chickens are housed.

(ii) Example 12. Farmer purchases natural gas that is used to heat structures housing chickens. The natural gas is used exclusively to heat the structures, and the structures are used exclusively to house chickens. The chickens are used to produce eggs. When the chickens are no longer useful for producing eggs, Farmer sells the chickens to food processors for soup and stew meat. The purchase of natural gas by Farmer is exempt from retail sales tax as long as Farmer provides the seller with a completed Farmers' Certificate for Wholesale Purchases and Sales Tax Exemptions. See subsection (10) of this rule for where to find an exemption certificate. The seller must retain a copy of the exemption certificate for its records.

The exemption merely requires that the chickens be sold as agricultural products. It is immaterial that Farmer primarily houses these chickens to produce eggs.

- (iii) Example 13. Farmer purchases natural gas that is used to heat structures used in the incubation of chicken eggs and structures used for washing, packing, and storing eggs. The natural gas used to heat these structures is not exempt from retail sales tax because the structures are not used exclusively to house chickens that are sold as agricultural products.
 - (1) ((Farm fuel used for agricultural purposes.
- $\frac{(i)}{(i)}$) Diesel, biodiesel, and aircraft fuel((s)) for farm fuel users. RCW 82.08.865 and 82.12.865 exempt from retail sales and use taxes the sale and use of diesel fuel, biodiesel fuel, and aircraft fuel, to farm fuel users for agricultural purposes. The exemptions apply to a fuel blend if all of the component fuels of the blend would otherwise be exempt if the component fuels were sold as separate products. The buyer must provide the seller with a completed Farmers' Certificate for Wholesale Purchases and Sales Tax Exemptions. See subsection (10) of this rule for where to find an exemption certificate. The seller must retain a copy of the exemption certificate for its records.
- $((\frac{A}{A}))$ (i) The exemptions apply to nonhighway uses for production of agricultural products and for providing horticultural services to farmers. Horticultural services include:
 - (((1))) (A) Soil preparation services;
 - (((II))) <u>(B)</u> Crop cultivation services;
 - (((III))) <u>(C)</u> Crop harvesting services.
- $((\frac{B}{B}))$ (ii) The exemptions do not apply to uses other than for agricultural purposes. Agricultural purposes do not include:
- ((((1)))) (A) Heating space for human habitation or water for human consumption; or
- (((II))) (B) Transporting on public roads individuals, agricultural products, farm machinery or equipment, or other tangible personal property, except when the transportation is incidental to transportation on private property and the fuel used for such transportation is not subject to tax under chapter 82.38 RCW.
- (((ii) Propane and natural gas used in distilling mint on a farm. Effective October 1, 2013, RCW 82.08.220 and 82.12.220 exempt from retail sales and use taxes sales to and use by farmers of propane or natural gas used exclusively to distill mint on a farm. The buyer must provide the seller with a completed Farmers' Certificate for Wholesale Purchases and Sales Tax Exemptions. The seller must retain a copy of the exemption certificate for its records. See subsection (10) of this rule for where to find an exemption certificate. The seller must also report amounts claimed for exemption when electronically filing excise tax returns. This exemption is scheduled to expire July 1, 2017.))

- (m) Nutrient management equipment and facilities. RCW 82.08.890 and 82.12.890 provide retail sales and use tax exemptions for the sale to or use by eligible persons of:
 - (i) Qualifying livestock nutrient management equipment;
- (ii) Labor and services rendered in respect to installing, repairing, cleaning, altering, or improving qualifying livestock nutrient management equipment; and
- (iii) Labor and services rendered in respect to repairing, cleaning, altering, or improving qualifying livestock nutrient management facilities, or to tangible personal property that becomes an ingredient or component of qualifying livestock nutrient management facilities in the course of repairing, cleaning, altering, or improving such facilities.
- (iv) Nonqualifying labor and services. This subsection (9) (m) (iii) of this rule does not include the sale of or charge made for labor and services rendered in respect to the constructing of new, or replacing previously existing, qualifying livestock nutrient management facilities, or tangible personal property that becomes an ingredient or component of qualifying livestock nutrient management facilities during the course of constructing new, or replacing previously existing qualifying livestock nutrient management facilities.
- (v) Nutrient management plan must be certified or approved. The exemptions provided by RCW 82.08.890 and 82.12.890 apply to sales made after the livestock nutrient management plan is:
 - (A) Certified under chapter 90.64 RCW;
- (B) Approved as part of the permit issued under chapter 90.48 RCW; or
- (C) Approved by a conservation district and who qualifies for the exemption provided under RCW 82.08.855. ((Effective June 12, 2014, the requirement for the department to issue exemption certificates was removed.)) A Farmers' Certificate for Wholesale Purchases and Sales Tax Exemptions should be completed and provided to the seller. In lieu of the exemption certificate, a seller may capture the relevant data elements as allowed under the streamlined sales and use tax agreement. The seller must retain a copy of the certificate or the data elements for the seller's files.
- (vi) Definitions. For the purpose of these exemptions, the following definitions apply:
- (A) "Animal feeding operation" means a lot or facility, other than an aquatic animal production facility, where the following conditions are met:
- · Animals, other than aquatic animals, have been, are, or will be stabled or confined and fed or maintained for a total of 45 days or more in any 12-month period; and
- Crops, vegetation, forage growth, or post-harvest residues are not sustained in the normal growing season over any portion of the lot or facility.
- (B) "Conservation district" means a subdivision of state government organized under chapter 89.08 RCW.
 - (C) "Eligible person" means a person:
- · Licensed to produce milk under chapter 15.36 RCW who has a certified dairy nutrient management plan, as required by chapter 90.64 RCW; or
- Who owns an animal feeding operation and has a permit issued under chapter 90.48 RCW; or
- Who owns an animal feeding operation and has a nutrient management plan approved by a conservation district as meeting natural re-

source conservation service field office technical guide standards and who qualifies for the exemption provided under RCW 82.08.855.

- (D) "Handling and treatment of livestock manure" means the activities of collecting, storing, moving, or transporting livestock manure, separating livestock manure solids from liquids, or applying livestock manure to the agricultural lands of an eligible person other than through the use of pivot or linear type traveling irrigation systems.
- (E) "Permit" means either a state waste discharge permit or a National Pollutant Discharge Elimination System permit, or both.
- (F) "Qualifying livestock nutrient management equipment" means the tangible personal property listed below for exclusive use in the handling and treatment of livestock manure, including repair and replacement parts for the same equipment:

Aerators Agitators Augers Conveyers Gutter cleaners Hard-hose reel traveler irrigation systems Lagoon and pond liners and floating covers Loaders Manure composting devices Manure spreaders Manure tank wagons Manure vacuum tanks Poultry house cleaners Poultry house flame sterilizers Poultry house washers Poultry litter saver machines Pipes Pumps Scrapers Separators Slurry injectors and hoses Wheelbarrows, shovels, and pitchforks.

(G) "Qualifying livestock nutrient management facilities" means the exclusive use in the handling and treatment of livestock manure of the facilities listed below:

Flush systems

Lagoons

Liquid livestock manure storage structures, such as concrete tanks or glass-lined steel tanks

Structures used solely for dry storage of manure, including roofed stacking facilities.

- (n) Anaerobic digesters (effective July 1, 2018).
- (i) RCW 82.08.900 and 82.12.900 provide retail sales and use tax exemptions for purchases and uses by eligible persons:
- (A) In respect to equipment necessary to process biogas from a landfill into marketable coproducts including, but not limited to, biogas conditioning, compression, and electrical generation equipment, or to services rendered in respect to installing, constructing, repairing, cleaning, altering, or improving equipment necessary to process biogas from a landfill into marketable coproducts; and

- (B) Establishing or operating anaerobic digesters or to services rendered in respect to installing, constructing, repairing, cleaning, altering, or improving an anaerobic digester. The exemptions include sales of tangible personal property that becomes an ingredient or component of the anaerobic digester. Anaerobic digester means a facility that processes organic material into biogas and digestate using microorganisms in a decomposition process within a closed, oxygen-free container as well as the equipment necessary to process biogas or digestate produced by an anaerobic digester into marketable coproducts including, but not limited to, biogas conditioning, compression, nutrient recovery, and electrical generation equipment.
- (ii) ((Exemption certificate. Effective July 24, 2015, eligible persons no longer need to apply for an exemption certificate. An "eligible person" is any person establishing or operating an anaerobic digester or landfill or processing biogas from an anaerobic digester or landfill into marketable coproducts.
- (iii))) Records retention and exemption certificate. Persons claiming the exemptions under RCW 82.08.900 and 82.12.900 must keep records necessary for the department to verify eligibility. Sellers may make tax exempt sales only if the buyer provides the seller with a completed Farmers' Certificate for Wholesale Purchases and Sales Tax Exemptions, and the seller retains a copy of the certificate for its files. See subsection (10) of this rule for where to find an exemption certificate.
- (o) (i) Anaerobic digesters (effective until ((July 1, 2018)) January 1, 2029). RCW 82.08.900 and 82.12.900 provide retail sales and use tax exemptions for purchases and uses by eligible persons establishing or operating anaerobic digesters or to services rendered in respect to installing, constructing, repairing, cleaning, altering, or improving an anaerobic digester. The exemptions include sales of tangible personal property that becomes an ingredient or component of the anaerobic digester. The anaerobic digester must be used primarily (more than 50 percent measured by volume or weight) to treat livestock manure. Anaerobic digester is a facility that processes manure from livestock into biogas and dried manure using microorganisms in a decomposition process within a closed, oxygen-free container.
- (((i) Exemption certificate. Effective July 24, 2015, eligible persons no longer need to apply for an exemption certificate. An "eligible person" is any person establishing or operating an anaerobic digester to treat primarily livestock manure.))
- (ii) Records retention and exemption certificate. Persons claiming the exemptions under RCW 82.08.900 and 82.12.900 must keep records necessary for the department to verify eligibility. Sellers may make tax exempt sales only if the buyer provides the seller with a completed Farmers' Certificate for Wholesale Purchases and Sales Tax Exemptions, and the seller retains a copy of the certificate for its files. See subsection (10) of this rule for where to find an exemption certificate.
- (p) Animal pharmaceuticals. RCW 82.08.880 and 82.12.880 exempt from retail sales and use taxes the sale of and use of certain animal pharmaceuticals when sold to, or used by, farmers or veterinarians. To qualify for the exemption, the animal pharmaceutical must be administered to an animal raised by a farmer for the purpose of producing an agricultural product for sale. In addition, the animal pharmaceutical must be approved by the United States Department of Agriculture (USDA) or the United States Food and Drug Administration (FDA).

- (i) Who is a veterinarian? A "veterinarian" means a person who is licensed to practice veterinary medicine, surgery, or dentistry under chapter 18.92 RCW.
- (ii) How can I determine whether the FDA or USDA has approved an animal pharmaceutical? The FDA and USDA have an established approval process set forth in federal regulations. The FDA maintains a list of all approved animal pharmaceuticals called the "Green Book." The USDA maintains a list of approved biotechnology products called the "Veterinary Biologics Product Catalogue." Pharmaceuticals that are not on either of these lists have not been approved and are not eligible for the exemption.
- (iii) Example 17. Dairy Farmer purchases sterilizing agents. The sterilizing agents are applied to the equipment and facilities where Dairy Farmer's cows are milked. Dairy Farmer also purchases teat dips, antiseptic udder washes, and salves that are not listed in either the FDA's Green Book of approved animal pharmaceuticals or the USDA's Veterinary Biologics Product Catalogue of approved biotechnology products. The purchases of sterilizing agents are not exempt as animal pharmaceuticals because the sterilizing agents are not administered to animals. The teat dips, antiseptic udder washes, and salves are likewise not exempt because they have not been approved by the FDA or US-
- (iv) What type of animal must the pharmaceutical be administered to? As explained above, the exemptions are limited to the sale and use of animal pharmaceuticals administered to an animal that is raised by a farmer for the purpose of producing an agricultural product for sale. The conditions under which a farmer may purchase and use tax-exempt animal pharmaceuticals are similar to those under which a farmer may purchase and use feed at wholesale. Both types of purchases and uses require that the particular product be sold to or used by a farmer (or a veterinarian in the case of animal pharmaceuticals), and that the product be given or administered to an animal raised by a farmer for the purpose of producing an agricultural product for sale.
- (V) Examples of animals raised for the purpose of producing agricultural products for sale. For purposes of the exemptions, the following is a nonexclusive list of examples of animals that are being raised for the purpose of producing an agricultural product for sale, presuming all other requirements for the exemption are met:
- (A) Horses, cattle, or other livestock raised by a farmer for sale;
- (B) Cattle raised by a farmer for the purpose of slaughtering, if the resulting products are sold;
- (C) Milk cows raised and/or used by a dairy farmer for the purpose of producing milk for sale;
- (D) Horses raised by a farmer for the purpose of producing foals for sale;
- (E) Sheep raised by a farmer for the purpose of producing wool for sale; and
- (F) "Private sector cultured aquatic products" as defined by RCW 15.85.020 (e.g., salmon, catfish, and mussels) raised by an aquatic farmer for the purpose of sale.
- (vi) Examples of animals that are not raised for the purpose of producing agricultural products for sale. For purposes of the exemptions, the following nonexclusive list of examples do not qualify because the animals are not being raised for the purpose of producing an agricultural product for sale:

- (A) Cattle raised for the purpose of slaughtering if the resulting products are not produced for sale;
 - (B) Sheep and other livestock raised as pets;
- (C) Dogs or cats, whether raised as pets or for sale. Dogs and cats are pet animals; therefore, they are not considered to be agricultural products. (See subsection (3) of this rule); and
- (D) Horses raised for the purpose of racing, showing, riding, and jumping. However, if at some future time the horses are no longer raised for racing, showing, riding, or jumping and are instead being raised by a farmer for the purpose of producing foals for sale, the exemption will apply if all other requirements for the exemption are met.
- (vii) Do products that are used to administer animal pharmaceuticals qualify for the exemption? Sales and uses of products that are used to administer animal pharmaceuticals (e.g., syringes) do not qualify for the exemptions, even if they are later used to administer a tax-exempt animal pharmaceutical. However, sales and uses of tax-exempt animal pharmaceuticals contained in a product used to administer the animal pharmaceutical (e.g., a dose of a tax-exempt pharmaceutical contained in a syringe or cotton applicator) qualify for the exemption.
- $(\ensuremath{\mathtt{q}})$ Replacement parts for qualifying farm machinery and equipment. RCW 82.08.855 and 82.12.855 exempt from retail sales and use taxes sales to and uses by eligible farmers of replacement parts for qualifying farm machinery and equipment. Also included are: Labor and services rendered during the installation of repair parts; and labor and services rendered during repair as long as no tangible personal property is installed, incorporated, or placed in, or becomes an ingredient or component of the qualifying equipment other than replacement parts.
 - (i) The following definitions apply to this subsection:
 - (A) "Eligible farmer" as defined in RCW 82.08.855(4).
- (B) "Qualifying farm machinery and equipment" means machinery and equipment used primarily by an eligible farmer for growing, raising, or producing agricultural products, and effective July 1, 2015, providing bee pollination services, or both.
 - (C) "Qualifying farm machinery and equipment" does not include:
- Vehicles as defined in RCW 46.04.670, other than farm tractors as defined in RCW 46.04.180, farm vehicles and other farm implements. "Farm implements" means machinery or equipment manufactured, designed, or reconstructed for agricultural purposes and used primarily by an eligible farmer to grow, raise, or produce agricultural products, but does not include lawn tractors and all-terrain vehicles;
 - Aircraft;
 - · Hand tools and hand-powered tools; and
 - Property with a useful life of less than one year.
- (D) "Replacement parts" means those parts that replace an existing part, or which are essential to maintain the working condition, of a piece of qualifying farm machinery or equipment. Paint, fuel, oil, hydraulic fluids, antifreeze, and similar items are not replacement parts except when installed, incorporated, or placed in qualifying farm machinery and equipment during the course of installing replacement parts as defined here or making repairs as described above in (p) of this subsection.
- (ii) Records retention and exemption certificate. ((Prior to June 12, 2014, the department was required to provide an exemption certifi-

cate to an eligible farmer or renew an exemption certificate when the eligible farmer applied for a renewal.

- (A))) Persons claiming the exemptions must keep records necessary for the department to verify eligibility. Sellers making tax-exempt sales must obtain, and retain in its files, a completed Farmers' Certificate for Wholesale Purchases and Sales Tax Exemptions from the farmer. In lieu of the exemption certificate, a seller may capture the relevant data elements as allowed under the streamlined sales and use tax agreement.
- (((B) The exemptions provided by RCW 82.08.890 and 82.12.890 do not apply to sales made from July 1, 2010, through June 30, 2013.))
- (10) Sales tax exemption certificates. As indicated in subsection (9) of this rule, certain sales of tangible personal property and retail services either to or by farmers are exempt from retail sales tax. A person claiming an exemption must keep records necessary for the department to verify eligibility for each claimed exemption. ((Effective June 12, 2014, the requirement for the department to issue certificates to qualified farmers was removed. Instead, farmers)) A person claiming an exemption under subsection (9) of this rule may complete and use the department's Farmers' Certificate for Wholesale Purchases and Sales Tax Exemptions. Refer to the department's website at dor.wa.gov for the exemption certificate. In lieu of an exemption certificate, a seller may capture the relevant data elements as provided under the streamlined sales and use tax agreement as allowed under RCW 82.08.050. Sellers must retain a copy of the exemption certificate or the data elements in their files. Without proper documentation, sellers are liable for payment of the retail sales tax on sales claimed as exempt.